Florida International University FIU Digital Commons

FIU Electronic Theses and Dissertations

University Graduate School

11-7-2022

# A Resilience Assessment Framework of Infrastructure Systems by Integrating Social Equity to Support Disaster Resilience Decision Making

Sunil Dhakal Florida International University, sdhak003@fiu.edu

Follow this and additional works at: https://digitalcommons.fiu.edu/etd

Part of the Civil Engineering Commons

## **Recommended Citation**

Dhakal, Sunil, "A Resilience Assessment Framework of Infrastructure Systems by Integrating Social Equity to Support Disaster Resilience Decision Making" (2022). *FIU Electronic Theses and Dissertations*. 5170. https://digitalcommons.fiu.edu/etd/5170

This work is brought to you for free and open access by the University Graduate School at FIU Digital Commons. It has been accepted for inclusion in FIU Electronic Theses and Dissertations by an authorized administrator of FIU Digital Commons. For more information, please contact dcc@fiu.edu.

FLORIDA INTERNATIONAL UNIVERSITY Miami, Florida

# A RESILIENCE ASSESSMENT FRAMEWORK OF INFRASTRUCTURE SYSTEMS BY INTEGRATING SOCIAL EQUITY TO SUPPORT DISASTER RESILIENCE DECISION MAKING

A dissertation submitted in partial fulfillment of

the requirements for the degree of

DOCTOR OF PHILOSOPHY

in

CIVIL ENGINEERING

by

Sunil Dhakal

2022

To: Dean John L. Volakis College of Engineering and Computing

This dissertation, written by Sunil Dhakal, and entitled A Resilience Assessment Framework of Infrastructure Systems by Integrating Social Equity to Support Disaster Resilience Decision Making, having been approved in respect to style and intellectual content, is referred to you for judgement.

We have read this dissertation and recommend that it be approved.

Armin Mehrabi

Ioannis Zisis

Xia Jin

Lu Zhang, Major Professor

Date of Defense: November 07, 2022.

The dissertation of Sunil Dhakal is approved.

Dean John L. Volakis College of Computing and Engineering

Andrés Gil Vice President for Research and Economic Development and Dean of the University Graduate School

Florida International University, 2022

#### ACKNOWLEDGMENTS

First, I would like to express my great appreciation to my academic advisor, Dr. Lu Zhang, for her continuous support and endless encouragement during my PhD study. I am thankful for her continued guidance and mentoring in research throughout my PhD journey at Florida International University. I am extremely proud to be one of her students and admire her work and efforts she puts to provide profound knowledge to her students.

I want to convey my grateful thanks to my PhD committee members, Dr. Armin Mehrabi, Dr. Ioanis Zisis, and Dr. Xia Jin for their guidance, encouragement, and wonderful recommendations in improving my doctoral dissertation. I would also like to thank Dr. Xuan Lv for his support and guidance throughout my PhD. His suggestions and efforts have been beneficial in developing some of the chapters of my dissertation.

I would also like to express my gratitude to the Moss Department of Construction Management and Department of Civil and Environmental Engineering for providing me an opportunity to pursue my PhD study at Florida International University, Miami. I would like to express my sincere thanks to all faculty, staff, students, and colleagues at Moss School of Construction, Infrastructure, and Sustainability for their support during my PhD study.

Finally, I am thankful to my wife, Arju Parajuli, for loving me and supporting me in the toughest times. My sincere appreciation to my parents and rest of my family members for their love, encouragement, and support throughout my life.

## ABSTRACT OF THE DISSERTATION

# A RESILIENCE ASSESSMENT FRAMEWORK OF INFRASTRUCTURE SYSTEMS BY INTERGRATING SOCIAL EQUITY TO SUPPORT DISASTER RESILIENCE DECISION MAKING

by

Sunil Dhakal

Florida International University, 2022

Miami, Florida

#### Assistant Professor Lu Zhang, Major Professor

Resilient infrastructure, which better withstands, adapts, and quickly recovers from disasters, can limit disaster impacts, such as disruptions to infrastructure services and time and efforts needed for recovery. However, in the context of a disaster, the impacts or disruptions on the infrastructure are not evenly distributed across different communities. Thus, we need to account for such disparities (or inequalities) when assessing infrastructure resilience.

To address this need, this dissertation presents a new social-welfare-based infrastructure resilience assessment (SW-Infra-RA) model for quantifying the collective resilience of infrastructure serving multiple communities. This model accounts for (1) disaster inequality – the unequal distribution of disaster impacts on infrastructure across different communities, and (2) disaster vulnerability – the disaster impacts on infrastructure of the communities that suffer from the most severe impacts, both of which have impacts on the collective resilience of infrastructure. The proposed model is theoretically grounded on the social welfare theory and social welfare functions. It also leverages studies related to Social Vulnerability Index and the Resilient Triangle framework. The dissertation presents the conceptual notions and mathematical functions of the SW-Infra-RA model. A set of

hypothetical and real case studies were conducted to illustrate the use of the proposed model to assess infrastructure resilience.

The results generated using this model could be utilized by decision makers to better understand the uneven distribution of disaster impacts across communities and identify communities that are severely impacted from a disaster. Such information about inequalities and vulnerabilities of the impacted region could help decision makers prioritize disaster assistance, resources for recovery, and future infrastructure investment toward the vulnerable communities. Overall, the study has the potential to facilitate equitable resilience planning by allowing both decision makers and community personnel to better understand the links between resilience planning and equity in their communities.

v

TABLE OF CONTENTS
TIDLE OF CONTLINE

CHAPTER	PAGE
CHAPTER 1 INTRODUCTION	1
1.1 Background	
1.2 Problem Statement	
1.2.1 Knowledge gaps	
1.3 Research Objectives	5
1.4 Research Methodology	7
1.5 Research Significance	8
1.5.1 Benefits to the Society	8
1.6 Dissertation Organization	9
CHAPTER 2 LITERATURE REVIEW	
2.1 Introduction	
2.2 Interrelationship between Infrastructure Resilience and Social Equity	
2.2.1 Infrastructure Resilience and Social Equity in Disaster Literature	
2.2.2 Social Media Analysis in Disaster Literature	
2.3 Infrastructure Resilience Assessment in Disaster Literature	15
2.3.1 Resilience Triangle Framework	
2.4 Resilience Analysis of Infrastructure in Disasters	
2.5 Social Welfare Functions	
2.6 Social Inequality Measurements	25
CHAPTER 3 UNDERSTANDING INFRASTRUCTURE RESILIENCE, SOCIAL EQUIT AND THEIR INTERRELATIONSHIPS: EXPLORATORY STUDY USING SOCIAL	Υ,
MEDIA IN HURRICANE MICHAEL	
3.1 Introduction	
3.2 Research Context	
3.3 Methodology	
3.3.1 Data Collection Methods	
3.3.2 Data Analysis Methods	
3.4 L Analyzing Polationshing between Infrastructure Posilionee and Twitter Activitie	
3.4.2 Analyzing Relationships between Social Equity and Twitter Activities	75 41 14
3.4.3 Analyzing Relationships between Social Equity and Infrastructure Resilience	
3.4.4 Analyzing Sentiment Indices Infrastructure Resilience and Social Equity	
3.5 Conclusions	54
CHAPTER 4 A SOCIAL WELFARE BASED INFRASTRUCTURE RESILIENCE	
ASSESSMENT FRAMEWORK: TOWARD EQUITABLE RESILIENCE FOR	
INFRASTRUCTURE DEVELOPMENT	
4.1 Introduction	
4.2 Proposed Infrastructure Resilience Evaluation Framework	
4.2.1 Determining Disaster Impacts on Individual Communities	
4.2.2 Modeling Inequality in Disaster Impacts	61
4.2.3 Modeling Vulnerability to Disaster Impacts	63
4.2.4 Measuring Collective Disaster Impacts	65
4.2.5 Assessing Collective Infrastructure Resilience	67

4.3 Case Studies	69
4.3.1 Hypothetical Case Study	69
4.3.2 Hurricane Michael Case Study	74
4.4 Conclusions	80
CHAPTER 5 PROTOTYPE DEVELOPMENT OF DECISION SUPPORT SYSTEM TO	
FACILITATE EOUITABLE INFRASTRUCTURE PLANNING	81
CHAPTER 6 RESILIENCE ANALYSIS OF INFRASTRUCTURE SYSTEMS	
INTEGRATING SOCIAL EQUITY: CASE STUDIES USING HISTORICAL DISASTERS	5 86
6.1 Introduction	86
6.2 Case Study Design	88
6.2.1 Contexts of Cases studies	89
6.2.2 Community Groups under Analysis	91
6.2.3 Case Study Data Collection	92
6.2.4 Case Study Data Analysis	93
6.3 Case Study Results	94
6.3.1 Results of Case Study I	94
6.3.2 Results of Case Study II	95
6.3.3 Results of Case Study III	97
6.3.4 Results of Case Study IV	98
6.4 Results Discussion	100
6.4.1 Analysis of disaster inequality across different communities	100
6.4.2 Analysis of disaster vulnerability across different communities	101
6.4.3 Analysis of collective resilience of infrastructure serving different communities	103
6.5 Conclusions	104
CHAPTER 7 CONCLUSIONS	105
7.1 Summary	105
7.2 Research Contributions	106
7.3 Limitations and Recommendations for Future Research	107
LIST OF REFERENCES	
VITA	130

LIST	OF	TABLES
------	----	--------

TABLE	PAGE
Table 3-1 Infrastructure resilience variables	
Table 3-2 Social equity variables	
Table 3-3 Statistically significant results of correlation analyses between infras resilience variables and Twitter activities	tructure
Table 3-4 Statistically significant results of correlation analyses between social variables and Twitter activities	equity 44
Table 3-5 Correlation coefficients of correlation analyses between social equity varial   Twitter activities	bles and
Table 3-6 Statistically significant results of correlation analyses between social variables and normalized infrastructure resilience variables	equity 49
Table 3-7 Partial list of positive and negative Hurricane Michael related tweets with se scores	ntiment 52
Table 4-1 Functional loss and recovery time of highway infrastructure of City A and C	City B 71
Table 4-2 Results of resilience assessment of highway infrastructure of City A and Ci	ty B74
Table 4-3 Functional loss and recovery time of electric power infrastructure in Hu Michael	ırricane 76
Table 4-4 Results of resilience assessment of electric power infrastructure in Hu Michael.	ırricane 77
Table 6-1 List of disaster, infrastructure system and source of infrastructure damage d	ata92
Table 6-2 Results for case study I	
Table 6-3 Results for case study II	96
Table 6-4 Results for case study III	
Table 6-5 Results for case study IV	

## LIST OF FIGURES

FIGURE	PAGE
Figure 1-1 Overview of research tasks.	8
Figure 2-1 A conceptual diagram of resilience triangle	
Figure 3-1 Ratio indices of the disaster affected counties of Florida.	
Figure 3-2 Line charts showing the relationships between infrastructure resilience varial and Twitter activities (RI and SI)	bles 43
Figure 3-3 Line charts showing relationships between social equity variables and Twi activities (RI and SI)	itter 46
Figure 3-4 Line charts showing relationships between social equity variables and normali infrastructure resilience variables	ized 50
Figure 4-1 A Lorenz curve for the distribution of infrastructure functional loss	
Figure 4-2 A conceptual diagram for a collective resilience triangle (adapted from Brunea al. 2003).	u et 68
Figure 4-3 Lorenz curves for the distribution of disaster impacts in City A due to Hurrican	ie X 72
Figure 4-4 Lorenz curves for the distribution of disaster impacts in City B due to Hurric X	ane
Figure 4-5 Counties of Florida Panhandle region that issued disaster declarations (FE 2018).	MA 75
Figure 4-6 Lorenz curves for the distribution of disaster impacts due to Hurricane Michae	1 78
Figure 5-1 Main modules of the proposed prototype system	81
Figure 5-2 A flowchart of the proposed prototype system	
Figure 5-3 A user interface of the disaster inequality module	
Figure 5-4 A user interface of the collective infrastructure resilience module	

## **CHAPTER 1 INTRODUCTION**

#### 1.1 Background

Infrastructure systems provide essential services including electricity, water supply, communication, and transportation services to communities. With the increase in frequency and intensity of extreme weather events and natural disasters by climate change, these infrastructure systems serving different communities remain at high risk to from disasters. Failures on infrastructure systems due to disaster impacts could result in severe consequences on the regional economic system and affects people's accessibility to infrastructure services. Past experience in disaster have shown that the disaster accounts for huge economic losses mostly in the form of damage to infrastructure systems (UN 2016). To allow infrastructure to resist or absorb disturbance, and remain basic functional and service capacities, investing on and implementing disaster resilience strategies have become a "national imperative" (NRC 2012). Thus, it is crucial that we focus on improving the resilience of our infrastructure systems, so they offer the essential services to society in the aftermath of disruptive events. Government and scientific scholars in the domain of disaster management have made significant contributions on improving infrastructure resilience (Ouyang and Wang 2015, Chang et al. 2013, UN 2016). However, one of the overlooked problems with infrastructure resilience is that a damaged infrastructure due to disaster could result in varying level of disturbances to the community residents. Such damage is typically not evenly distributed across different communities; the low-income and the minority communities are more vulnerable to disaster risks, and they are more likely to struggle more to recover (Emrich et al. 2019). For example, after Hurricane Harvey, more severe flooding damage was found in communities or households with lower income as lower income Americans are more likely to live in neighborhoods r buildings that are more susceptible to flooding or other impacts from storms (Krause and Reeves 2017).

To reduce or eliminate disparities of access to infrastructure due to disasters, the 2030 Global sustainable development agenda of the United Nations highlights the importance of understanding the interlinkages and integrated nature of infrastructure, inequality, and resilience (UN 2016). For example, how does the infrastructure resilience affect social equity? How is social equity integrated into the infrastructure resilience assessment or planning? How to evaluate resilience of infrastructure systems while accounting for social equity? Benchmarking the definitions of social equity in a number of literature (e.g., Emrich et al. 2019), in my dissertation, social equity is defined as equal opportunities and resources provided to different populations through the functions offered by infrastructure. Achieving social equity means reducing or elimination disparate access to goods, services, and amenities among different populations, including the socially vulnerable populations. Socially vulnerable populations include the economically disadvantaged, racial and ethnic minorities, the elderly, the uninsured, the homeless, the disabled, those with chronic health conditions, and those with language barriers (Rao et al. 2109, AJMC 2006). They often have the fewest resources for disaster preparedness, live in disaster prone areas, and lack social capital, political, and economic capital needed to adapt to and recover from disasters (IWR 2016). Multiple studies (e.g., Frigerio et al. 2019, Fatemi et al. 2017, Constible 2018) have shown that socially vulnerable communities experienced more severe disturbances caused by infrastructure damage, which could exacerbate social inequities if not addressed in a timely manner (Fothergill and Peek 2004).

Resilient infrastructure, which is able to withstand disasters and recover quickly, remains vital in offering stabilized essential services (e.g., water, power, communication, transportation) to socially vulnerable populations. Thus, it plays an important role in supporting social equity in disasters (Doorn 2019, UN 2016). A resilient infrastructure has a potential to reduce "disaster-induced poverty" - such as shortage of supplies, inaccessibility to goods or services. In addition, implementing the disaster resilient strategies in the infrastructure planning could facilitate

infrastructure improvement and potentially led to economic growth. This will eventually increase productivity and employment opportunities, reduce poverty and contribute to social equity. It also has better capabilities of offering "disaster equity" by reducing the vulnerabilities and disparities in different regions and communities by allowing less infrastructure damage and access to adequate resources for recovery during disasters. Despite such linkage between infrastructure resilience and social equity, there is a lack of study that provides more explicit understanding of the complex interrelationship between infrastructure resilience and social equity. Without such understanding, our "next generation infrastructure" cannot be planned, constructed and operated with great human awareness and social adaptability.

#### **1.2 Problem Statement**

Over the last few decades, many resilience assessment methods and models have been proposed to assess the resilience of infrastructure systems. However, there is still a lack of assessment model that incorporates the quantification of inequalities and vulnerabilities with it. One of the crucial challenges in the domain of disaster resilience is to measure such inequalities and vulnerabilities in the context of infrastructure resilience. This knowledge gap creates the need for a change in resilience assessment approaches. It remains crucial that we develop a better understanding of the relationships between equity and infrastructure resilience as well as improve the assessment of infrastructure resilience by accounting for equity. Based on a comprehensive literature review in the domain of infrastructure, resilience, and social equity, a number of knowledge gaps were identified. These knowledge gaps will be discussed in detail in the following sections.

#### 1.2.1 Knowledge gaps

 Lack of theoretical and explicit understanding on how infrastructure resilience affects social equity. Scholars in the domain of built environments, transportation systems, design and manufacturing, logistics systems, and systems operation and management have contributed to understanding, assessing, and enhancing the resilience of infrastructure systems (e.g., Aydin et al. 2018, Zhang et al. 2018, Heinimann and Hatfield 2017). Research in these domains focused on different aspects of infrastructure resilience, such as structural integrity (e.g., Chopra et al. 2016), system reliability (e.g., Nateghi 2018), system recovery (e.g., Aydin et al. 2018), resource allocation (e.g., Zhang et al. 2018), and system resilience measurement (e.g., Heinimann and Hatfield 2017). Social equity, on the other hand, has been extensively studied by social scientists in the domain of social science, political science, psychology, and anthropology. A long history of social equity research in the disaster domain shows that multiple social characteristics are associated with disparate exposures and impacts in disasters -including race, income, age, disability, and language proficiency (Domingue and Emrich 2019, Thomas et al. 2013). Most literature in social equity, in the domain of disasters, focused on studying social vulnerability (e.g., Fatemi et al. 2017, Frigerio et al. 2019, Cutter et al. 2003), equitable recovery (e.g., Emrich et al. 2019), social justice (Shively 2017), and social resilience (e.g., Comes et al. 2019). Collectively, the research efforts in the domain of engineering and social science have offered valuable contributions to infrastructure resilience and social equity in disasters, respectively. However, there is still a lack of study that integrates both infrastructure resilience and social equity, which can offer a holistic understanding of the interrelationships between them to facilitate better infrastructure decision making that account for social equity impacts.

(2) There is a lack of study on the methods for measuring social equity in the context of infrastructure resilience. Many studies in the domain of disasters have focused on assessing social equity in disaster recovery or analyzing the social vulnerability across different regions (Domingue and Emrich 2019, Emrich et al. 2019, Doorn et al. 2019). These studies have emphasized the need to pay special attention to socially vulnerable populations and facilitate social equity across different phases of disaster. Although these studies contribute to better understanding of social inequality in disasters and support decision makers and practitioners to implement policies to maintain equity in the community, there is a lack of theoretically grounded quantitative method to measure social equity. To improve the collective resilience of infrastructure systems of the communities, it is crucial that we understand the unequal distribution of disaster impacts to infrastructure systems across different communities and potential severe impacts to infrastructure systems experienced by the most vulnerable communities.

(3) There is a lack of study that quantitatively integrates social equity into infrastructure resilience assessment framework. Scholars in the disaster resilience domain have proposed different infrastructure resilience assessment framework for quantifying the resilience of infrastructure systems (Liu et al. 2019, Panteli et al. 2017). In addition, studies in the infrastructure resilience domain have developed multiple resilience metrics to measure the resilience of infrastructure, such as Resilience Star developed by Department of Homeland Security (Kangior 2013), Resilience-based Earthquake Design Initiative (REDi<sup>TM</sup>) rating system (ARUP 2020), RELi rating system (GBCI 2020). Most of the literature in the field of resilience evaluation primarily focuses on evaluating the performance of whole system (e.g., Meerow 2019, Comes et al. 2019), while lacking consideration of social equity conditions. Thus, there is a need to integrate equity and vulnerability into the resilience evaluation framework to develop a better resilience metrics for infrastructure systems.

#### **1.3 Research Objectives**

The main goal of this research study is to analyze the interrelationships between infrastructure resilience and social equity and establish an infrastructure resilience assessment framework that accounts for the unequal distribution of disaster impacts on infrastructure serving different

communities and potentially severe impacts on infrastructure of the vulnerable communities. In order to achieve this goal, the research objectives along with the research questions are formulated: Objective 1:

To understand the interrelationships between social equity and infrastructure resilience in the context of a disaster.

Associated Research Questions:

RQ 1: How are infrastructure resilience conditions in the disaster-affected communities reflected by Twitter activities?

RQ 2: How are social equity conditions in the disaster-affected communities reflected by Twitter activities?

RQ 3: Do social equity characteristics of communities have impacts on the infrastructure resilience conditions of the communities?

Objective 2

To develop a new infrastructure resilience assessment framework by integrating equity into the resilience assessment framework.

Associated Research Questions:

RQ 1: How to quantitatively measure the unequal distribution of disaster impacts on infrastructure across different communities?

RQ 2: How to quantitatively measure the potentially severe disaster impacts on infrastructure of the vulnerable communities?

RQ 3: How to mathematically integrate the disparity and vulnerability in disaster impacts with infrastructure resilience assessment?

Objective 3

To validate the proposed resilience assessment framework using real case studies and develop a decision support system that assesses equity-integrated infrastructure resilience automatically.

Associated Research Questions:

RQ 1: How do communities with different characteristics (e.g., coastal vs inland, urban vs rural, more vulnerable vs less vulnerable) compare against each other in terms of disaster inequality? RQ 2: How do communities with different characteristics compare against each other in terms of disaster vulnerability?

RQ 3: How do communities with different characteristics compare against each other in terms of collective resilience?

To limit the scope of the research work, the dissertation focused on the Florida communities, infrastructure systems that serve Florida communities, and disasters that threatened Florida communities.

#### 1.4 Research Methodology

In order to achieve the above-mentioned research objectives, the research methodology is divided into four main research tasks as follows: (a) understanding the social equity, infrastructure resilience and their interrelationships in the context of a disaster, (b) develop a new infrastructure resilience assessment framework while integrating equity with infrastructure resilience assessment, (c) develop a prototype of a decision support system based on the proposed model for facilitating automatic infrastructure resilience assessment, and (d) validate the proposed assessment framework through multiple case studies. Figure 1-1 shows overview of the research framework.



Figure 1-1 Overview of research tasks.

#### 1.5 Research Significance

This research study contributed to (1) advancing the fundamental understanding of the interrelationships between infrastructure resilience and social equity, (2) developing new methods on quantifying the unequal distributions of disaster impacts on infrastructure across different communities and evaluating severe impacts on infrastructure of vulnerable communities, and (3) formulating a theory-based infrastructure resilience assessment model that integrates disaster inequalities and vulnerabilities.

#### 1.5.1 Benefits to the Society

The SW-Infra-RA model could allow the decision makers to quantitatively assess the resilience of infrastructure system and facilitate equity-incorporated decision making. The results generated by the model can be utilized to understand the inequalities that exists during disasters and identify communities that are more vulnerable to such disasters. This will help decision makers, emergency managers, and infrastructure planners prioritize disaster aids and funds, resources for recovery, and

future infrastructure investments and planning efforts to communities that are more vulnerable to the disasters. With the help of the model, decision makers can better understand the importance of considering inequalities and vulnerabilities in disasters and predict the potential consequences of future disasters. In addition, this model could potentially allow community residents and disaster practitioners to understand the relationship between resilience planning and social equity and help facilitate equitable resilience planning of infrastructure systems.

#### **1.6 Dissertation Organization**

This dissertation is organized into six chapters. Each of these chapter consists of its own introduction, methodology, results and analysis, and conclusions. Chapter 1 introduces the research background, knowledge gaps, and problem statement. This chapter also introduces research objectives along with research questions that need to be addressed, methodology adopted to achieve these objectives, and research significance. Chapter 2 provides a comprehensive literature review for all the research tasks. Chapter 3 presents the research task about understanding the interrelationships between social equity and infrastructure resilience in the context of a disaster. This chapter has been published in Dhakal et al. (2021). Chapter 4 presents the research task about developing a new social-welfare-based infrastructure resilience assessment (SW-Infra-RA) model that accounts for disaster inequality and vulnerability. This chapter has been published in Dhakal and Zhang (2022). Chapter 5 presents the research task about the development of a prototype decision support system based on the SW-Infra-RA model. Chapter 6 presents the research task about the resilience analysis of different infrastructure systems using four case studies. Chapter 7 provides a summary of the overall research and conclusions of this dissertation. It also presents the contribution of the research study, limitations, and recommendation for future works.

## **CHAPTER 2 LITERATURE REVIEW**

#### 2.1 Introduction

The main objective of Chapter 2 is to review the literature related to infrastructure resilience, social equity, and resilience assessment of infrastructure systems. This comprehensive literature review provided a detailed overview of different conceptual notions related to the dissertation topic, explored knowledge gaps, and identified parameters required for formulating the infrastructure resilience assessment framework. The literature review is divided into five parts: (1) review the research about infrastructure resilience, social equity, and their interrelationships, (2) review the research about resilience assessment methods and frameworks, (3) review the research about resilience analysis of infrastructure systems, (4) review the research about social welfare theory and social welfare functions, and (5) review the research about social inequality measurements.

## 2.2 Interrelationship between Infrastructure Resilience and Social Equity

#### 2.2.1 Infrastructure Resilience and Social Equity in Disaster Literature

Over the last decade, disasters caused by natural hazards have resulted in over \$900 billion in economic losses worldwide, mostly in the form of damage to infrastructure (UN 2016). Developing resilient infrastructure systems becomes a "national imperative" to address the threat caused by increasingly frequent and intensive disasters (Chopra et al. 2016). In addition, there are still significant disparities in access to infrastructure. For example, over 1.1 billion people still have no access to electricity worldwide, and about one-third of the world's population is not served by all-weather roads (Badré 2015). Minimizing or closing these disparities would require significant investment and development on infrastructure in a way that not only enhances its resilience but also reduces inequality of the society. While infrastructure resilience could potentially lead to better outcomes of social equity. Disasters cause disproportionate impacts to communities through their

impacts on infrastructure, which offers essential services (e.g., water supply, energy, communication, and transportation) to meet basic needs of disaster victims (Lynn et al. 2011). Multiple studies (e.g., Frigerio et al. 2019, Fatemi et al. 2017, Constible 2018) have shown that socially vulnerable communities experienced more severe disturbance caused by infrastructure damage, which could exacerbate social inequities if not addressed in a timely manner (Fothergill and Peek 2004). Resilient infrastructure, which poetically has less functional damage and/or is able to refunctionalize rapidly, may close the inequality gaps across different communities; it plays an important role in catering the necessities of all communities (Braese et al 2019).

On one hand, the concept of infrastructure resilience has drawn significant attention among researchers in the disaster domain (Karamouz et al. 2019). Infrastructure resilience is defined as the ability of infrastructure to withstand, adapt, and quickly recover from the effects of disasters. The concept of resilience, originally, was used to indicate the capacity of a system to return to its original functional level after disruptive events (Rus et al. 2018). It was first introduced by Holling (1973) to define the persistence of relationships within a natural ecosystem and the ability of the system to absorb changes (Holling 1973). It was then widely adapted into different scientific fields, such as engineering, social science, material science, and economics, etc.

Over the last few decades, engineers in the domains of built environments, transportation systems, design and manufacturing, logistic systems, and systems operation and management have contributed to understanding, assessing, and enhancing the resilience of infrastructure systems (e.g., Aydin et al 2018, Zhang et al. 2018, Yodo and Wang 2016, Hosseini and Barker 2016, Heinimann and Hatfield 2017). Research in these domains focused on different aspects of infrastructure resilience, such as structural integrity (e.g., Chopra et al. 2016, Zhao et al. 2015), system reliability (e.g., Nateghi 2018), system recovery (e.g., Aydin et al. 2018, Croope and McNeil 2011), resource allocation (e.g., Zhang et al. 2018, MacKenzie and Zobel 2016), and system

resilience measurement (e.g., Heinimann and Hatifield 2017, Yodo and Wang 2016). For example, Chopra and colleagues (2016) developed a multi-pronged framework that analyzes information on the network structure, spatial location, passenger flow, and structural and functional vulnerabilities for improving the resilience of the London Metro system. Nateghi (2018) proposed a predictive tool to assess various investment strategies for enhancing the resilience of electric power systems in hurricanes. Aydin and colleagues (2018) proposed a methodology that evaluates road recovery strategies for restoring the services after blockage due to natural disasters. Zhang and colleagues (2018) proposed a numerical modeling-based approach for allocating restoration resources that could enhance the resilience of infrastructure systems. Yodo and Wang (2016) explored and evaluated the challenges of incorporating resilience into engineering design, which contributes to the development of an engineering resilience analysis framework.

Social equity, on the other hand, has been extensively studied by social scientists in the domains of social science, political science, psychology, and anthropology. In the context of disasters, allowing all disaster-affected individuals, including the socially vulnerable populations, to have equal access to resource distributions and opportunities is the key to achieve equitable resilience (Emrich et al. 2019). A long history of social equity research in the disaster domain shows that multiple social characteristics are associated with disparate exposures and impacts in disasters—including race, income, age, disability, and language proficiency (Domingue and Emrich 2019, Thomas et al. 2013). Most literature on social equity, in the domain of disasters, focused on studying social vulnerability (e.g., Fatemi et al. 2017, Frigerio et al. 2019, Cutter et al. 2003), equitable recovery (e.g., Emrich et al. 2019), social justice (Shively 2017, Gil-Rivas and Kilmer 2016), and social resilience (e.g., Comes et al. 2019, Kim et al. 2018). For example, Cutter and colleagues (2003) studied socioeconomic and demographic conditions of different counties and developed social vulnerability index to encapsulate the socioeconomic conditions associated with disaster inequalities. Emrich and colleagues (2019) explored how social characteristics influenced the

equitable disaster recovery process for the 2015 South Carolina floods. Gil-Rivas and Kilmer (2016) proposed an ecological framework that accounts for social justice, empowerment, and diversity in building community resilience. Comes and colleagues (2019) highlighted the role of new information and communication technologies for improving social resilience during crisis across three different eras (1991-2005, 2005-2015, and 2016-onwards).

Collectively, the research efforts in the domains of engineering and social science have offered valuable contributions to infrastructure resilience and social equity in disasters, respectively; research in the engineering domain advances the design, operation, and management of infrastructure systems in ways that improve their capabilities to resist, respond, and adapt to disasters; while research in the social science domain leads to the important recognition and understanding of the disproportionate impacts of disasters on communities. However, researchers focusing on each of these fields are typically from different research backgrounds, making links between infrastructure resilience and social equity less commonly studied than any of the two areas taken in isolation. There is still limited convergence research that integrates both social equity and infrastructure resilience to support better infrastructure decision making that accounts for social impacts.

#### 2.2.2 Social Media Analysis in Disaster Literature

Social media is a collection of platforms that allow users to create public or semi-public profiles, generate multimedia contents, connect with other users, and share contents, opinions, insights, and perspectives in real time (Houston et al. 2015). Social media is characterized as a low-cost, easy-to-use, scalable, relatively reliable multimedia network that allows for real-time information sharing and exchange (Mills et al. 2009). In addition, with the prevalence of Global Positioning System (GPS)-enabled personal mobile devices, every social media user could become part of a

location-enabled large sensor network. Thus, compared with traditional data sources, social media data is more spatially comprehensive and relatively rich in offering situational awareness information (Li et al. 2019). Over the last decade, social media has gained immense popularity for understanding information sharing and exchange in different domains, such as healthcare (e.g., Surani et al. 2017), emergency management (e.g., Harrison and Johnson 2019), marketing (e.g., Shareef et al.2019), politics (e.g., Kwak et al. 2018), and entertainment (e.g., Khan 2017).

In the disaster domain, social media has been proved to be a good alternative to traditional data sources (Beigi et al. 2016, Cobo et al. 2015, Lindsay 2011). The massive data generated from social media can be used to analyze human activities in different spatiotemporal dimensions and provide insights on disaster-related knowledge. Researchers in the disaster field have worked on analyzing social media activities to address a variety of issues, such as damage assessment (e.g., Resch et al. 2018, Chen et al. 2020), disparities of disaster impacts (e.g., Zou et al. 2018b), crisis communication (e.g., Roshan et al. 2016), disaster response and recovery (e.g., Young et al. 2020), and real-time disaster mapping (e.g., Li et al. 2018). For example, Resch and colleagues (2018) conducted a spatiotemporal analysis of social media data using machine learning techniques to analyze the regions with significant damage due to disasters. Chen and colleagues (2020) employed a systematic approach to identify and assess the damage on highways using social media in the context of Hurricane Harvey. Zou and colleagues (2018b) studied the social and geographical disparities that existed in the twitter activities during Hurricane Sandy. Roshan and colleagues (2016) analyzed the use of social media for communication among different organizations in the time of crisis. Young and colleagues (2020) studied social media and its potential use for emergency communication during the response and recovery phases of disasters. Li and colleagues (2018) proposed a novel approach for mapping the flood in real time using social media data.

The previous research has collectively provided important contributions to the utilization of social media data in advancing disaster resilience knowledge. However, existing research also suggests that due to the many inherent issues of social media data, such as false information, lack of validation, malicious use, using social media data alone to draw scientific conclusions or generate new knowledge is still challenging (Li et al. 2018, Zou et al. 2018a). There is a need to integrate social media data with traditional data to provide informative analysis results, and more research is necessary to address the question of synthesizing social media data with other sources of data to offer meaningful knowledge that supports disaster resilience (Zou et al. 2018a).

#### 2.3 Infrastructure Resilience Assessment in Disaster Literature

Over the last two decades, the concept of infrastructure resilience has gained significant attention from scientific scholars and researchers around the world (Karamoutz et al. 2019, Cimellaro et al. 2010). The concept of "resilience" was first introduced by Holling (1973) to describe the "persistence of relationships within a system" and the ability of the system to "absorb changes of state variables, driving variables, and parameters, and still persist" (Holling 1973). Holling (1996) also explained the difference between engineering resilience and ecological resilience. Ecological resilience measures how a system can persist by absorbing changes and disturbances, while engineering resilience measures the capacity of the system to recovery to its original functional level after a disturbance. Integrating these definitions, infrastructure resilience is typically defined as the ability of infrastructure to anticipate and absorb the shock, adapt to, and quickly recover to its original functional level (Berkeley et al. 2010).

Over the years, many resilience assessment frameworks have been proposed to assess the resilience of different types of infrastructure, such as transportation infrastructure (e.g., Tonn et al. 2020), electric power systems (e.g., Hossain et al. 2019), water and sanitation infrastructure (e.g., Assad et al. 2019), and telecommunication infrastructure (e.g., Mawgaud et al. 2021). These studies used different approaches to assess infrastructure resilience, including simulation-based approaches

(e.g., Hossain et al. 2019, Hosseini and Barker 2016, Lam and Tai 2018), mathematical approaches (e.g., Cimellaro et al. 2010, Shin et al. 2018, Bruneau et. al 2003), index-based approaches (e.g., Rehak et al. 2019, Petit et al. 2012, Fisher and Norman 2010), and data-driven approaches (e.g., MacKenzie and Barker 2020, Zhu et al. 2017).

Simulation-based approaches were mostly employed in resilience assessment of system networks, such as water supply networks (e.g., Assad et al. 2019), electric grid networks (e.g., Hossain et al. 2019), and transportation networks (e.g., Hosseini and Barker 2016). In these studies, Bayesian networks, Monte Carlo simulation, and Fuzzy models were commonly used for the analysis. For example, Hossain et al. (2019) employed Bayesian networks to quantitatively assess the resilience of electric infrastructure systems. Similarly, Hosseini and Barker (2016) built a resilience assessment framework to quantify the resilience capacity of an inland waterway network using Bayesian networks. Nogal et al. (2017) proposed a resilience assessment framework to estimate the resilience of a transportation network impacted by extreme events using the Monte Carlo simulation method. Lam and Tai (2018) used a fuzzy modeling approach to model the interdependencies between entities in infrastructure networks by simulating the effects of disruptions.

Using mathematical approaches, the resilience of infrastructure can be assessed through mathematical structures, notions, or equations. The mathematical approaches can be classified into deterministic (e.g., Bruneau et al. 2003, Cimellaro et al. 2010) and probabilistic (e.g., Decò et al. 2013, Nogal et al. 2017) approaches. The deterministic approach utilizes the value of the input parameters to obtain a precise outcome without accounting for uncertainties. In contrast, the probabilistic approach can model the uncertainties that exist in the inputs of metrics to obtain the distributions of infrastructure failure and recovery (Mottahedi et al. 2021). For example, Cimellaro et al. (2010) proposed a comprehensive conceptual model that includes a loss estimation model and

a recovery model to quantitatively assess the seismic resilience of a network of health care facilities. Decò et al. (2013) used a probabilistic approach to assess the seismic resilience of bridges.

Using index-based approaches, a resilience index is developed by identifying and aggregating a set of indicators that represent the characteristics of infrastructure resilience. The resilience index can then be used to compare or rank the resilience of several infrastructure alternatives by collecting the data of each infrastructure alternative. To frame the resilience index, both qualitative (measured with ordinal or nominal scales) and quantitative indicators (measured with interval or ratio scales) can be used (Cardoni et al. 2020). Multiple resilience indexes have been developed over the years, such as the Resilience Star developed by the Department of Homeland Security (Kangior 2013), the U.S. Resiliency Council (USRC) Building Rating System (USRC 2021), the Resilience-based Earthquake Design Initiative (REDi<sup>TM</sup>) rating system (ARUP 2021), and the Resilience Action List (RELi) rating system (GBCI 2021). Similarly, many researchers have used the index-based approaches to develop different infrastructure resilience assessment frameworks. For example, Yang et al. (2018) developed the Resilience Index Considering Duration of events (RICD), which assesses the resilience of power transmission systems under typhoon weather. Argyroudis et al. (2020) built a cost-based resilience index that quantifies the seismic resilience of bridges. Cardoni et al. (2020) developed the Power Resilience Index (PRI) that assesses the seismic resilience of urban electric power distribution systems.

Data-driven approaches refer to those methods that rely on collecting, analyzing, and interpreting data to derive insight, knowledge, or solutions. This approach can be used to develop new models to calibrate and reduce uncertainties when assessing infrastructure resilience (Argyroudis et al. 2021). By deriving knowledge from a large amount of data, the data-driven approach may provide a high level of reliability that cannot be achieved through other conventional scientific approaches (Maass et al. 2018). In the last decade, with the advancement on data analytics techniques, there has been a growing tendency of adopting data-driven approaches for resilience

assessment. For example, Argyroudis et al. (2021) proposed a data-driven resilience assessment framework for critical transportation infrastructure that is exposed to multiple hazards by interactively analyzing multiscale monitoring data (e.g., terrestrial data, airborne data), crowd data, and environmental measurements. Chandramouleeswaran and Tran (2018) used a data-driven approach for quantitatively assessing the resilience of air transportation networks using publicly available data (e.g., total cancellation flights, average flight delay).

The existing research has offered valuable contributions to advance the understanding and methods of assessing infrastructure resilience. However, one of the major concerns that lie behind the demand for better resilience assessment is the need to pay more attention to equity and vulnerability in resilience assessment (Meerow et al. 2019, Meerow and Newell 2019). Much of the resilience assessment literature focuses on evaluating the performance of a whole (e.g., a complete infrastructure network) while lacking consideration of the inequalities and trade-offs among different parts that compose the whole. Some resilience assessment studies (e.g., RF 2021, GBCI 2021) proposed to integrate equity as one dimension or a characteristic of resilience. These studies tend to mix the conceptualizations of equity and resilience and simplify their relationships. Although disaster resilience and equity are interconnected, they are not the same. Integrating equity with resilience requires us to explicitly assess the unequal distributions of disaster impacts on various communities and evaluate the different levels of vulnerability that these communities face (Wescoat et al. 2018). There is, thus, a need to develop a new resilience assessment framework that assesses the collective infrastructure resilience while accounting for the disparities among the communities and potentially severe impacts on infrastructure of vulnerable communities.

#### 2.3.1 Resilience Triangle Framework

The Resilience Triangle framework was based on the work by Bruneau et al. (2003), who defined and quantitatively measured the seismic resilience of communities. According to Bruneau et al. (2003), a resilient system has three key characteristics: (1) reduced failure probabilities, (2) reduced consequences from the failures, and (3) reduced time to recovery. They then proposed to measure the resilience of a community by defining and measuring the area of a resilience triangle (Figure 2-1). In the resilience triangle, the vertical axis represents the quality of infrastructure in a community [Q(t)], which varies with time. Q(t) ranges from 0% to 100%, where 100% represents no degradation in infrastructure quality or service, and 0% means no infrastructure service is available. An earthquake occurring at time  $t_0$  would cause damage to infrastructure so that the quality of infrastructure service is immediately reduced. Restoration of infrastructure is a process that takes time, and the quality of infrastructure gradually increases as the restoration process goes on. The infrastructure is completely recovered to its original functional level at time  $t_t$ . Therefore, the community loss of resilience is determined by aggregating the degradation of the quality of infrastructure over the total recovery time ( $t_t$  - $t_0$ ).



Figure 2-1 A conceptual diagram of resilience triangle

Over the last two decades, many research studies have been conducted to apply or adapt Bruneau et al. (2003)'s framework in assessing the resilience of various types of infrastructure, such as healthcare facilities (Shang et al. 2020), electric power systems (Ouyang and Duenas-Osorio 2014) and transportation infrastructure (Argyroudis et al. 2020). The framework has also been adapted to analyze disruptions caused by disasters other than earthquakes, such as hurricanes (e.g., Ouyang and Duenas-Osorio 2014) and flooding (e.g., Zamanian et al. 2020).

## 2.4 Resilience Analysis of Infrastructure in Disasters

Over the last decade, changing climatic conditions and frequent extreme weather events have caused a detrimental effect on infrastructure serving different communities in the United States. According to the National Oceanic and Atmospheric Administration (NOAA), there were approximately 323 extreme weather events and other natural disasters since 1980 in which the total damage cost of these events exceeded 2.2 trillion US dollars (NOAA 2022). To mitigate the disaster impacts on infrastructure, researchers and practitioners have been championing for more resilient infrastructure over the last several decades. However, these extreme weather events and natural disasters cause disproportionate impacts on infrastructure serving various communities, which may be linked to the variations in the quality of infrastructure serving communities with different characteristics (e.g., spatial, demographic, and socioeconomic statuses). For example, research shows that socially vulnerable communities bear more than their fair share of the physical and economic burden (e.g., physical damage, service disruptions, economic losses) caused by disasters (Emrich et al. 2019, Domingue and Emrich et al. 2019, SAMSHA 2017). These communities typically lack knowledge and resources in disaster risk reduction, mitigation, and infrastructure resilience planning. They are found to be neglected or left behind in infrastructure planning due to discriminatory policies, practices, and biases within infrastructure planning (NASEM 2022). Thus, it remains crucial that we identify the fundamental disaster inequalities and vulnerabilities across different communities when assessing the resilience of infrastructure.

The concept of resilience has gained significant attentions in disaster risk research and applications over the last few decades. Improving the resilience of infrastructure can enhance the capabilities of

infrastructure in resisting, adapting to, and quickly recovering from extreme events. Systematically analyzing the resilience of infrastructure systems and identifying those communities with more vulnerable infrastructure is a first step toward building a universally resilient community. Over the years, extensive studies (e.g., Diao et al. 2016, Shafieezadeh and Burden 2014, Klise et al. 2017, Espinoza et al. 2016, Wei et al. 2020, Zhu et al. 2016) have been conducted to analyze the resilience of various infrastructure, such as water infrastructure (e.g., Diao et al. 2016, Klise et al. 2017), transportation infrastructure (e.g., Zhu et al. 2016, Shafieezadeh and Burden 2014), and electric power system (e.g., Espinoza et al. 2016, Wei et al. 2020). For example, Klise et al. (2017) conducted the resilience analysis of a water distribution system after an earthquake. Zhu et al. (2016) performed an assessment of the resilience of subway and road networks in Hurricanes Sandy and Irene. Espinoza et al. (2016) assessed the resilience of Great Britain's electric power system from the impacts of floods and windstorms. These studies offered valuable contribution to advance the understanding and analyses of infrastructure resilience. However, these studies have not accounted for disparities in disaster impacts on infrastructure serving various communities.

Over the years, scholars (e.g., Yabe and Ukkusuri 2020, Emrich et al. 2019, Mitsova et al. 2018, Zou et al. 2018b, Cutter et al. 2003) in the domain of disasters have highlighted the importance of assessing disparities in disaster impacts to better understand inequities in disaster contexts. Studies on disaster disparity analyses aim to understand unique characteristics and challenges that exist in the various communities and recommend strategies for resilience planning and investment aligned with the severity of needs. Analyzing the disparities of disaster impacts across various communities can be helpful to identify those communities that are the most vulnerable in disasters and facilitate equitable resilience planning in the future. Extensive studies (e.g., Yabe and Ukkusuri 2020, Emrich et al. 2019, Domingue and Emrich 2019, Coleman et al. 2020, Zou et al. 2018b) have been conducted to analyze disparities of disaster management efforts of or disaster impacts on communities or population groups with various characteristics (e.g., income, race, age). For example, Yabe and Ukkusuri (2020) quantified the effects of income inequality on evacuation before disasters and re-entry after disasters, and they found that evacuees from higher income communities were more likely to evacuate from the impacted areas. Emrich et al. (2019) analyzed how social characteristics influenced the federal disaster recovery fund allocation following the 2015 South Carolina floods. Coleman et al. (2020) studied disparities in hardships experienced by different communities due to infrastructure service disruption caused by Hurricane Harvey. Zou et al. (2018b) found social and geographical disparities in social media use during Hurricane Harvey. These studies offer valuable insight and contributions toward better understanding of disparities or inequalities in disasters. However, they have not integrated such disparities with infrastructure resilience assessment. There is also a lack of research on how the integration of disaster inequalities would impact the collective resilience of infrastructure. Furthermore, these studies have not comprehensively studied how the various characteristics of communities (e.g., coastal, inland, urban, rural, more vulnerable and less vulnerable) would impact the level of disaster inequalities and/or the collective resilience of their infrastructure.

#### 2.5 Social Welfare Functions

Welfare economics is the study of how the distribution of resources and goods impacts social welfare; it evaluates well-being (welfare) at the aggregate level (Deardorff 2016). In welfare economics, several functions were proposed to evaluate or compare alternative social states (e.g., income distributions, life expectancy, literacy rate), and these functions are called social welfare functions (SWFs) (Weymark 2016). A SWF can be thus defined as a function that measures or ranks the collective welfare of the society in different social states (Arrow 1963). It can be used to determine the optimal distribution of well-being among individuals to achieve the maximum wellbeing for the whole society (Arrow 1963). In a SWF, well-being is generally expressed in terms of utilities (e.g., incomes, benefits) or preferences.

In welfare economics literature, SWFs can be generally classified into (1) the Bergson-Samuelson SWFs, (2) the Arrow SWFs, and (3) Cardinal SWFs. The Bergson-Samuelson SWFs determine the social preference (in the form of social ordering or ranking) of alternative social states (Weymark 2016) based on individual utilities (e.g., income, life expectancy). Through the functions, the individual utilities in alternative social states are first determined and further aggregated to determine the collective social preference. With the Arrow SWFs, the social preference of alternative social states is determined as a function of individual preferences (Weymark 2016). Unlike the Bergson-Samuelson SWFs, the Arrow SWFs only use information about individual preferences to determine the social preference.

Cardinal SWFs, on the other hand, are functions that determine the collective welfare (in the form of numerical value) based on individual utilities. They do not necessarily require comparisons among individual utilities in alternative social states, and they yield a numerical representation of the collective welfare for each social state. Some of the Cardinal SWFs found in the literature include Utilitarian SWF (Harsanyi 1955), Rawlsian SWF (Rawls 1971), Bernouli-Nash SWF (Jagtenberg 2017), Sen's SWF (Sen 1997), and Atkinson and Brandoloni SWF (Atkinson and Brandoloni 2010). The Utilitarian SWF measures the social welfare as the average welfare of the individuals in the society. With the Utilitarian SWF, the collective social welfare of a society increases if the welfare of any individual increases and none decreases, with everyone indifferent (Harsanyi 1955, Schneider and Kim 2020). This function does not account for the equality of welfare (e.g., fair distributions of income) among the individuals in a society. With the Rawlsian SWF, the welfare of the society is determined by the welfare of the individuals with the lowest welfare in a society (Rawls 1971). According to the Rawlsian SWF, the social welfare increases if the welfare of the poorest individuals increases; it does not consider the welfare of other individuals in the society. Similar to Utilitarian SWF, the Rawlsian SWF does not consider equality in welfare distributions in a society. The Bernoulli-Nash SWF, in general, can be seen as the

mixture of the Rawlsian and the Utilitarian SWFs. With the Bernoulli-Nash SWF, the collective social welfare is calculated as the product of all individual welfare (Jagtenberg 2017).

To account for inequality in welfare distributions, Sen (1997) proposed a SWF (Sens's SWF) that accounts for unequal distributions of welfare across the individuals in a society. In Sen's SWF, a Gini coefficient is used to measure welfare inequality. The Sen's SWF determines the social welfare as the product between average welfare of all individuals and an inequality indicator. According to Sen's SWF, the social welfare increases if the fairness in distributing the welfare increases. However, with Sen's SWF, it is possible that the total amount of welfare increases at the expense of increased equality and reduced average welfare. In another word, the social welfare could increase by allowing all individuals to be equally poorer (Mostafa and El-Gohary 2014). Thus, to account for poverty in social welfare, a poverty line is defined based on the minimum amount of income an individual or a household needs to meet their basic needs (Callan and Nolan 1991). The individual or household whose income falls below the poverty line is considered as being poor. Leveraging the poverty line, Atkinson and Brandoloni (2010) proposed a SWF that accounts for the poorest individuals with the minimum welfare in a society.

Over the year, scholars in the domains of social science and economics have used the SWFs to solve various problems, such as reducing health inequalities (Dolan and Robinson 2001), assessing climate policies (Füssel 2006), and improving cost benefit analysis (Adler 2017). In recent years, the SWFs have been further adapted to address issues in other domains, such as transportation, architecture, engineering, and construction. For example, Zhang and Sanake (2020) proposed a social welfare-based group comfort analysis model to measure the collective comfort level of a group of individuals in the indoor environments. Kinjo and Ebina (2017) developed a mathematical model based on Utilitarian SWF and Nash SWF to determine autonomous vehicle (AV) driving behaviors by evaluating individual utilities of passengers inside the AV and pedestrians on a street. Mostafa and El-Gohary (2014) presented a social welfare-based

sustainability benefit analysis model that evaluates the distribution of benefits of infrastructure project alternatives to their stakeholders by accounting for both equality and poverty in benefit distributions.

#### 2.6 Social Inequality Measurements

Inequality refers to an absence of equal distributions of goods, services, opportunities, rights, and/or dignity (UNCTAD 2021). Inequality is often measured by assigning a certain value to a specific distribution in order to facilitate direct and objective comparisons across different distributions (UNCTAD 2021). There are multiple methods to measure inequality, and these methods can be categorized as using "ratios" or using "indices".

Measuring inequality through "ratios" is a relatively easy and straightforward method. The most commonly used ratios for measuring inequality are 20/20 ratio and Palma ratio. 20/20 ratio represents the ratio of average income of the richest 20 percent of the population to the average income of the poorest 20 percent of population (Afonso et al. 2015, UNCTAD 2021). Palma ratio is defined as the ratio of total income of the richest 10 percent of households to the poorest 40 percent of households (Afonso et al. 2015, UNCTAD 2021). Although ratios are relatively easy to understand, these methods do not measure how social welfare (e.g., income) is equally or unequally distributed across the population. For example, they do not consider the welfare (e.g., income) distributions within the highest and lowest percentiles of population (Trapeznikova 2019).

As compared to "ratios", "indices" are more commonly used to measure inequality. Some of the popular indices are Atkinson's index (Afonso et al. 2015), Hoover index (Hoover 1941), Theil index (Theil 1967), and Gini index (Trapeznikova 2019). Atkinson's index is a welfare-based measure of inequality, and it represents the percentage of total income that could be sacrificed to have more equal shares of income among individuals without reducing social welfare (Afonso et al. 2015). Hoover index, also known as Schutz index, defines inequality as the share of total income that needs to be redistributed from the population with income above mean to the those with income below mean to achieve income equality (Afonso et al. 2015). A higher value of Hoover index indicates a higher level of inequality, and more redistributions are needed to achieve equality. Theil index belongs to general entropy (GE) measures; it measures an entropic "distance" the population is away from the ideal equitable state, in which all individuals have the same income (Conceição and Ferreira 2000). Since Theil index is not a relative measure of inequality, the values of this index are not always comparable across different groups and sizes of population (Trapeznikova 2019). Gini index is the most widely used and recognized measure of inequality (Trapeznikova 2019). It can be used to measure the inequality of any distributions. One of the benefits for using Gini coefficient is to allow for direct comparisons of inequality states across different groups of population, irrespective of their sizes (Afonso et al. 2015). A higher Gini coefficient value indicates higher inequality. The Gini coefficient has been used in measuring inequality in various domains, such as energy consumption (Jacobson et al. 2005), water consumption (Wang et al. 2012), indoor environmental quality (Zhang and Sanake 2020), and healthcare resource allocation (Jian et al. 2015).

In our study, we chose to adapt Gini coefficient in measuring the inequality of disaster impacts for the following reasons: (1) it allows for measurement of distributions of disaster impacts across multiple communities, (2) it allows for comparisons of distributions of disaster impacts across communities with different sizes of population, and (3) it is not affected by the characteristics (e.g., poverty) of the communities, and (4) it is relatively straightforward and easy to interpret.
# CHAPTER 3 UNDERSTANDING INFRASTRUCTURE RESILIENCE, SOCIAL EQUITY, AND THEIR INTERRELATIONSHIPS: EXPLORATORY STUDY USING SOCIAL MEDIA IN HURRICANE MICHAEL

## 3.1 Introduction

Resilience has emerged as an increasingly important factor in developing and maintaining infrastructure in response to both acute (e.g., hurricanes, earthquakes) and slow on-set disasters (e.g., sea level rise) (Doorn 2019). Over the last decade, the growing intensity and frequency of disasters have resulted in huge economic losses mostly in the form of damage to infrastructure, which significantly impacts people's access to services, such as clean water, electricity, transportation, and health care (UN 2016). To allow infrastructure to resist or absorb disturbance, and retain basic functional and service capacities, investing in and implementing disaster resilience strategies have become a national imperative for all Americans (NRC 2012). However, one of the overlooked problems with infrastructure resilience is that damaged infrastructure due to disasters could result in varying levels of disturbance to the residents. Such damage is typically not evenly distributed across different communities; low income and minority communities are more vulnerable to disaster risks, and they also struggle more to recover (Emrich et al. 2019). For example, after Hurricane Harvey, more severe flooding damage was found in communities or households with lower incomes as lower income Americans are more likely to live in neighborhoods or buildings that are more susceptible to flooding or other impacts from storms (Krause and Reeves 2017).

To reduce or eliminate disparities of access to infrastructure due to disasters, the 2030 Global Sustainable Development Agenda of the United Nations highlights the importance of understanding the integrated nature of infrastructure, inequality, and resilience (UN 2016). For example, how does infrastructure resilience affect social equity? How is social equity integrated

into resilience assessment or planning? Benchmarking the definitions of social equity in the literature (e.g., Emrich et al. 2019, UN 2019, APA 2021), social equity is defined as this study as equal opportunities and resources provided to different populations through the functions offered by infrastructure. Achieving social equity means reducing or eliminating disparate access to goods, services, and amenities among different populations, including socially vulnerable populations. Socially vulnerable population include the economically disadvantaged, racial and ethnic minorities, the elderly, the uninsured, the homeless, the disabled, those with chronic health conditions, and those with language barriers (Rao et al. 2019, AJMC 2006). They often have the fewest resources for disaster preparedness, live in disaster prone areas, and lack social, political, and economic capital needed to adapt to and recover from disasters (IWR 2016). Resilient infrastructure is vital in offering stabilized essential services (e.g., water, power, communication, and transportation) to socially vulnerable populations, thus playing an important role in supporting social equity in disasters (Doorn 2019, UN 2016).

Despite such interlinkage between infrastructure resilience and social equity, there is limited research that provides an explicit understanding about the complex relationships between infrastructure resilience and social equity. Extensive research efforts have focused on either infrastructure resilience or social equity. For instance, on one hand, the research on improving the resilience of infrastructure systems has received significant attention in the engineering circles (e.g., Karamouz et al. 2019, Rasoulkhani et al. 2019, Karamouz et al. 2018, Aydin et al. 2018). On the other hand, social equity has been widely studied in the domains of psychology, social science, political science, and anthropology (e.g., Domingue and Emrich 2019, Rodríguez-Izquierdo 2018, Castillo et al. 2019, Riccucci and Van Ryzin 2017). However, researchers from different research domains or backgrounds usually focus on one of these distinct fields, making links between these two areas less commonly studied than any of these areas taken in isolation (UN 2016). Thus, researchers and organization (e.g., UN 2016, Rockfeller Foundation 2020) have been calling for

the need to understand the complex links between infrastructure resilience and social equity to uncover important synergies and tradeoffs.

To fill the knowledge gap, this chapter aims to explore the interlinkages between infrastructure resilience and social equity using a data-driven method. Data from different sources were collected and analyzed, including social media data, census data, and disaster damage, relief, and recovery data. In recent years, social media has become one of the emerging data sources to understand human activities and behaviors in a disaster setting (Resch et al. 2018, Zou et al. 2018a). Compared with traditional data sources (e.g., surveys), social media offers real time human generated data with spatiotemporal characteristics. Social media data allows researchers to conduct diverse studies in the context of disasters; the topics range from those that are related to infrastructure, such as damage assessment (e.g., Resch et al. 2018, Cervone et al. 2017, Wu and Cui 2018), infrastructure accessibility (e.g., Hamstead et al. 2018), and infrastructure recovery (e.g., Nazer et al. 2016, Schempp et al. 2019), to those that are relevant to social impacts, such as communication patterns (e.g., Wukich et al. 2019Goldgruber et al. 2017), public awareness (e.g., Martín et al. 2017), social disparities (e.g., Zou et al. 2018b). Among different sources of social media data (e.g., Facebook, Instagram, Twitter, and Tumblr), Twitter is the most widely used data source for conducting research as Twitter data are relatively easy to access, cost-effective, have less privacy concerns, and have proven to be a relatively reliable source of valuable information (Kryvasheyeu et al. 2016, Zou et al. 2018a).

As a first step toward understanding of the complex relationships between infrastructure resilience and social equity, this chapter aims to explore whether social media data can be used as indicators of either infrastructure resilience or social equity conditions in the context of a disaster. It aims to address the following research questions (RQs): RQ1: How are infrastructure resilience conditions in the disaster-affected communities reflected by Twitter activities?

RQ2: How are social equity conditions in the disaster-affected communities reflected by Twitter activities?

RQ3: Do social equity characteristics of communities have impacts on the infrastructure resilience conditions of the communities?

To address these questions, Twitter activities generated by 12 disaster-affected counties in Florida during Hurricane Michael in 2018 were collected and analyzed. In addition, socioeconomic data were selectively collected to represent the social equity conditions of these disaster affected counties, while infrastructure damage, relief, and recovery data were collected to reveal the infrastructure resilience conditions of these counties. Statistical correlation analyses were then conducted (1) between the social equity variables and Twitter variables, (2) between the infrastructure resilience variables and the Twitter variables, and (3) between the social equity variables and the infrastructure resilience variables. The remainder of the chapter presents the research context and methodology, discusses the results and findings, and summarizes the contributions and conclusions.

### 3.2 Research Context

Hurricane Michael was a Category 5 hurricane that made landfall near Mexico Beach, Florida on October 10, 2018, with a maximum speed sustained wind speed of 257.50 kph (160 mph) (Wamsley 2019). It is one of the strongest hurricanes to have ever made a landfall in the Florida Panhandle region.

#### RIs of twelve affected counties of Florida



Figure 3-1 Ratio indices of the disaster affected counties of Florida.

Hurricane Michael was selected as the research context for three reasons. First, it caused massive damage and destruction to the infrastructure of coastal communities in the Florida Panhandle region. According to National Oceanic and Atmospheric Administration (NOAA), the storm surges brought floods with water levels rising 2.74-4.27 m (9-14 ft) above the normal level in the Panhandle area (NOAA 2018a). High storms surges and intense wind speeds caused significant damage to buildings and infrastructure. According to a preliminary damage assessment report (NHC 2018), Hurricane Michael caused approximately \$25 billion in direct damage. These surges and wind forces caused complete power outages and a significant portion of the communication network outages in the Florida Panhandle region, with some of these outages lasting for more than a month. Physical structures, such as communication towers, electric poles, substations, and transmission towers, were severely damaged due to intense wind forces combined with fallen and flying debris and flash floods. In addition, transportation infrastructure (e.g., roads and bridges), was blocked, damaged, or completely washed away due to fallen trees and flash floods (NHC 2018). Second, the regions struck by Hurricane Michael are among the most socially vulnerable

regions in the United States (DirectRelief 2018, Pathak et al. 2020). According to Federal Emergency Management Agency (FEMA 2018), 12 counties in Florida were severely impacted and issued disaster declaration as of November 15, 2018. These counties include Bay, Calhoun, Franklin, Gadsden, Gulf, Holmes, Jackson, Leon, Liberty, Taylor, Wakulla, Washington Counties (Figure 3-1). Third, there is relatively limited disaster research that focuses on Hurricane Michael as compared to studies on other hurricanes (e.g., Hurricane Sandy, Hurricane Harvey, Hurricane Irma).

#### 3.3 Methodology

### 3.3.1 Data Collection Methods

## 3.3.1.1 Twitter Data Collection

In this study, Twitter was used as the source of social media data. Twitter provides an online social networking platform where people can communicate in short messages, share images, or webpages links, all of which are known as tweets. With 100 million daily active users and around 500 million daily tweets (Forsey 2019), Twitter is one of the most popular social networks that allow for the collection of a huge amount of information on human thoughts and activities in a disaster setting (Zou et al. 2018a). Twitter data collection and processing methods proposed by Zou et al. (2018a) was benchmarked, and the following steps were taken to collect and process Twitter data for analysis:

### Step 1: Background Tweets Collection

Background tweets are the tweets generated from the Florida Panhandle area during the preparedness, response, and initial recovery phases of Hurricane Michael. The background tweets were collected by combining two different types of tweets: geotagged and non-geotagged tweets. The geotag of a tweet can be either an exact GPS coordinate (latitude and longitude) that represents the precise location of a user's mobile device or an approximate place name selected by the user

from a list of place names suggested by Twitter, such as a city or a neighborhood (Twitter 2020). Tweets are generally not embedded with geolocations unless enabled by the users, and it is estimated that less than 1% of all the tweets are geotagged (Ajao et al. 2015).

Twint is used to collect geotagged tweets generated from Florida Panhandle area. Twint is a twitter scraping tool that utilizes Twitter search operators to scrape tweets from specific users, certain topics, or geographic locations (PyPI 2018). Twint is able to extract all the tweets that fall within a predetermined geographical coordinate and a radius of coverage. A Twitter search query was then scripted to extract all the geo-tagged tweets that were generated from each of the 12 Florida Panhandle counties. The tweets were extracted from October 1, 2018 to November 16, 2018 to cover preparedness, response, and initial recovery phases of Hurricane Michael. Each extracted tweet contains information including the time of creation, tweet ID, tweet content, tweet status (i.e., if the tweet is a reply or retweet), coordinates, place, and information about the user who posted the tweets (e.g., names, screen name, locations, numbers of followers, friend, and list).

The non-geotagged tweets in the background collection are the tweets without geotags but sent by users whose addresses are in the Florida Panhandle area. A publicly available Twitter data archive (Internet Archive 2020) was used to collect these non-geotagged tweets. The Internet Archive provides a chronological collection of tweets randomly selected from general twitter stream since 2012. In the archive, each tweet is stored in JavaScript Object Notation (JSON) format and contains information such as the textual content of the tweet and user profile. To collect non-geotagged tweets, a place-name lexicon was created including all the municipality names of each Florida Panhandle county. For each tweet, the address in the user profile was examined. The tweet was extracted if the address contains any municipality name from the place-name lexicon. Finally, the extracted non-geotagged tweets from Internet Archive were combined with extracted geotagged tweets to form the background tweets collection.

## Step 2: Disaster-related Tweets Filtering

After the background tweets were collected, the disaster-related tweets were further filtered. The filtering process includes two main steps. First, all the extracted tweets were filtered to include only the following relevant information to this study: (1) time when the tweet was generated, (2) the tweet content, (3) user name, (4) user's profile information, (5) tweet's geolocation if enabled, and (6) user's location. Second, the disaster-related tweet data were further filtered based on the disaster-related keywords. A total of 39 keywords were used, such as hurricane, Michael, storm, response, preparedness, power, flood, infrastructure, and damage, etc. These keywords were derived through a combination of deductive approach and inductive approach. The deductive approach identifies the keywords based on the terms that are commonly used for filtering disaster-related tweets according to other social media literature (e.g., Zou et al. 2018b, Kryvasheyeu et al. 2016) in the disaster domain. The inductive approach identifies the keywords based on empirical observation of tweet contents. The keywords and the approaches that were used to derive them are listed as follows:

Deductive approach: Hurricane, power, weather, damage, storm, recovery, flood, local government, FEMA, climate, safe, food, and water

Inductive approach: Michael, infrastructure, emergency, rain, wind, surge, panhandle, Panama, Mexico, beach, relief, wave, responder, gulf, federal aid, resource, rebuild, supply, response, mitigate, prepare, highway, pray, rescue, search, and survivor.

For each county, the total number of background tweets and disaster-related tweets were counted and tabulated. A Python 3.6 script was used to filter the original tweets and count the disasterrelated and total background tweets for each of the 12 studied counties.

34

## 3.3.1.2 Infrastructure Resilience Data Collection

Infrastructure resilience can be characterized by robustness, rapidity, resourcefulness, and redundancy (Bruneau and Reinhorn 2006). Each of these characteristics can be further represented through concrete dimensions and variables (Table 3-1). Eight infrastructure resilience variables were selected for analysis in this study for two reasons: (1) these variables can represent the key characteristics of infrastructure resilience, and (2) their data can be obtained through public sources. The selected variables include damage value per capita (II), percentage of power outages (I2), percentage of communication service outages (I3), power outage recovery time (I4), communication service outage recovery time (I5), disaster recovery cost per capita (I6), disaster relief and emergency assistance fund per capita (I7), and number of insurance claims per capita (I8). As indicated in Table 1, the data for these variables were collected from different public sources, including Florida Department of Transportation (FDOT), Federal Emergency Management Agency (FEMA), Florida Division of Emergency Management (FDEM), Federal Communications Commission (FCC), Florida Department of Economic Opportunity (FDEO), and Florida Office of Insurance Regulations (FOIR).

Main characteristic	Dimension	Variable	Data source		
	Everetional loss of	Damage value per capita (I1)	(FDEM 2019)		
Robustness	Functional loss of	Percentage of power outages (I2)	(POR 2018)		
	mitasuucture	Percentage of communication service outages (I3)	(FCC 2018)		
Rapidity	Time required to recover	Power outage recovery time (I4)	(POR 2018)		
	to previous functional levels	Communication service outage recovery time (I5)	(FCC 2018)		
	Cost required to recover to previous functional levels	Disaster recovery work cost per capita (I6)	(FDOT 2019)		
Resourceful- ness	Availability of economic resources	Disaster relief and emergency assistance fund per capita (I7)	FDEO (2019)		
Redundancy	Alternate plan to maintain the functional level of infrastructure	Number of insurance claims per capita (I8)	(FOIR 2019)		

Table 3-1 Infrastructure resilience variables

## 3.3.1.3 Social Equity Data Collection

A total of 18 social equity variables (Table 3-2) were selected based on two main criteria: (1) they are representative indicators of social equity verified based on the review of literature (e.g., Schneiderbauer et al. 2006, Cutter et al 2010), and (2) they have consistent and high-quality data available from the national sources. These variables include percentage of population under 18 years (S1), percentage of population 65 years and above (S2), percentage of male population (S3), percentage of female population (S4), percentage of white population (S5), percentage of black or African American population (S6), percentage of Hispanic or Latino population (S7), percentage of population speaking other than English language at home (S8), percentage of population having high school degree and higher (S11), percentage of population without health insurance (S12), percentage of population with disability (S13), per capita income (S14), percentage of population under poverty (S15), median household income (S16), median value of owner occupied housing units (S17), and total employment (S18). For each variable, the data of each of the twelve affected counties were collected from the U.S. Census Bureau (U.S. Census 2019). The U.S. Census Bureau provides data with quality, reliability, and consistency (Santos 2019).

Dimension	Variable	Data source
	Percentage of population under 18 years (S1)	(U.S.
Age		Census
	Percentage of population 65 years and above (S2)	2019)
	Percentage of male population (S3)	(U.S.
Gender		Census
	Percentage of female population (S4)	2019)
	Percentage of white population (S5)	(U.S.
Race	Percentage of black or African American population (S6)	Census
	Percentage of Hispanic or Latino population (S7)	2019)
Language	Demonstrate of nonvolation encoding other than Enclick	(U.S.
	Percentage of population speaking other than English	Census
	language at nome (58)	2019)
Technology	Percentage of households with internet connection (S9)	

Table 3-2 Social ed	quity variab	les
---------------------	--------------	-----

		(U.S.
	Percentage of households with computer (S10)	Census
		2019)
	Percentage of population having high school degree or higher	(U.S.
Education	(\$11)	Census
	(311)	2019)
Health	Percentage of people without health insurance (S12)	(U.S.
	Demonstrate of nonvelotion with dischility (\$12)	Census
	Percentage of population with disability (S13)	2019)
	Per capita income (S14)	
Economics	Percentage of population under poverty (S15)	(U.S.
	Median household income (S16)	Census
	Median value of owner-occupied housing units (S17)	2019)
	Total employment (S18)	

## 3.3.2 Data Analysis Methods

## 3.3.2.1 Twitter Data Indices

To analyze Twitter activities during Hurricane Michael, the Ratio Index (RI), Normalized Ratio Index (NRI), and Sentiment Index (SI) were calculated for each of the twelve affected counties in Hurricane Michael. Ratio Index (RI) is a Twitter index that can be used to represent the intensity of twitter activities in certain topics or domains. In this study, it is calculated using the number of disaster-related tweets divided by the total number of background tweets [Eq.(3-1)] (Zou et al. 2018b).

$$RI = \frac{Total number of disaster - related tweets}{Total number of background tweets}$$
(3-1)

In order to eliminate the effects of disaster threat levels on Twitter activities, a normalized ratio index was defined so that disparities of Twitter activities under the same disaster threat level can be investigated. The NRI can be calculated as RI divided by the average sustained wind speed [Eq.(3-2))]. The sustained wind speed data for each of the most affected counties were collected from the National Hurricane Center under NOAA (2018), and the NRI was calculated for each county using Eq. (3-2).

$$NRI = \frac{RI}{Average sustained wind speed}$$
(3-2)

Sentiment analysis aims to evaluate people's opinions, thoughts, and feelings, expressed in Twitter by assigning sentiment scores based on tweet contents (Caragea et al. 2014). Previous studies on social media data analysis have exhibited that sentiment analysis of tweet contents can be used to understand human perceptions, concerns, or psychological impacts during disasters (Caragea et al. 2014, Kryvasheyeu et al. 2016). This study used the valence aware dictionary and sentiment reasoner (VADER), a lexicon and rule-based python tool, to quantify the sentiment score for each of the tweet contents (Hutto and Gilbert 2014). VADER combines a manually created comprehensive sentiment lexicon with a set of grammatical and syntactical heuristics to determine the overall sentiment intensity of an input text (Hutto and Gilbert 2014). The comprehensive lexicon of VADER was constructed by examining existing well-established sentiment word banks [e.g., linguistic inquiry word count (LIWC), affective norms for english words (ANEW), and general inquirer (GI)] and incorporating numerous lexicon features related to sentiment expressions, including a full list of emotion and sentiment related acronyms (e.g., LOL), and commonly used slang with sentiment value (e.g., meh, nah) (Hutto and Gilbert 2014). In developing VADER, twenty independent human raters were employed for the intensity rating of lexical features, where the features were rated on a scale from extremely negative (-4) to extremely positive (+4), with neutral (0) in between (Hutto and Gilbert 2014). VADER has been found to perform exceptionally well in the social media domain and even outperform human raters at correctly identifying the sentiment intensity of tweets (Hutto and Gilbert 2014).

In our study, we employed VADER to determine if the text in the tweet content expresses positive, negative, or neutral opinion. For each tweet, VADER assigns a sentiment score ranging from 1 (extremely positive) to -1 (extremely negative), and a score between -0.05 and 0.05 is considered

neutral. For a county, the sentiment index is calculated as the mean sentiment score of each tweet content from the county [Eq. (3-3)].

$$SI = \frac{Sum of sentiment scores for disaster - related tweets}{Total number of disaster - related tweets}$$
(3-3)

## 3.3.2.2 Infrastructure Variable Index

When analyzing the infrastructure resilience conditions of the affected communities with different social equity characteristics, it is acknowledged that the counties that are close to the hurricane path naturally had more severe damage and could also take longer time to recover. Therefore, to eliminate the effects of disaster threat levels on the infrastructure, a set of normalized infrastructure resilience (NIR) variables were developed. Accordingly, the NIR data were calculated by dividing the original infrastructure resilience data with the average sustained wind speed during the hurricane period in each county [Eq. (3-4)]

$$NIR = \frac{Infrastructure\ resilience\ data}{Average\ sustained\ wind\ speed}$$
(3-4)

#### 3.3.2.3 Correlation Analyses

To answer the research questions, three sets of correlation analyses were conducted (1) between Twitter variables (RI and SI) and infrastructure resilience variables, (2) between Twitter variables (RI, NRI and SI) and social equity variables, and (3) between normalized infrastructure resilience variables (NIR) and social equity variables. Both the Pearson's Product-Moment Correlation (Pearson's correlation for short) and Spearman' Rank Order Correlation (Spearman's correlation for short) were used to conduct the correlation analyses. The Pearson's correlation coefficient is a measure of the strength of a linear association that exists between two continuous variables and is denoted by r (Laerd 2020a). The Spearman's correlation is a nonparametric version of the Pearson's correlation of the Pearson's correlation.

monotonic association between two variables rather than strength and direction of the linear relationship between two variables, which is what Pearson's correlation determines (Laerd 2020b). The Spearman's correlation can be used for both continuous variables and ordinal variables. Additionally, compared to Pearson's correlation, Spearman's correlation is more robust to outliers (Mukaka 2012).

The results of the analyses were interpreted based on both the correlation coefficients (Pearson's r and Spearman's  $\rho$ ) and the probability value (p-value). For the correlation coefficients, an absolute value of 0.50 and higher represents a high association between two variables, while an absolute value between 0.30 and 0.49 represents a medium association, and an absolute value between 0.10 to 0.29 represents a small association (SS 2020, Cohen et al. 2013). For the probability value, most researchers consider a standard significance level as 0.1, 0.05, or 0.01 for hypothesis tests (Frost 2020). In our study, a significance level of 0.1 was selected because (1) it allows the test to be more sensitive to detect significance in the data, (2) it is suitable for exploratory research to identify new hypothesis (Gaus et al. 2015), and (3) it is suitable to use for small sample size data sets (Kim and Choi 2019). Thus, if the p-value is less than 0.1, the association results are considered as statistically significant. The following sections discuss about the main findings of the analyses.

## 3.4 Results Analysis and Discussion

During the study period (from October 1, 2018, to November 16, 2018), 128 million tweets were collected. A total of 1,827,624 tweets were collected as the background tweets. Among the background tweets, a total of 103,660 disaster-related tweets were identified based on the disaster-related keywords. The RIs for the twelve affected counties were first calculated using Eq. (3-1), as shown in Fig. 1. Similarly, the NRIs and SIs for the twelve counties were calculated using Eq. (3-2) and Eq. (3-3), respectively. The three sets of correlation analyses were then conducted.

#### 3.4.1 Analyzing Relationships between Infrastructure Resilience and Twitter Activities

To answer RQ1, the correlation analyses were conducted to assess the relationships between the infrastructure resilience variables and the Twitter activity variables (i.e., RI, SI). **Error! Reference source not found.** and Figure 3-2 present the correlation results that are statistically significant. As per Table 3, three infrastructure resilience variables, including damage value per capita (I1) (Pearson's r = 0.750, p=0.058), communication service outage recovery time (I5) (Pearson's r = 0.556, p=0.060), and disaster recovery cost per capita (I6) (Pearson's r = 0.547, p =0.066) show statistically significant, strong positive linear associations with the Twitter activity variable, RI. In general, the results indicate that communities that experienced more severe damage to infrastructure and spent a longer time on recovery were more active on Twitter in Hurricane Michael.

Dolotionahina	Pearson's correlation		Spearman's correlation		
Kelationship	r value	p-value	ρ value	p-value	
RI vs I1	0.750	0.058 <sup>b</sup>	0.149	0.645	
RI vs I5	0.556	0.060 <sup>b</sup>	0.467	0.125	
RI vs I6	0.547	0.066 <sup>b</sup>	0.168	0.602	
RI vs I7	0.544	0.068 <sup>b</sup>	0.224	0.484	
RI vs I8	0.626	0.029 <sup>b</sup>	0.427	0.167	
SI vs I4	0.598	0.040 <sup>b</sup>	0.687	0.014 <sup>b</sup>	

Table 3-3 Statistically significant results of correlation analyses between infrastructure resilience variables and Twitter activities

Note: RI = ratio index; and SI = sentiment index

<sup>a</sup>The numbering of infrastructure resilience variables follow that in Table 1. <sup>b</sup>The p-value is significant at 0.1 level.

These results are consistent with a number of studies (e.g., Zou et al. 2018b, Kryvasheyeu et al. 2016) that indicate disaster-related Twitter activities are higher in those regions that have severe damage and destruction due to disasters. Other studies (Kent and Capello 2013, Starbird and Palen 2010) show that disaster-related Twitter activities originate more from the communities that are proximal to the crisis events compared to the communities located farther away. Social media plays an increasingly important role in the context of disasters. It has changed the ways of crisis

communication, and it has turned out to be an important tool for information dissemination and exchange during emergency events. For example, during Hurricane Michael, the Twitter accounts of government officials were used for disseminating hurricane-related news, instructions, and educational resources for hurricane preparedness and response. The community residents were concerned about the damage and destruction that happened in their surroundings, and they turned to Twitter for disaster-related communication and information exchange. In the recovery process, social media is more commonly used for locating friends and families, facilitating volunteering inquiries, requesting and offering resources, and communicating and coordinating the recovery supplies (CivicPlus 2020). After Hurricane Michael, different government officials, public agencies, and non-government organizations (NGOs) used their official Twitter accounts to provide updates on recovery status, coordinate relief and recovery efforts, and offer resources or support.



Figure 3-2 Line charts showing the relationships between infrastructure resilience variables and Twitter activities (RI and SI)

(The numbering of infrastructure resilience variable follow that in Table 1): (a) I1 versus RI; (b) I5 versus RI; (c) I6 versus RI; (d) I7 versus RI; (e) I8 versus RI; and (f) I4 versus SI.

Besides the previous findings, in our study, the hurricane damage value data from Franklin County were found to be an outlier; the relatively high damage value per capita was not aligned with the relatively low Twitter activities in that county. A further investigation on the data of damage value showed that Franklin County has a significantly higher damage value compared to the other eleven affected counties (Figure 3-1) of the Florida Panhandle. This is mainly because of the tremendous amount of damage on coastal Highway 98 connecting Carrabelle to Saint George Island in Franklin County; the gulf side of the two-lane roadway was completely washed out (FDEM 2019). As a scenic highway along the shoreline, coastal Highway 98 has fewer coastal barrier protections installed to resist the potential high tides and storm surges. Without adequate and robust barrier protections that serve as the mainland's first line of defense against the impacts of severe storms and erosions, the roadway infrastructure along the gulf coast of the county was especially

vulnerable during Hurricane Michael. Hurricane Michael generated strong wind forces and storm surges that ranged from 1.52 to 5.79 m (5 to 19ft), which caused extensive damage to residential buildings, critical facilities, and infrastructure such as roads and highways. The damage value was estimated to be \$10 billion for Franklin County (FDEM 2019).

## 3.4.2 Analyzing Relationships between Social Equity and Twitter Activities

To answer RQ2, the correlation analyses were conducted to assess the relationships between the social equity variables and the Twitter activity variables (i.e., RI, NRI, and SI). **Error! Reference source not found.** and Figure 3-3 present the correlation results that are statistically significant. As per Table 4, the RI has statistically significant, strong negative correlations with the percentage of population speaking other than English language at home (S8) (Spearman's  $\rho = -0.592$ , p = 0.043), the percentage of Hispanic or Latino population (S7) (Spearman's  $\rho = -0.573$ , p = 0.051), and the percentage of population with disability (S13) (Spearman's  $\rho = -0.565$ , p = 0.056). In contrast, the RI has a statistically significant, strong positive correlation with the percentage of households with internet connection (S9) (Spearman's  $\rho = 0.566$ , p = 0.055). In general, these results indicate that the communities with a higher percentage of vulnerable populations (e.g., those with language barriers, the minority, the disabled) are less represented on social media, while the communities with relatively high socioeconomic status are more active on social media. The following paragraphs provide the discussion of the main findings from the results.

Relationship <sup>a</sup>	Pearson's correlation		Spearman's correlation	
	r value	p-value	ρ value	p-value
RI vs S7	-0.408	0.187	-0.573	0.051 <sup>b</sup>
RI vs S8	-0.383	0.220	-0.592	0.043 <sup>b</sup>
RI vs S9	0.390	0.210	0.566	0.055 <sup>b</sup>
RI vs S13	-0.267	0.402	-0.565	0.056 <sup>b</sup>
SI vs S13	-0.435	0.158	-0.618	0.032 <sup>b</sup>
SI vs S15	-0.392	0.208	-0.730	0.007 <sup>b</sup>

Table 3-4 Statistically significant results of correlation analyses between social equity variables and Twitter activities

Note: RI = ratio index; and SI = sentiment index.

<sup>a</sup>The numbering of social equity variables follow that in Table 2.

<sup>b</sup>The p-value is significant at 0.1 level.

Based on the results, the communities with higher percentages of vulnerable populations were less active on Twitter during Hurricane Michael. Despite the efforts and goals to reduce or eliminate disparities in the context of disasters, significant disparities in different aspects such as risk levels, access to capital, and disaster-related knowledge and resources, continue in these disaster-affected communities. Vulnerable populations could face various obstacles that result in their "silence" on social media. For example, people who are disabled are exposed to a higher constant risk in disasters due to personal health concerns, higher chance of injuries and mental health problems, lack of awareness of situations, isolation from communities, and physical barriers in evacuation (Stough 2017). Previous studies (e.g., Morris et al 2014, USDoC 2019) also suggest that people with disabilities show a lower rate of technology use. These obstacles often force disabled people to strive to address physiological needs and maintain their personal safety in disasters, leaving less time and lower chances of using or communicating through social media.



Figure 3-3 Line charts showing relationships between social equity variables and Twitter activities (RI and SI)

(The numbering of social equity variables follow that in Table 2): (a) S7 versus RI; (b) S8 versus RI; (c) S9 versus RI; (d) S13 versus RI; (e) S13 versus SI; and (f) S15 versus SI.

Similarly, the minority populations (e.g., Hispanic populations) and the populations speaking other than English language, were less active on Twitter during Hurricane Michael. Language barriers have a significant impact on how people perceive and prepare for disasters. For example, disaster warning alerts, preparedness strategies, and disaster-related knowledge are mostly communicated through the English language in the United States. People who do not speak English have to rely on the secondary sources of information to prepare for and respond to disasters. Although social media, such as Twitter, is a platform for the global community, the analysis of Twitter user behavior shows that users tend to confine their connectivity within those who speak the same language; the interactions among the users are fragmented and often limited by the language (Young 2020). Language barriers impede effective communication through social media between the affected minority populations and relief operations during disasters.

To eliminate the effects of disaster threat levels on Twitter activities, the correlation analyses between NRI and social equity variables were also conducted to explore the relationships between Twitter activities and social equity variables under the same disaster threat level. Table 3-5 summarizes the correlation results between the social equity variables and (1) RI, and (2) NRI. By comparing the results between (1) social equity variables vs. RI, and (2) social equity variables vs. NRI, it is observed that the correlation coefficients (Spearman's  $\rho$ ) of three social equity variables, including the percentage of black or African American population (S6), percentage of population without health insurance (S12), and percentage of population under poverty (S15), change from negative values to positive values.

Social	Pearson's r value		Spearman	ı's ρ value
equity variableª	RI	NRI	RI	NRI
S5	0.243	0.174	0.245	-0.007
S6	-0.212	-0.105	-0.238	0.154
S9	0.390	-0.146	0.566	-0.146
S10	0.128	-0.327	0.329	-0.327
S11	0.129	-0.344	0.252	-0.344
S12	-0.146	0.263	-0.098	0.277
S13	-0.267	0.086	-0.565	-0.028
S14	0.100	-0.329	0.280	-0.329
S15	-0.071	0.182	-0.425	0.242
S16	0.068	-0.446	0.224	-0.446
S17	0.340	-0.215	0.455	-0.215
S18	-0.034	-0.317	0.259	-0.317

Table 3-5 Correlation coefficients of correlation analyses between social equity variables and Twitter activities

Note: RI = ratio index; and NRI = normalized ratio index.

The number of social equity variables follow that in Table 2.

On the other hand, the correlation coefficients of eight social equity variables, such as the percentage of population having high school degree and higher (S11), per capita income (S14), median household income (S16), median value of owner-occupied housing units (S17), and total employment (S18), change from positive values to negative values. Collectively, the shifts in correlation tendencies reveal that, by accounting for the hurricane wind threat levels, communities

with higher percentages of vulnerable populations became more active on Twitter. In another word, under the same threat level, vulnerable populations were more active on Twitter during Hurricane Michael. This is probably because, facing with the same level of disaster threat, vulnerable populations perceived a greater level of difficulty and hardship in disasters, and they reflected this hardship by expressing their concerns, needs, and difficulties on social media.

#### 3.4.3 Analyzing Relationships between Social Equity and Infrastructure Resilience

To answer RQ3, correlation analyses were conducted between the normalized infrastructure resilience variables and the social equity variables. Table 3-6 and Figure 3-4 present the statistically significant correlation results. Three main findings are discussed in the following paragraphs.

First, according to Table 3-6, under the same disaster threat level, there is a significant and strong positive correlation between the damage value per capita (I1\*) and the percentage of Hispanic or Latino population (S7) (Spearman's  $\rho = 0.559$ , p = 0.059), and there is a significant and strong positive correlation between the disaster recovery cost per capita (I6\*) and the percentage of population without health insurance (S12) (Spearman's  $\rho = 0.518$ , p = 0.084). In addition, there is a significant and strong positive linear association between disaster recovery cost per capita (I6\*) and the percentage of population over 65 years old (S2) (Pearson's r =0.738, p =0.006). Collectively, these results may imply that the communities with higher percentages of vulnerable populations (e.g., the minority, the uninsured, the elderly) might have experienced more severe damage during Hurricane Michael, which also required higher expenses on recovery. Existing research shows that vulnerable populations are often under-prepared before disasters (e.g., lack of home insurances or flooding insurances, inadequate financial resources); thus, they may experience more severe losses (Constible 2018). These populations are also more likely to live in the disaster-prone regions with older and structurally deficient houses. In addition, a large percentage of houses, in the Florida Panhandle, were not able to withstand the strength of Hurricane Michael as they were

constructed before the implementation of stricter building codes, which happened after Hurricane

Andrew in 1992 (Allen 2018).

Deletionshina	Pearson's correlation		Spearman's correlation	
Relationship	r value	p-value	ρ value	p-value
S1 vs I6 <sup>b</sup>	-0.809	0.001°	-0.543	0.068°
S2 vs I6 <sup>b</sup>	0.738	0.006°	0.476	0.118
S3 vs I6 <sup>b</sup>	0.411	0.184	0.501	0.097°
S3 vs I7 <sup>b</sup>	0.550	0.064 <sup>c</sup>	0.585	0.046 <sup>c</sup>
S4 vs I6 <sup>b</sup>	-0.411	0.184	-0.501	0.097
S4 vs I7 <sup>b</sup>	-0.550	0.064 <sup>c</sup>	-0.585	0.046
S7 vs I1 <sup>b</sup>	0.010	0.975	0.559	0.059 <sup>c</sup>
S10 vs I4 <sup>b</sup>	-0.385	0.216	-0.517	0.085°
S10 vs I5 <sup>b</sup>	-0.494	0.102	-0.531	0.075 <sup>c</sup>
S11 vs I5 <sup>b</sup>	-0.438	0.154	-0.529	0.077°
S12 vs I5 <sup>b</sup>	0.424	0.170	0.567	0.054°
S12 vs I6 <sup>b</sup>	0.330	0.295	0.518	0.084°

Table 3-6 Statistically significant results of correlation analyses between social equity variables and normalized infrastructure resilience variables

<sup>a</sup>The numbering of infrastructure resilience and social equity variables follow that in Tables 1 and 2, respectively.

<sup>b</sup>Normalized infrastructure resilience variables by the wind threat levels.

<sup>c</sup> The p-value is significant to 0.1 level.

Second, as per Table 3-6, under the same disaster threat level, the communication service outage recovery time (I5\*) has significant and strong negative correlations with the percentage of households with computer (S10) (Spearman's  $\rho = -0.531$ , p = 0.075) and percentage of population with high school degree and higher (S11) (Spearman's  $\rho = -0.529$ , p = 0.077), and it has a significant and strong positive correlation with the percentage of population without health insurance (S12) ( $\rho = 0.567$ , p = 0.054). The power outage recovery time (I4\*) also shows a significantly negative correlation with the percentage of households with computer (S10) (Spearman's  $\rho = -0.517$ , p = 0.085). These results may imply that the communities with higher socioeconomic status are more likely to require shorter time for recovery, and vice versa. Previous recovery experiences show that wealthier communities typically receive more reinvestment on their infrastructure compared to low-income communities (Nexus 2017). In addition, communities with highly educated populations tend to take short recovery time after disasters. Highly educated populations are likely

to be aware of the ongoing situations in their surroundings during disasters. They have the capability to communicate with local agencies, share information, and seek for aid and resources to recover from disasters.



Figure 3-4 Line charts showing relationships between social equity variables and normalized infrastructure resilience variables

(The numbering of infrastructure resilience and social equity variables follow that in Tables 1 and 2, respectively): (a) S1 versus I6\*; (b) S2 versus I6\*; (c) S3 versus I6\*; (d) S3 versus I7\*; (e) S4 versus I6\*; (f) S4 versus I7\*; (g) S7 versus I1\*; (h) S10 versus I4\*; (i) S10 versus I5\* (j) S11 versus I5\*; (k) S12 versus I5\*; and (l) S12 versus I6\*.

Third, the results in Table 3-6 show the significantly positive correlations between the percentage of male population (S3) and (1) the disaster relief and emergency assistance fund per capita (I7\*) (Spearman's  $\rho = 0.585$ , p = 0.046), and (2) the disaster recovery cost per capita (I6\*) (Spearman's  $\rho = 0.501$ , p = 0.097). In contrast, significantly negative correlations are observed between the percentage of female population (S4) and (1) the disaster relief and emergency assistance fund per

capita (I7\*) (Spearman's  $\rho = -0.585$ , p = 0.046), and (2) the disaster recovery cost per capita (I6\*) (Spearman's  $\rho = -0.501$ , p = 0.097). In addition, disaster relief and emergency assistance fund per capita (I7\*) is found to have a significantly positive linear association with the percentage of male population (S3) (Pearson's r = 0.550, p=0.064), while having a significantly negative linear association with the percentage of female population (S4) (Pearson's r = -0.550, p=0.064). These results may reflect gender-based disparities in disaster recovery and relief efforts in Hurricane Michael. Existing studies suggest that male populations have a higher sense of responsibility in an emergency event (Ariyabandu 2009, Olson 2017); male populations may be more aware of damage in their communities, volunteer to take responsibilities in reconstruction works, and seek aid and support from relief agencies to support recovery. On the contrary, women are more likely to take the role of caregivers; they protect, nurture, and assist family members during emergency events (Ashraf and Azad 2015, Ariyabandu 2009), which may inhibit their participation in community disaster recovery activities. Some studies (e.g., Neumayer and Plümper 2007, Ariyabandu 2009) show that women are marginalized in access to disaster recovery and relief resources compared to men within the same community.

#### 3.4.4 Analyzing Sentiment Indices, Infrastructure Resilience, and Social Equity

Sentiment scores of each of the tweets in the affected counties were calculated using VADER and the mean values of these sentiment scores were calculated to determine the SI for each county. In this study, the sentiment scores range from 0.946 (extremely positive) to -0.898 (extremely negative). To further exemplify the tweets associated with the sentiment scores, examples of positive and negative tweets are listed in Table 3-7. According to the tweet contents, most of the negative tweets are related to the damage caused by Hurricane Michael, such as death toll, damaged property, power outages, and fallen trees. For the most positive tweets, the contents are related to the aid, support, supplies, and services people received after Hurricane Michael.

Rank	Tweet Content	Sentiment score	County			
Examples of highly ranked negative sentiment tweets						
1	A closer look at damage in Mexico Beach, FL. The death toll right now is 4 people in Mexico Beach alone. 15 total in Bay County. #hurricanemichael #mexicobeach @ Mexico Beach, Florida	-0.8442	Gulf			
2	My thoughts and prayers are with our brothers and sisters along the #Florida #panhandle. #hurricanemichael looks to be devastating. Scary for me to see another storm of this magnitude	-0.8176	Gulf			
3	Even when power gets restored finding fuel will still be a problem around Panama City, FL. Hereâ€ <sup>™</sup> s why. @weatherchannel #hurricanemichael @ Lynn Haven, Florida	-0.8126	Bay			
4	This is what I #live for. Poor guy lost his #home in the #hurricane, has been outside since #Wednesday and overheated. His temp was 106.8.	-0.8126	Bay			
5	Every tree down except the new ones #hurricanemichael #panhandlestrong @vacasarentals @RickyHaskins @ Calhoun County, Florida	-0.8074	Liberty			
6	from @WFTV - CRUSHED: @GWarmothWFTV got a bird's eye view of the damage from #HurricaneMichael along the panhandle. #hurricane #mexicobeach #florida #orlando #floridaweather	-0.7739	Gulf			
	Examples of highly ranked positive sentiment two	eets				
1	Inspiring to see the determination and positive attitudes of people hit hard by #hurricanemichael. And the selfless service of those here to help. #recover #rebuild #restore #hope	0.926	Jackson			
2	Listen, Tallahassee. This is my spot. The whole family ate for \$11. They were opened after the hurricane. No struggling over here. @ Los Compadres Express	0.9168	Leon			
3	Our hearts go out to the panhandle and all of the communities terribly affected by #hurricanemichael Today, we witnessed, firsthand, the devastation in Marianna, FL as we delivered 47	0.8885	Jackson			
4	@cityofdeltona Our @DeltonaFireRescue deployment teamis still working in #calhouncounty They have been doingdamageassessmentsforthelocal#emergencyoperationscenter	0.886	Calhoun			
5	Thank you edwardcutie mr_chad_barnett and fmpolice for driving a truck load of supplies up to Mexico Beach law enforcement officers and first responders today #buyingthekeys@vacasarentals	0.8687	Gulf			
6	I love this beautiful bride. She is such a sweet pure soul. During this big mess of a hurricane she offered to help us with anything we needed. I sure she was able to help many others	0.807	Bay			

Table 3-7 Partial list of positive and negative Hurricane Michael related tweets with sentiment scores

The correlation results between SI and (1) infrastructure resilience variables, and (2) social equity variables are shown in Table 3-3 and Table 3-4, respectively. According to Table 3-3, a significant and strong positive correlation is observed between the SI and the power outage recovery time (I4) (Spearman's  $\rho = 0.687$ , p = 0.014). This may imply that the communities tend to have a positive and optimistic attitude toward the recovery of the communities and the infrastructure services, even though they spend longer time in recovery. This result coincides with another research study on Hurricane Michael (Pathak et al. 2020), which indicates that the impacted communities emphasized that Hurricane Michael opened doors for growth and change. Local residents look forward to more opportunities that Hurricane Michael could bring to their slowly developing communities. In the recovery, they were determined to rebuild stronger structures instead of restoring to pre-disaster conditions. Many local stakeholders called for a change of policies, such as raising the standards of the building codes (Pathak et al. 2020).

In addition, the results in Table 3-4 show significantly negative correlations between the SI and (1) the percentage of population under poverty (S15) (Spearman's  $\rho = -0.730$ , p = 0.007) and (2) the percentage of population with disability (S13) (Spearman's  $\rho = -0.618$ , p = 0.032). These results suggest that the communities with higher percentages of populations that are disabled or under poverty were more likely to show a higher level of anxiety and deeper concerns regarding the impacts of Hurricane Michael. This result is supported by a number of studies (e.g., Shultz and Galea 2017, Galea et al. 2005, Fothergill and Peek 2004) that indicate vulnerable populations are more likely to develop anxiety, depression, and post-traumatic stress disorder (PTSD) as a result of exposure to disasters. For example, people with physical and mental disabilities are disproportionately affected by the impacts of disasters. These people with pre-existing medical conditions are more prone to develop additional mental health problems as a result of disasters. A study conducted after Hurricane Sandy found the residents with chronic health conditions and disabilities developed sleep disorders, pains, and suicidal ideation as the outcomes of adverse

mental health problems (Boscarino et al. 2014). Similarly, the financially disadvantaged individuals are at a greater mental and emotional risk in disasters. Previous studies (e.g., Rhodes et al. 2010, Kessler et al. 2008, Mills et al. 2007) show that the lack of access to both social and economic resources is correlated with declining mental health conditions, which may result in serious mental illness and higher perceived stress levels after disasters.

#### 3.5 Conclusions

This chapter presents a study that aims to explore the interrelationships between infrastructure resilience and social equity in the context of Hurricane Michael. As part of the study, this chapter examines whether Twitter data can be used as an indicator of the infrastructure resilience or social equity conditions in a disaster setting. Twitter activities generated by the twelve disaster-affected counties in Florida during Hurricane Michael in 2018 were collected and analyzed. In addition, the socioeconomic data were selectively collected to represent the social equity conditions of these disaster-affected counties, while the infrastructure damage, relief, and recovery data were collected to reflect the infrastructure resilience conditions of these counties. Statistical correlation analyses were then conducted (1) between the social equity variables and the Twitter variables, (2) between the infrastructure resilience variables and the Twitter variables, and (3) between the social equity variables and the infrastructure resilience variables. The results indicate that, in the context of a disaster, Twitter activities have the potential to be used as an important indicator of infrastructure resilience conditions. In general, socially vulnerable populations are less active and representative on social media. However, under the same disaster threat level, the vulnerable populations become more active, and this is probably because of more difficulties and hardship they perceive during disasters. In addition, the impacted counties with different social equity conditions experienced different levels of damage and different speeds of recovery. The communities with higher percentages of socially vulnerable populations experienced relatively higher level of damage and required longer time for recovery. While some of the findings were discovered in other literature (e.g., Krause and Reeves 2017, Emrich et al. 2019, Constible 2018) and in the context of other disasters, this study offers a data-driven understanding by integrating social media data with traditional data and providing synthesized data analysis results that further explore and reinforce the knowledge of infrastructure resilience and social equity in disasters.

# CHAPTER 4 A SOCIAL WELFARE BASED INFRASTRUCTURE RESILIENCE ASSESSMENT FRAMEWORK: TOWARD EQUITABLE RESILIENCE FOR INFRASTRUCTURE DEVELOPMENT

## 4.1 Introduction

From meeting everyone's basic needs to supporting trade, economy, and technology advancement, infrastructure services are the key enablers of human well-being and development. With climate change and the growth in intensities and frequencies of natural hazards, there is an increasing urgency and priority on investing in and developing resilient infrastructure (Hallegatte et al. 2019). A resilient infrastructure with high quality and robust structural components can potentially limit the impacts from natural hazards in terms of physical damage, economic losses, and functional disruptions (Braese et al. 2019). Over the last two decades, significant efforts have been made for the investment, development, and maintenance of resilient infrastructure to better withstand, adapt to, and rapidly recover from disaster impacts. However, in the context of a disaster, large disparities may exist in the levels of damage and/or recovery processes of the infrastructure across various communities. Such disparities may be caused by the differences in the severity of disaster exposure, and it may also be caused by the variations in the quality and adequacy of infrastructure services across different communities (Coleman et al. 2020). Some communities (e.g., wealthier communities) may have more investment in the development and rehabilitation of existing infrastructure (Hirsch et al. 2016, Nexus 2017). In contrast, some disaster vulnerable communities, which refer to those communities that suffer from the most severe disaster impacts, may struggle with unmet infrastructure needs, such as unreliable electric power systems, lack of adequate water and sanitation systems, overstrained transportation networks, and degraded school buildings, even before the disaster (Huang and Taylor 2019, Hallegatte et al. 2019). Underinvestment, insufficient maintenance, and mismanagement are some of the key factors that result in inadequate infrastructure services in these disaster vulnerable communities (Hallegatte et al. 2019). In addition, research (e.g., Hallegatte et al. 2019, SMASHA 2017) shows that such disaster vulnerability is associated with social vulnerability. Socially vulnerable communities may include those with higher percentages of economically disadvantaged, racial and ethnic minorities, elderly, uninsured, homeless, disabled, those with chronic health conditions, and those with language barriers (Rao et al. 2019, AJMC 2006). These communities often have the fewest resources for disaster preparedness, are located in disaster-prone areas, and lack social, political, and economic capital needed to withstand, adapt to, and recover from a disaster. As a result, they are more likely to suffer from severe disaster impacts (e.g., higher percentages of power outages and traffic disruptions, longer recovery time) (Hallegatte et al. 2019, SAMSHA 2017). Due to the unequal distributions of disaster impacts and potentially more severe impacts on the infrastructure of vulnerable communities, there is sorely a need to systematically integrate disaster inequality and vulnerability with infrastructure resilience assessment.

Despite such need, we have identified a number of knowledge gaps in the domain of infrastructure resilience assessment. Over the last two decades, many research studies (e.g., Panteli et al. 2017, Tonn et al. 2020, Cimellaro et al. 2010, Mao and Li 2018, Yang et al. 2018) have focused on developing models or frameworks to measure or assess infrastructure resilience. Various approaches or methods have been used in resilience assessment, such as simulation-based approaches (e.g., Hossain et al. 2019), mathematical approaches (e.g., Cimellaro et al. 2010), indexbased approaches (e.g., Fisher and Norman 2010), and data-driven approaches (e.g., Zhu et al. 2017). These studies have provided valuable contributions toward advancing the understanding and facilitating infrastructure resilience. However, there remains limited research that integrates the disparity and vulnerability in disaster impacts with infrastructure resilience assessment. In another word, there is a lack of study that (1) measures the unequal distributions of disaster impacts (e.g., infrastructure functional loss, infrastructure recovery time) across different communities and

potentially more severe impacts on vulnerable communities, and (2) investigates how they would impact the collective resilience of infrastructure that serves multiple communities.

To address these knowledge gaps, we propose a Social-Welfare-Based Infrastructure Resilience Assessment (SW-Infra-RA) model that assesses the collective resilience of infrastructure serving multiple communities by accounting for (1) disaster inequality – the unequal distribution of disaster impacts on infrastructure across the various communities, and (2) disaster vulnerability – the higher severity of disaster impacts on infrastructure serving vulnerable communities. The proposed model is theoretically grounded on the social welfare theory and social welfare functions. It also adapts the methods from Bruneau et al. (2003)'s Resilience Triangle framework and Cutter et al. (2003)'s Social Vulnerability Index. The proposed model aims to address the following research questions: How to quantitatively measure the unequal distributions of disaster impacts on infrastructure across different communities? How to quantitatively measure the potentially more severe disaster impacts on infrastructure of vulnerable communities? How to mathematically integrate the disparity and vulnerability in disaster impacts with infrastructure resilience assessment? This chapter focuses on presenting and discussing the conceptual notions and mathematical functions in the SW-Infra-RA model. The remainder of the chapter first reviews and discusses the relevant literature. It then presents the SW-Infra-RA model, including all the mathematical functions in the model. At the end, it discusses two sets of case studies (including a hypothetical and a real case study) to illustrate the use of the SW-Infra-RA model in determining the collective resilience of infrastructure serving multiple communities.

#### 4.2 Proposed Infrastructure Resilience Evaluation Framework

The proposed SW-Infra-RA model aims to define the collective resilience of infrastructure that serves multiple communities by integrating (1) disaster inequality – the unequal distributions of disaster impacts on infrastructure across the various communities, and (2) disaster vulnerability – the higher severity of disaster impacts on infrastructure serving vulnerable communities. The

framework is grounded in the social welfare theory and functions. It also adapts the methods from Bruneau et al. (2003)'s Resilience Triangle framework and Cutter et al. (2003)'s Social Vulnerability Index. The model assesses the collective resilience of infrastructure in five main steps, including (1) determining disaster impacts on individual communities, (2) modeling inequality of disaster impacts, (3) modeling vulnerability in disaster impacts, (4) measuring collective disaster impacts, and (5) assessing collective infrastructure resilience. The following sections discuss about each step in detail.

## 4.2.1 Determining Disaster Impacts on Individual Communities

Disasters may cause severe damage to infrastructure, which results in the reduction of its functionality, and it may take weeks or months to restore the infrastructure to its original functional level. According to Bruneau et al. (2003)'s Resilience Triangle framework, such characterization of infrastructure performance during a disaster leads to a broader conceptualization of resilience. Resilience can be understood as the ability of infrastructure (1) to reduce the possibility or extent of disaster impacts, and (2) to recover rapidly after a disaster (Bruneau et al. 2003). Such conceptualization of resilience is widely adopted in different disaster literature (e.g., Cimellaro et al. 2010, Rehak et al. 2019, Yang et al. 2018). Benchmarking the Resilience Triangle framework, two main types of indicators were identified to determine the disaster impacts on individual communities. These indicators include those that represent (1) the functional loss of infrastructure (e.g., percentage of power outages, percentage of road closures), and (2) the recovery time of infrastructure (e.g., time required to resume electric power services, time required to resume road and highway services).

Depending on the time of analysis, the selected disaster, the level of analysis (e.g., state level, county level, city level, community level), and data availability, there are two main methods for collecting the data for these indicators (i.e., functional loss and recovery time of infrastructure). For analyzing infrastructure resilience in the context of historical disasters, we can extract the relevant data that are available in public or private sources; the data can be collected directly from (1) public sources, such as state, county, or local Department of Emergency Management, Department of Transportation, Office of Communications Commission, or Office of Insurance Regulations, or (2) private sources, such as electric power companies and telecommunication companies. The data then need to be tabulated by the level of analysis (e.g., state, county, city, community levels). For analyzing infrastructure resilience in the context of ongoing disasters, we need to collect firsthand data on infrastructure damage and recovery works by following damage assessment procedures and using the relevant tools and methods. For example, according to Federal Emergency Management Agency (FEMA)'s Preliminary Damage Assessment Guide (FEMA 2021), damage information of infrastructure needs to be captured by visually and technically inspecting and confirming the conditions of damaged infrastructure and identifying and documenting relevant disaster impacts (FEMA 2021).

In general, damage assessment is conducted using either a rapid approach or a detailed approach (Kwasinski 2011; Massarra 2012). Rapid damage assessment usually takes place as soon as conditions allow inspectors to operate after the occurrence of a disaster. It aims to generally estimate the nature and magnitude of damage and quickly inspect and assess the damage conditions. Thus, rapid assessment typically relies on an exterior observation and investigation of the structures. The magnitude of damage recorded on damage assessment forms (e.g., FEMA 2021) is typically a general estimate of the percentage of damage without accurate measurements (Massarra 2012). In recent years, many technologies have been proposed to facilitate the efficiency of rapid damage assessment. For example, GIS-based hazard modeling platforms (e.g., HAZUS) can be used to estimate potential damage from disasters, such as hurricanes and floods (CCSF 2021). Remote sensing technologies, which detect and monitor the physical characteristics of an area by measuring its reflected and emitted radiation from a certain distance, can be used to quickly estimate locations, causes, and severity of disaster damage conditions (Hao et al. 2020). If more

detailed information is required regarding the damage conditions, rapid assessment should be followed by detailed assessments. Detailed damage assessment usually takes place in about two to four weeks after the occurrence of a disaster (Massarra 2012). Detailed damage assessment aims to collect more thorough and accurate information regarding the impacts of a disaster, including estimation of loss value, determination of recovery progress, and identification of recovery needs (Planitz 1999). Detailed damage assessment is based on the inspection of both structural (e.g., girder, column) and non-structural components (e.g., railing, coating) of infrastructure (Massarra 2012). In our research context, both rapid and detailed assessment can be used to collect the data for determining disaster impacts on individual communities. The selection of the methods depends on the level of details that is needed for infrastructure resilience analysis.

#### 4.2.2 Modeling Inequality in Disaster Impacts

In our research context, disaster inequality refers to the unequal distributions of disaster impacts (i.e., functional loss, recovery time) on infrastructure of various communities. The unequal distribution of disaster impacts is analogous to the welfare inequality in a society, which is commonly measured through the Gini coefficient (Atkinson and Brandolini 2010). Thus, we adapted Gini coefficient into the domain of infrastructure resilience assessment to measure the unequal distributions of disaster impacts (i.e., functional loss and recovery time) on infrastructure that serves multiple communities.

A Gini coefficient ranges from 0 to 1. A Gini coefficient of 0 means complete equality in disaster impacts – the infrastructure of all communities of analysis has the same level of functional loss, and/or it takes the same length of time for recovery. A Gini coefficient of 1 means complete inequality in disaster impacts – the infrastructure of only one community has the highest level of functional loss and spends the longest time for recovery. Graphically, Gini coefficient can be represented through the Lorentz curve (Figure 4-1). As per Figure 4-1, it is measured by dividing the area between the Lorentz curve and line of complete equality (i.e., Area X) by the area covered

under the line of complete equality (i.e., Area X+Y) (Wodon and Yitzhaki 2008, Mostafa and El-Gohary 2014). In the SW-Infra-RA model, the Lorentz curve illustrates the percentage of cumulative infrastructure functional loss (or recovery time) experienced by the percentage of communities in analysis. For example, as per Figure 4-1, a point on the Lorenz curve represents a statement such as, "the bottom 40% of all communities suffered from 10% of the total disaster impacts (e.g., functional loss, recovery time)". A Lorentz curve is always bowed downward from the line of equality or coincides with the line of equality if there exists complete equality among the individuals of analysis. The Lorentz curve being farther away from the line of equality indicates a higher level of inequality (i.e., the value of Gini coefficient becomes closer to 1) and vice-versa.





The Gini Coefficient can also be defined through Eqs. (4-1) and (4-2), which are mathematically equivalent to the Lorenz Curve. Eq. (4-1) and Eq. (4-2) define the Gini coefficients that measure the unequal distributions of functional loss and recovery time (i.e., two main indicators of disaster impacts), respectively.
$$G_{k(FL)} = \frac{\sum_{i=1}^{n} \sum_{j=1}^{n} |FL_{ik} - FL_{jk}|}{2n \sum_{i=1}^{n} FL_{ik}}$$
(4-1)

where  $G_{k(FL)}$  = Gini coefficient for functional loss of a group *k* of multiple communities;  $FL_{ik}$  = functional loss of infrastructure in an individual community *i* of group *k*;  $FL_{jk}$  = functional loss of infrastructure in an individual community *j* of group *k*; and n= total number of communities in group *k*.

$$G_{k(RT)} = \frac{\sum_{i=1}^{n} \sum_{j=1}^{n} |RT_{ik} - RT_{jk}|}{2n \sum_{i=1}^{n} RT_{ik}}$$
(4-2)

where  $G_{k(RT)}$  = Gini coefficient for recovery time of a group *k* of multiple communities;  $RT_{ik}$  = recovery time of infrastructure in an individual community *i* of group *k*;  $RT_{jk}$  = recovery time of infrastructure in an individual community *j* of group *k*; and n= total number of communities in group *k*.

#### 4.2.3 Modeling Vulnerability to Disaster Impacts

In our research context, vulnerable communities in a disaster refer to those communities that suffer from the most severe impacts from a disaster. The concept of vulnerability to disaster impacts is analogous to the concept of poverty in welfare economics. Thus, benchmarking the methods for measuring poverty in welfare economics, we proposed a "line of vulnerability" to define and measure vulnerability in the SW-Infra-RA model. The line of vulnerability is a benchmark that indicates the vulnerability level of infrastructure serving different communities. If the value of disaster impacts (i.e., infrastructure functional loss and recovery time) is above the line of vulnerability, the community is identified as one of the vulnerable communities in the disaster. However, unlike the poverty line that has been extensively studied, there are no established methods to measure the line of vulnerability in the disaster domain. In our research, we adapted Cutter et al. (2003)'s work on social vulnerability. Cutter et al. (2003) constructed a Social Vulnerability Index (SoVI) for all the counties in the U.S. based on the county-level socioeconomic and demographic data. The counties with SoVI scores greater than the average plus standard deviation are identified as the most vulnerable counties. In our proposed model, the line of vulnerability can be defined as the sum of the mean and the standard deviation of infrastructure functional loss (or recovery time) experienced by the communities of analysis. Eq. (4-3) and Eq. (4-4) define the line of vulnerability for functional loss and recovery time, respectively:

$$LV_{(FL)_k} = \frac{1}{n} \sum_{i=1}^{n} FL_{ik} + \alpha S_{nk}$$
(4-3)

where  $LV_{(FL)_k}$  = line of vulnerability for functional loss of infrastructure serving a group k of multiple communities;  $FL_{ij}$  = functional loss of infrastructure serving an individual community i of group k; n = total number of communities in group k;  $S_{nk}$  = standard deviation for the functional losses of infrastructure serving a group k of multiple communities; and  $\alpha$  = a coefficient that controls the line of vulnerability ( $0 \le \alpha \le 1$ ).

$$LV_{(RT)_{k}} = \frac{1}{n} \sum_{i=1}^{n} RT_{ik} + \beta S_{nk}$$
(4-4)

where  $LV_{(RT)_k} =$  line of vulnerability for recovery time of infrastructure serving a group *k* of multiple communities;  $RT_{ik} =$  recovery time of infrastructure serving an individual community *i* of group *k*; n = total number of communities in group *k*;  $S_{nk} =$  standard deviation for the recovery time of infrastructure serving a group *k* of multiple communities; and  $\beta =$  a coefficient that controls the line of vulnerability ( $0 \le \beta \le 1$ ).

Depending on the context of analysis, users of the model have the flexibility to define and control the line of vulnerability through the coefficients of  $\alpha$  and  $\beta$ . If the value of  $\alpha$  (or  $\beta$ ) is close to 0, the line of vulnerability is close to the average value. This means the criterion or benchmark for vulnerability is stringent, i.e., approximately half of the communities whose damage (or recovery time) is above the average will be accounted as vulnerable communities. If the value of  $\alpha$  (or  $\beta$ ) is close to 1, the line of vulnerability is close to the average value plus standard deviation. This means the criterion or benchmark for vulnerability is loose as a relatively smaller number of communities will be accounted as vulnerable communities. Defining such line of vulnerability is important in identifying those communities that experience the most severe impacts during a disaster, and this could allow decision makers to prioritize efforts and investments on those communities in disaster assistance, recovery, and/or future mitigation efforts.

# 4.2.4 Measuring Collective Disaster Impacts

The SW-Infra-RA model measures the collective disaster impacts on infrastructure that serves multiple communities based on the distribution of impacts among individual communities. If we want to reduce the overall impact of a disaster, attention must be given to improve the overall equity and to reduce sensitivity of vulnerable communities to disasters (Nicholson 2014). Previous studies (e.g., Tselios and Tompkins 2019, Ward and Shively 2016) also show that higher inequality is associated with worse losses from disasters. Thus, when modeling the collective disaster impacts, we can assume that inequality and vulnerability are both unfavorable situations. Inequality and vulnerability will then be accounted as factors that will further augment the collective disaster impacts.

In the SW-Infra-RA model, the function of collective disaster impacts includes the Collective Functional Loss (CFL) function (Eq. (4-5)) and the Collective Recovery Time (CRT) function (Eq. (4-6)). Both functions incorporate the unequal distributions of disaster impacts on the infrastructure serving multiple communities and the potentially severe impacts on infrastructure in the vulnerable communities. These two functions are developed by adapting the social welfare functions (e.g., Mostafa and El-Gohary 2014, Zhang and Sanake 2020). The equation for the CFL function is presented as:

$$CFL_{k} = \frac{1}{n} \sum_{i=1}^{n} FL_{ik} \times \left(1 + \gamma G_{k(FL)}\right) + \delta \frac{1}{n} \sum_{i=1}^{n} \max[0, (FL_{ik} - LV_{(FL)_{k}})]$$
(4-5)

where  $CFL_k$ = the collective functional loss of the infrastructure that serves a group k of multiple communities;  $FL_{ik}$  = the functional loss of the infrastructure that serves an individual community iof group k; n = the total number of communities;  $G_{k(FL)}$ = the Gini coefficient for the functional loss of group k;  $LV_{(FL)_k}$ = the line of vulnerability for functional loss of infrastructure serving a group k of multiple communities;  $\gamma$  = a coefficient that controls the degree of accounting for inequality in augmenting the disaster impacts ( $0 \le \gamma \le 1$ ); and  $\delta$  = a coefficient that controls the degree of accounting for vulnerability in augmenting the disaster impacts ( $0 \le \delta \le 1$ ).

Similarly, the equation for CRT function is presented as:

$$CRT_{k} = \frac{1}{n} \sum_{i=1}^{n} RT_{ik} \times \left(1 + \lambda G_{k(RT)}\right) + \mu \frac{1}{n} \sum_{i=1}^{n} \max[0, (RT_{ik} - LV_{(RT)_{k}})]$$
(4-6)

where  $CRT_k$ = the collective recovery time of the infrastructure that serves a group k of multiple communities;  $RT_{ik}$  = the recovery time of the infrastructure that serves an individual community iof group k; n = the total number of communities;  $G_{k(RT)}$ = the Gini coefficient for the recovery time of group k;  $LV_{(RT)_k}$ = the line of vulnerability for recovery time of infrastructure serving a group kof multiple communities;  $\lambda$  = a coefficient that controls the degree of accounting for inequality in augmenting the disaster impacts ( $0 \le \lambda \le 1$ ); and  $\mu$  = a coefficient that controls the degree of accounting for vulnerability in augmenting the disaster impacts ( $0 \le \mu \le 1$ ).

Each of the CFL and the CRT functions consists of a subfunction for inequality and a subfunction for vulnerability. The inequality subfunction penalizes the unequal distributions of disaster impacts across different communities. In another word, inequality further augments the collective disaster impacts on these communities. The inequality is measured through the Gini Coefficient [ $G_{k(FL)}$ ,  $G_{k(RT)}$ ] using Eq. (4-1) or Eq. (4-2). Additionally, a coefficient  $\gamma$  (or  $\lambda$ ) is introduced to allow users to adjust the degree of penalizing unequal distributions of disaster

impacts. The value of  $\gamma$  (or  $\lambda$ ) ranges from 0 to 1, where  $\gamma$  (or  $\lambda$ ) =1 represents the full extent of penalization, and  $\gamma$  (or  $\lambda$ ) = 0 represents no penalization at all. Thus, users of the model have the flexibility in determining to what extent they want to account for the inequality factor in infrastructure resilience assessment.

The vulnerability subfunction acknowledges that the potentially severe disaster impacts on the infrastructure of vulnerable communities could compromise the overall infrastructure resilience and should be penalized when assessing the collective resilience of infrastructure. In another word, more severe impacts on some vulnerable communities further augment the collective disaster impacts on all communities of analysis. In this function, a coefficient  $\delta$  (or  $\mu$ ) is introduced, and it allows users to control the degree of accounting for vulnerability in collective disaster impacts. The value of  $\delta$  (or  $\mu$ ) ranges from 0 to 1, where  $\delta$  (or  $\mu$ ) =1 represents the full extent of penalization, and  $\delta$  (or  $\mu$ ) = 0 represents no penalization at all. Thus, users may have the flexibility in determining to what extent they want to account for the vulnerability factor in infrastructure resilience assessment.

## 4.2.5 Assessing Collective Infrastructure Resilience

The collective infrastructure resilience assessment function aims to measure the collective infrastructure resilience based on the collective disaster impacts – collective functional loss and collective recovery time. The function was developed by adapting Bruneau et al. (2003)'s Resilience Triangle framework.

Benchmarking Bruneau et al. (2003), the SW-Infra-RA model measures the infrastructure resilience by defining and measuring the area of a collective resilience triangle (Figure 4-2). In the collective resilience triangle, the vertical axis of the triangle represents the collective functionality of infrastructure, which varies over time. The collective functionality of infrastructure ranges from 0% to 100%, where 100% means no degradation in functions or services and 0% means no service is available. A disaster occurring at time  $t_0$  could cause damage to the infrastructure that the

functionality of the infrastructure immediately reduced. The extent to which the functionality is reduced can be measured by the CFL function [Eq. (4-5)]. The recovery of the infrastructure is a process that takes time, and the infrastructure is completely restored to the original functional level when it is time  $t_t$ . The collective length of recovery (from time  $t_0$  to  $t_t$ ) can be measured through the CRT function [Eq. (4-6)].



Figure 4-2 A conceptual diagram for a collective resilience triangle (adapted from *Bruneau et al.* 2003).

To measure the area of the collective resilience triangle, the collective loss of infrastructure

resilience can be measured through Eq.(4-7):

$$CLR_{k} = \int_{t_{0}}^{t_{t}} (CFL_{k}) dt$$
(4-7)

where  $CLR_k$  = the collective loss of resilience of infrastructure that serves a group k of multiple communities;  $CFL_k$  = the collective functional loss of infrastructure that serves a group k of multiple communities;  $t_0$  = time at which a disruptive event occurs; and  $t_t$  = time at which the infrastructure is fully recovered.

If we assume the infrastructure is recovered in a steady pace, the collective loss of infrastructure resilience (CLR) function can be further simplified, as shown in Eq. (4-8)

$$CLR_k = \frac{CFL_k \times CRT_k}{2} \tag{4-8}$$

where  $CLR_k$  = the collective loss of resilience of infrastructure that serves a group k of multiple communities;  $CFL_k$  = the collective functional loss of infrastructure that serves a group k of multiple communities; and  $CRT_k$  = the collective recovery time of infrastructure that serves a group k of multiple communities.

As per Eq. (8), a higher value of collective loss of resilience indicates poorer resilience performance of the infrastructure against disasters. In other words, the infrastructure is more likely to experience severe damage, resulting in longer disruptions to the functions and services of infrastructure.

# 4.3 Case Studies

### 4.3.1 Hypothetical Case Study

A hypothetical case study was first conducted to illustrate the use of the SW-Infra-RA model in assessing and comparing collective infrastructure resilience across different communities. Hypothetical case studies have been widely used in research in different domains to evaluate or illustrate the use of new methods, models, or frameworks (Balaei et al. 2018, Mostafa and El-Gohary 2014, Zhang and Sanake 2020). This case study aims to analyze and compare the collective resilience of transportation infrastructure in two cities that are composed of various neighborhoods. In this process, we account for the inequality in and vulnerability to disaster impacts among these neighborhoods.

In the case study, Hurricane X caused major damage to the highway infrastructure of City A and City B, which were designed as two hypothetical cities that were composed of twenty neighborhoods each. The highway infrastructure (e.g., roads, highways, bridges) of both cities connects the neighborhoods and supports socioeconomic development of the local communities. In the event of a disaster, the highway infrastructure plays a vital role by offering links to emergency services, relief, and evacuation routes. The highway infrastructure, however, were in different conditions in City A and City B before struck by Hurricane X. According to a report on the quality of highway pavement and bridges of City A, approximately 22% of pavement in City A's highway infrastructure was in "poor" pavement ride quality, and around 17% of bridges were inspected as "structurally deficient". The majority of the pavement and bridges in poor conditions are located in the neighborhoods with lower average household income. Further investigation found that these neighborhoods received less financial support on maintaining, repairing, or rehabilitating their highway infrastructure over the last decade. For City B, the report shows that only 4% of pavement in the city was in "poor" pavement ride quality, while 56% was in "good" quality, and 40% was in "fair" quality. Similarly, only 6% of bridges were inspected as "structurally deficient". The generally good performance of highway infrastructure in City B is attributed to the higher financial support and expenses on maintenance and repair, which may be partially due to relatively better socioeconomic backgrounds (e.g., higher average income, higher housing prices, higher percentages of educated population) of all the neighborhoods in City B.

In the event of Hurricane X, the highway infrastructure of both City A and City B suffered from severe disaster impacts, such as strong wind forces, storm surges, and flash flooding. The damage on the roads, highways, and bridges ranged from pavement failures or structural damage to completely washed off road sections, which resulted in road and highway closures lasting days to weeks. In City A, disparities on the road and highway damage were observed across the twenty neighborhoods. Some neighborhoods suffered from more severe impacts on their highway infrastructure. The roads and bridges were blocked, damaged, or partially washed away due to fallen trees, flying debris, strong storm surges, and flash floods. The road and highway services were disrupted and took three to four weeks to repair before resuming normal operation. On the other hand, some neighborhoods had relatively mild damage, such as erosion of road pavement, poles, and trees fallen down on roads. After removing the debris and repairing the damaged roadway segments, the highway infrastructure resumed its normal function. In City B, the road and highway infrastructure across all neighborhoods experienced a similar level of disaster impacts. Table 4-1 presents the hypothetical data on the disaster impacts on the highway infrastructure in the twenty neighborhoods of each city. The data include (1) the percentage of road closures (functional loss), and (2) the time required to resume road services (recovery time).

	City A		City B				
Neighborhood	Percentage of road closures (FL) (%)	Time required to resume road services (RT) (days)	Neighborhood	Percentage of road closures (FL) (%)	Time required to resume road services (RT) (days)		
А	78	20	а	67	19		
В	68	19	b	65	19		
С	85	24	с	62	18		
D	82	20	d	65	18		
E	96	24	e	72	12		
F	6	8	f	63	16		
G	20	17	g	46	13		
Н	14	9	h	42	13		
Ι	10	8	i	44	17		
J	21	8	j	45	12		
K	95	24	k	51	16		
L	92	19	1	57	18		
М	97	23	m	58	15		
N	88	18	n	54	12		
0	97	18	0	58	11		
Р	94	19	р	52	15		
Q	18	10	q	56	15		
R	28	9	r	49	10		
S	13	13	S	60	14		
Т	44	6	t	61	17		

Table 4-1 Functional loss and recovery time of highway infrastructure of City A and City B.

Utilizing the dataset from Table 4-1, we followed five steps to assess the resilience of highway infrastructure in Cities A and B. In Step 1, normalization of the values of functional loss (percentage of road closures) and recovery time (time required to resume road services) was conducted to ensure that their units and scales are comparable (the values range between 0 to 1 after normalization). In Step 2, the Gini coefficients of functional loss and recovery time were determined through the Lorentz curve. The Lorentz curves that represent the distributions of road closures and time required for road reopening across the twenty neighborhoods in Cities A and B are depicted in Figure 4-3 and Figure 4-4, respectively.



Figure 4-3 Lorenz curves for the distribution of disaster impacts in City A due to Hurricane X (a) A Lorenz curve for the distribution of road closures across neighborhoods in City A due to Hurricane X. (b) A Lorenz curve for the distribution of time required to resume road services across neighborhoods in City A due to Hurricane X.

The results of Gini coefficients are summarized in Table 4-2. As per Table 4-2, although the average disaster impacts on the highway infrastructure were found to be similar for both cities, higher inequality in disaster impacts was found in City A. In Step 3, the lines of vulnerability for disaster impacts on highway infrastructure were determined using Eqs. (3) and (4). In Step 4, the collective disaster impacts on highway infrastructure in Cities A and B were calculated using Eqs. (5) and (6). A 0.5 coefficient  $\gamma$  (and  $\lambda$ ) was employed for the analysis; it represents a medium extent of penalization on the unequal distributions of disaster impacts on highway infrastructure across the twenty neighborhoods in each city. Similarly, a 0.5 coefficient  $\delta$  (and  $\mu$ ) was used; it represents a medium extent of accounting for the severe impacts on the vulnerable neighborhoods in collective impact analysis. In Step 5, the collective resilience of highway infrastructure for Cities A and B were calculated using Eq. (8). The results of lines of vulnerability, collective disaster impacts, and collective loss of resilience for Cities A and B are summarized in Table 4-2.



Figure 4-4 Lorenz curves for the distribution of disaster impacts in City B due to Hurricane X. (a) A Lorenz curve for the distribution of road closures across neighborhoods in City B due to Hurricane X. (b) A Lorenz curve for the distribution of time required to resume road services across neighborhoods in City B due to Hurricane X

These results indicate that the overall resilience performance of highway infrastructure of City B is better than that of City A as lower value in loss of resilience represents better resilience performance. The results imply that, collectively, the highway infrastructure of City B had less damage and was more likely to resume its services within a short period of time. It is worth noticing that although the disaster impact data for the twenty neighborhoods of each city have similar average value, City A receives a higher score on collective loss of resilience by using the SW-InfraRA model. This is mainly due to the inequality or unequal distributions of disaster impacts across different neighborhoods of City A. As per Table 4-1, neighborhoods M, O, P of City A had much higher percentages of road closures and also required almost three weeks to fully resume road services, whereas neighborhoods like F, I, S of City A had minimum road closures and resumed road services within 8 to 10 days. Furthermore, the low resilience performance of City A could be attributed to the severe impacts on the highway infrastructure of some vulnerable neighborhoods in City A, which further augments the collective disaster impacts. For example, a high percentage of road pavement and bridges in neighborhoods E, K, and M of City A were in poor conditions even before the strike of Hurricane X. Hurricane X further damaged these roads and bridges that were in vulnerable conditions, resulting in longer time for resuming road services. Thus, more neighborhoods were accounted as vulnerable neighborhoods as the disaster impacts on these neighborhoods from Hurricane X were above the line of vulnerability.

	City	γA	City B			
Parameter	Functional	Recovery	Functional	Recovery		
	loss	time	loss	time		
Gini coefficient	0.38	0.35	0.32	0.31		
Line of vulnerability	0.76	0.72	0.62	0.71		
Collective disaster impact	0.71	0.68	0.61	0.67		
Collective loss of	0.2	24	0.19			
resilience						

Table 4-2 Results of resilience assessment of highway infrastructure of City A and City B

# 4.3.2 Hurricane Michael Case Study

A real case study on Hurricane Michael was conducted to assess the collective resilience of electric power systems in twelve counties in the Florida Panhandle region. Hurricane Michael was a Category 5 hurricane that made landfall in Florida Panhandle region on October 10, 2018 (NOAA 2019). Twelve counties (Figure 4-5) in this region issued disaster declarations, including Bay, Calhoun, Franklin, Gadsden, Gulf, Holmes, Jackson, Leon, Liberty, Taylor, Wakulla, and Washington Counties. Hurricane Michael brought devastating winds and strong storm surges to

these counties, and it caused massive damage and destruction to the infrastructure of the local communities (NOAA 2019). According to a report (NOAA 2019), the inundation height due to the storm surges was estimated to be 9 to 14 feet above ground level in the Florida Panhandle region. It was estimated that Hurricane Michael caused \$18.4 billion in damage, primarily incurred due to damage on infrastructure (NWS 2018). The strong wind forces and storm surges caused damage to power substations, resulting in power outages lasting for weeks (FPSC 2021). The physical structures, such as utility poles and transmission towers, were severely damaged and destroyed due to fallen trees, flying debris, and flash flood (Dhakal et al. 2021, Pathak et al. 2020).



FEMA-4399-DR, Florida Disaster Declaration as of 11/15/2018

Figure 4-5 Counties of Florida Panhandle region that issued disaster declarations (FEMA 2018).

In this case study, we selected two disaster impact indicators for analysis: (1) percentage of electric power outages, and (2) time required for resuming electric power services. The data on electric power systems of the twelve Florida counties that issued disaster declarations were collected from Florida Public Service Commission (FPSC 2021). The data are summarized in Table

4-3. As per Table 4-3, Hurricane Michael caused disproportionate impacts to the electric power systems of the twelve counties. For example, Calhoun, Gulf, and Jackson, and Washington Counties suffered from more severe impacts on their electric power systems, with power outages ranging from 96.19% to 100%. It took more than three weeks for these counties to fully resume their electric power services (FPSC 2021). On the other hand, Taylor County had a relatively lower percentage of power outages (20.14%), and the county was able to resume power transmission and supply rapidly after the hurricane (FPSC 2021).

County	Percentage of electric power outage (FL) (%)	Time required to resume electric power services (RT) (days)				
Bay	96.6	23				
Calhoun	100	27				
Franklin	96.79	7				
Gadsden	92.12	17				
Gulf	99.05	23				
Holmes	93.82	12				
Jackson	99.78	27				
Leon	65.69	14				
Liberty	65.94	17				
Taylor	20.14	2				
Wakulla	93.49	14				
Washington	96.19	24				

Table 4-3 Functional loss and recovery time of electric power infrastructure in Hurricane Michael

By using the data in Table 4-3 and following the five steps as described in the Hypothetical Case Study section, we performed the resilience assessment of the electric power system of the twelve disaster impacted counties in three contexts. In Context I, we did not account for inequality in or vulnerability to disaster impacts in infrastructure resilience assessment. Thus, the coefficients of  $\gamma$ ,  $\delta$ ,  $\lambda$ , and  $\mu$  were assigned to 0, and the coefficients of  $\alpha$  and  $\beta$  were assigned to 1. In Context II, we accounted for disaster inequality and vulnerability in a medium extent. Thus, all the

coefficients, including  $\alpha$ ,  $\beta$ ,  $\gamma$ ,  $\delta$ ,  $\lambda$ , and  $\mu$ , were assigned to 0.5. In Context III, we fully accounted for disaster inequality and vulnerability in our analysis. Thus, the coefficients of  $\gamma$ ,  $\delta$ ,  $\lambda$ , and  $\mu$  were assigned to 1, and the coefficients of  $\alpha$  and  $\beta$  were assigned to 0. Figure 4-6 shows the Lorenz curves for the distributions of power outages and time required to resume electric power services. Table 4-4 summarizes the results of the resilience assessment in the three defined contexts.

	Cont	ext I	Conte	ext II	Context III		
Parameter	Functional	Recovery	Functional	Recovery	Functional	Recovery	
Gini coefficient	0.64	0.63	0.64	0.63	0.64	0.63	
Line of vulnerability	1.00	0.93	0.96	0.77	0.81	0.61	
Collective disaster impact	0.81	0.61	1.08	0.83	1.44	1.12	
Collective loss of resilience	0.25		0.4	45	0.81		

Table 4-4 Results of resilience assessment of electric power infrastructure in Hurricane Michael.

As per Table 4-4, the results of collective loss of resilience are 0.25, 0.45, and 0.81 in Context I, II, and III, respectively. These results show that the performance of the model is sensitive to the intensity of accounting for disaster inequality and vulnerability. The model is designed in a way that allows users to flexibly choose the coefficients that controls the intensity of accounting for disaster inequality and vulnerability. For example, if an engineer focuses solely on analyzing the functional loss and recovery time of infrastructure systems without emphasis on inequality and vulnerability among the impacted communities. He/she may assign the coefficients of  $\gamma$ ,  $\lambda$ ,  $\delta$ , and  $\mu$  to 0. Thus, as per Eqs. (5), (6), (7), and (8), the collective loss of resilience of these counties under study would be only based on the average disaster impacts on infrastructure of those counties. On the other hand, if a planner or a mitigation expert would like to consider disaster inequality and vulnerability, which may inform future recovery and mitigation efforts, he/she may choose to assign a relatively high value (e.g., 1) for the coefficients of  $\gamma$ ,  $\lambda$ ,  $\delta$ , and  $\mu$ , thus placing a higher emphasis on the impacts of disaster disparities and vulnerabilities on the collective loss of resilience. In this case, the collective loss of resilience would be augmented as the disproportionate disaster impacts across the counties and the potentially severe disaster impacts on the vulnerable counties are considered as negative factors that could exacerbate the collective loss of resilience. It is thus recommended to use a consistent set of coefficients when assessing the resilience of a set of infrastructure alternatives.



Figure 4-6 Lorenz curves for the distribution of disaster impacts due to Hurricane Michael. (a) A Lorenz curve for the distribution of power outages of Florida Panhandle counties due to Hurricane Michael. (b) A Lorenz curve for the distribution of time required to resume electric power services of Florida Panhandle counties due to Hurricane Michael.

In addition, the results show that, in the case of Hurricane Michael, the impacts of disaster inequality and vulnerability on the collective loss of resilience of electric power infrastructure was relatively high. The unequal distributions of disaster impacts on electric power infrastructure (including both power outages and time required to resume electric power services) across different counties in Hurricane Michael can be observed through the relatively high Gini coefficients ( $G_{FL} = 0.64$  and  $G_{RT} = 0.63$ ). Such disparities in disaster impacts could primarily be caused by the counties' different levels of disaster exposure. In this case study, Hurricane Michael impacted a large geographic region. Counties that are located in close proximity to the hurricane path experienced

much significant wind and storm forces compared to counties that are relatively farther away. For example, counties including Calhoun, Gadsden, Gulf, Jackson, and Washington counties experienced strong storm surges and had an average windspeed of approximately 74 mph (Senkbeil et al. 2020). The percentage of power losses in these counties range from 92.12% to 100%. On the other hand, counties, such as Taylor and Leon counties, had relatively lower average wind speeds of approximately 39 mph and 57 mph, respectively (Senkbeil et al. 2020), and their power losses are 20.14% and 65.69%, respectively. Thus, in a large-scale disaster such as Hurricane Michael, the different levels of disaster exposure are one of the primary reasons that contribute to the disparities in disaster impacts.

Another hidden reason may be the social inequality of these counties. The social inequality situation in Florida is among the worst in the U.S. and has been getting worse over time (Johnson 2019). In our case, some of the counties (e.g., Gadsden, Calhoun, Franklin, Holmes, Jackson counties) whose electric power infrastructure suffered from the most severe impacts are also among the most socially vulnerable counties in the region (CDC/ASTDR 2022). In addition, previous research (Dhakal et al. 2021) has found that counties with different socioeconomic and demographic characteristics (e.g., age, race, income, health) experienced different levels of infrastructure damage and speeds of recovery. Those counties with higher percentages of socially vulnerable populations experienced a relatively higher level of damage and required longer time for recovery (Dhakal et al. 2021). Research on other disasters (e.g., Ward and Shivley 2016, Yoon 2012, Flanagan et al. 2011) also show that disaster vulnerability is interrelated with social vulnerability; many social factors (e.g., age, gender, income, education) may impact the resilience of communities. Under the same level of exposure, communities that are socially vulnerable are more likely to suffer from severe impacts (e.g., higher power outages, traffic disruptions, higher congestion) (Hallegatte et al. 2019). This may be attributed to the fact that socially vulnerable populations often have the fewest resources for disaster preparedness, live in disaster-prone areas, and lack social, political, and economic capital needed to withstand, adapt to, and recover from a disaster (SAMSHA 2017).

# 4.4 Conclusions

This study presents a new social welfare-based infrastructure resilience assessment (SW-Infra-RA) model for assessing the collective resilience of infrastructure serving multiple communities while accounting for inequality in and vulnerability to disaster impacts. The SW-Infra-RA model is theoretically grounded on social welfare theory and social welfare functions. The Gini coefficient was adapted to model unequal distributions of disaster impacts on infrastructure of different communities. The line of vulnerability was proposed and measured by leveraging Cutter et al. (2003)'s work on Social Vulnerability Index to model disaster vulnerability. The collective disaster impact function was then defined by aggregating the impacts on infrastructure of individual communities, while integrating unequal distributions of disaster impacts and potentially more severe impacts on the vulnerable communities. The collective disaster impacts were then considered as input into the collective resilience assessment function, which was developed by adapting Bruneau et al. (2003)'s Resilience Triangle framework. The application of the SW-Infra-RA model was first illustrated through a hypothetical case study, which compares the collective resilience of highway infrastructure of two cities impacted by the same disaster. A real case study was further conducted to illustrate the use of the model for assessing the collective resilience of electric power systems in the context of Hurricane Michael.

# CHAPTER 5 PROTOTYPE DEVELOPMENT OF DECISION SUPPORT SYSTEM TO FACILITATE EQUITABLE INFRASTRUCTURE PLANNING

This chapter presents completed work of Research Task 3- Developing a decision support system to facilitate automatic infrastructure resilience assessment. The following paragraphs present the development of prototype for decision support system.

The proposed system consists of four main modules: (1) disaster inequality module (2) disaster vulnerability module (3) collective disaster impact module, and (4) collective infrastructure resilience module. The main modules of the proposed system are depicted in Figure 5-1. Figure 5-2 shows a flowchart that illustrates the main functions and flow of information of the proposed system.



Figure 5-1 Main modules of the proposed prototype system

The disaster inequality module aims to analyze the unequal distribution of disaster impacts on infrastructure of various communities. The input of this module includes the data of disaster impacts on infrastructure of each individual communities of analysis, which are manually loaded

into the system. The output of the module is disaster inequality, which is represented by the Gini coefficient. A screenshot of the user interface of the disaster inequality module is shown in Figure 5-3.



Figure 5-2 A flowchart of the proposed prototype system

The disaster vulnerability module aims to analyze the potential severe impacts on infrastructure of the vulnerable communities. The input of this module includes the data of disaster impacts on infrastructure of each individual communities of analysis. The output of this module includes (1) the line of vulnerability, and (2) the percentage of vulnerable communities.

The collective disaster impact module aims to quantify and analyze the collective disaster impacts on infrastructure serving multiple communities based on disaster impacts on infrastructure serving individual communities while accounting for disaster inequality and vulnerability. The input of the module is the disaster impacts on infrastructure of each individual communities of analysis, disaster inequality, disaster vulnerability, and coefficients that control the degrees of accounting for disaster inequality and disaster vulnerability. The output of the module is collective disaster impacts. The collective infrastructure resilience module aims to analyze the collective resilience of infrastructure serving multiple communities while accounting for disaster inequality and disaster vulnerability. The input of the module is collective disaster impacts. The output of the module is the collective loss of resilience. A screenshot of the user interface of the collective infrastructure resilience module is shown in Figure 5-4.

A Prototype	Decision S	Support System	n		
Modeling Ineq Disaster Impac	Modeling Inequality in Model Disaster Impacts Disaste		Inerbility to acts	Measuring Collective Disaster Impacts	Measuring Collective Resilience
Data of Disa	ister Impac	ts			
County	% FL	RT (days)	% of Pop.	Lorenz Curves for FL a	and RT
Broward	3.32	1	15.25		
Orange	6.79	8	10.88	100 8 ec	100 1900 1900 100
St. Lucie	5.26	2	2.56	Damet Of	Time Regulation Regulation
Palm Beach	23.38	3	11.69	40	Benutatives and an analysis
Lee	22.97	1	6.09	22 West	
Duval	41.76	3	7.48	0 20 40 60 90 100 % of No. of Counties	0 20 40 60 60 100 % of No. of Counties
Polk	38.63	4	5.66		
Marion	27.97	3	2.86	Gini Coefficent (FL)	0.72
Hendry	44.02	4	10.82	Gini Coefficent (RT)	0.69
Note: FL = F RT = R	unctional L Recovery T	.oss			

Figure 5-3 A user interface of the disaster inequality module



Figure 5-4 A user interface of the collective infrastructure resilience module

The main platform of the proposed prototype system was developed through Figma. Figma is an open source, web-based application that primarily focuses on user interface design utilizing a variety of graphic designing and prototype tools (Kopf 2022). The proposed prototype system was utilized to conduct a comparative analysis to analyze the resilience of infrastructure serving different communities with various characteristics, which is further discussed in Chapter 6.

# CHAPTER 6 RESILIENCE ANALYSIS OF INFRASTRUCTURE SYSTEMS INTEGRATING SOCIAL EQUITY: CASE STUDIES USING HISTORICAL DISASTERS

# 6.1 Introduction

To adapt and thrive in the face of climate change, there has been an increasing trend of adopting resilience thinking and/or strategies in infrastructure planning processes. Assessing and analyzing infrastructure resilience help planners and decision makers better assess the performance of infrastructure and provide the foundation for prioritizing infrastructure investment. However, investigation on previous infrastructure investment and policies shows that there have been discriminatory policies, practices, and embedded biases within infrastructure planning processes for decades (NASEM 2022). As a result, in disasters, large disparities exist in terms of disaster impacts (e.g., physical damage, economic losses, service disruption) on infrastructure across the various communities (Zou et al. 2018b, SAMSHA 2017). Such disparities, also known as disaster inequalities (Dhakal and Zhang 2022), are linked with the variations in infrastructure quality and services across different communities, which may be affected by the various characteristics (e.g., location, population, socioeconomic status) of communities. For example, socially vulnerable communities, which include those with higher percentages of economically disadvantaged, racial and ethnic minorities, elderly, uninsured, homeless, disabled, those with chronic health conditions, and those with language barriers (Rao et al. 2019), typically experience a disproportionate share of physical and economic burden caused by disasters; they may need more time, assistance, support, or investment for coping with and recovering from a disaster (Drakes et al. 2021). However, when assessing the resilience of infrastructure that serves multiple communities, such disparities and vulnerability are often neglected, and the infrastructure of all communities is treated as equally impacted by a disaster. Such resilience assessment may not fully reflect the actual damage level

and/or recovery processes. Thus, there is a need to account for disaster inequalities and vulnerabilities in infrastructure resilience assessment.

A comprehensive literature review in the disaster domain shows that there are two main knowledge gaps. First, research that focuses on analyzing infrastructure resilience does not typically account for disaster inequalities or vulnerabilities. Extensive studies (e.g., Ouyang and Duenas-Osorio 2014, Chan and Schofer 2016, Diao et al. 2016, Pagano et al. 2019, Tonn et al. 2020) have been conducted to assess the resilience of infrastructure of various types, such as transportation infrastructure (e.g., Chan and Schofer 2016), water infrastructure (e.g., Pagano et al. 2019), and electric power infrastructure (e.g., Ouyang and Duenas-Osorio 2014), etc. These studies have offered valuable contributions toward understanding and analyses of infrastructure resilience. However, these studies have not incorporated the varying levels of disaster impacts on infrastructure serving different communities into their resilience assessment. Second, there is a lack of research that focuses on analyzing disaster impacts experienced by communities with different characteristics. Studying disparities in disaster impacts would help understand the existing challenges faced by different communities and recommend infrastructure investment and resilience planning strategies that are prioritized based on the severity of needs. Over the years, research in the disaster domain have focused on assessing disaster impacts and/or emergency management efforts across various communities during disaster preparedness (Kim and Sutley), response (Yabe and Ukkusuri 2020) and recovery (Emrich et al. 2022) phases to better understand disaster inequalities. Assessing the resilience of infrastructure serving multiple communities with various characteristics could potentially help identify the vulnerable communities and facilitate more equitable planning in the future. Currently, there is still limited studies that comprehensively analyze or compare infrastructure resilience across communities with different characteristics and how such characteristics might impact the collective resilience of infrastructure.

To address these gaps, this chapter presents a study that analyzes infrastructure resilience while accounting for (a) disaster inequality: the unequal distribution of disaster impacts on infrastructure across the communities with various characteristics, and (b) disaster vulnerability: the potentially more severe disaster impacts on infrastructure of certain (e.g., socially vulnerabilities) communities (Dhakal and Zhang 2022). By utilizing the social welfare-based infrastructure resilience assessment (SW-Infra-RA) model proposed by Dhakal and Zhang (2022), the analyses focus on exploring and comparing (a) the levels of disaster inequalities, (b) the levels of disaster vulnerabilities, and (c) the collective resilience of infrastructure serving communities with various characteristics (e.g., spatial, demographic, and socioeconomic statuses). This chapter presents four case studies that assesses the resilience of four different types of infrastructure (i.e., electric power, communication service, transportation, and wastewater infrastructure) in the context of three hurricanes. The following sections of the chapter describe the background of the research, present and discuss the results of the case studies, and conclude with the summary and contributions of the study.

# 6.2 Case Study Design

We conducted four case studies in the context of three hurricanes, in order to answer the following research questions:

- (1) How do counties with different characteristics (e.g., coastal vs inland, urban vs rural, more vulnerable vs less vulnerable) compare against each other in terms of disaster inequality?
- (2) How do counties with different characteristics compare against each other in terms of disaster vulnerability?
- (3) How do counties with different characteristics compare against each other in terms of collective resilience?

We designed the four case studies in a similar manner in terms of structure and content. For each case study, by leveraging the SW-Infra-RA, we measured and compared disaster inequalities, disaster vulnerabilities, and collective resilience of infrastructure that serves different groups of communities in the context of a hurricane disaster. Case Study I focuses on communication infrastructure in the context of Hurricane Michael. Case Study II focuses on electric power infrastructure in the context of Hurricane Sally. Case Study III focuses on transportation infrastructure in the context of Hurricane Irma. Case Study IV focuses on wastewater infrastructure in the context of Hurricane Irma.

## 6.2.1 Contexts of Cases studies

We selected three hurricanes for analysis, including Hurricanes Michael, Irma, and Sally. Hurricane Michael was a Category 5 hurricane that made landfall near Mexico Beach in Florida Panhandle on October 10, 2018. It was one of the strongest hurricanes on record, and the maximum sustained windspeed was 161 mph at landfall (NOAA 2019). Hurricane Michael brought catastrophic storm surges ranging from 9 to 14 ft above ground level along the portion of Florida Panhandle coast with the highest inundation occurring in Mexico Beach (NOAA 2019). In addition to life threatening storm surges, Hurricane Michael produced devastating wind gusts that caused extensive structural damage and service disruptions to different infrastructure. For example, the damage to communication service infrastructure was catastrophic (Burgess 2018). Cellular services were knocked down as communication service towers, transformers, and power lines were destroyed by falling trees and flying debris (Burgess 2018). In addition, road and highways in the coastal region were washed out due to flash flooding in several locations between Panama City and Alligator point (NOAA 2019).

Hurricane Irma was a Category 5 hurricane that made landfall in Florida Keys on September 10, 2017. The southern region of Florida, especially Florida Keys experienced severe damage from

Hurricane Irma. The combined effect of strong winds and storm surges brought flood with maximum inundation level of 5 to 8 ft above the normal level for the portions of Lower Florida Keys (NOAA 2018b). According to a damage assessment report (NOAA 2018b), Hurricane Irma caused approximately \$50 billion in damage in the U.S., which made Hurricane Irma the fifth-costliest hurricane to affect the U.S. The infrastructure serving the communities of South Florida experienced the most severe impacts. For example, the strong storm surges and heavy rain from Hurricane Irma washed away part of US 1 highway (Lazo 2017). Most of the road and highways in southern part of Florida were shut down due to flooding and resumed after clearing debris and restoring traffic signals (Lazo 2017). In addition, the floodwater from Hurricane Irma damaged wastewater infrastructure, resulting in the sewage flowing from the wastewater systems to the surrounding water bodies (Gardner 2019). For example, according to a city record of St Augustine, the sewage spill caused the leakage of about 388,116 gallons of waste from the wastewater system to the nearby waterways (Gardner 2019).

Hurricane Sally was a Category 2 hurricane that made landfall along the coast of Alabama (immediately west of the Alabama/Florida state line) on September 16, 2020, with a sustained windspeed of 105 mph. Hurricane Sally's strong winds and storm surges caused extensive damage and destruction across the northwest coastal regions of Florida. Housing and infrastructure in Escambia and Santa Rosa Counties suffered from significant damage by the strong wind and storm surges (NOAA 2021a). Furthermore, strong storm surges along with heavy rainfall brought by Hurricane Sally caused massive flooding and damage to electric power infrastructure mostly in the western Panhandle region of Florida (Saunders 2020). A preliminary report on power outages by an electric power company suggests that hurricane Sally caused power outages for about 285,000 customers (Saunders 2020).

We selected these three hurricanes as our research context for three main reasons. First, all three hurricanes have impacted Florida communities, which are the geographic focus of our study. Second, these disasters caused devastating impacts (e.g., physical damage, service disruptions) on various infrastructure, such as electric power infrastructure, communication service infrastructure, and roadway infrastructure. Third, the regions that are impacted by these disasters have different spatial, demographic, and socioeconomic characteristics that can offer better understanding of in terms of how the characteristics of communities influence the resilience of infrastructure.

# 6.2.2 Community Groups under Analysis

To compare the resilience of infrastructure serving communities with different spatial, demographic, and socioeconomic characteristics, we classified all the counties under analysis into multiple groups, including (1) the coastal and inland groups, (2) the urban and rural groups, and (3) the more vulnerable and less vulnerable groups. In this study, the coastal group include those counties that have a coastline bordering of the ocean or contain coastal high hazards areas (NOAA 2021b). The inland group include those counties that share their borders with other adjacent counties with no coastline bordering. The rural counties are those with population densities lower than 100 individuals per square mile (Florida Health 2022), while the urban counties have population densities higher than 100 individual per square mile. To classify the counties based on their socioeconomic statues, we leveraged the Social Vulnerability Index (CDC/ATSDR 2022) (ranging from 0 to 1), which uses 15 U.S. census variables (e.g., poverty, access to transportation, housing to help identify communities that are vulnerable to disasters (CDC/ATSDR 2022). We then classified the counties into two groups based on SVIs; the less vulnerable group include those counties with SVIs ranging from 0 to 0.5, and the more vulnerable group include those with SVIs ranging from 0.5 to 1.

# 6.2.3 Case Study Data Collection

Based on the SW-Infra-RA model, to conduct the resilience analysis of infrastructure, we need to collect the data for two main types of infrastructure resilience indicators: (1) functional loss of infrastructure, and (2) recovery time of infrastructure. Functional loss is broadly defined as the reduction of functionality of infrastructure due to the impacts of a disaster, and it can be represented through different parameters, such as the percentage of electric power outages, the percentage of communication service outages, the percentage of wastewater system failures, and the cost in road debris removal. Recovery time is defined as the time required to resume infrastructure services to the original functional level, such as the time required to resume electric power and communication services, and the time required for road reopening. The data for both types of indicators (i.e., functional loss and recovery time) were collected from different public and private sources, including Florida Public Service Commission (FPSC), Federal Emergency Management Agency (FEMA), Federal Communications Commission (FCC), and Florida Department of Environmental Protection (FDEP). Table 6-1 summarizes the data we collected for the case study.

Case Study No.	Infrastructure	Disaster context	Infrastructure damage and recovery data	Data source
т	Communication	Hurricane	% Communication service outages	FCC (2018)
1	service	Michael	Time required to resume communication services	FCC (2018)
ч	Electric power	Hurricane	% Electric power outages	FPSC (2020)
11		Sally	Time required to resume electric power services	FPSC (2020)
	Tananatatian	Hurricane	Cost of road debris removal	FEMA (2017)
111	Transportation	Irma	Time required for road reopening	FDOT (2017)
IV/	Westernster	Hurricane	% Sewage spill from wastewater systems	FDEP (2017)
IV	wastewater	Irma	Time required to resume normal functioning of wastewater systems	FDEP (2017)

Table 6-1 List of disaster, infrastructure system and source of infrastructure damage data

# 6.2.4 Case Study Data Analysis

# 6.2.4.1 Data Preprocessing

During disasters (e.g., hurricanes), the geographical regions that are close to the hurricane path naturally experience more severe impacts compared to the regions that are located farther. In our analysis, we want to analyze the distribution of disaster inequality and disaster vulnerability across different counties assuming that all disaster-impacted counties experience the same or similar level of disaster impacts. Thus, to eliminate the effects of disaster threat level on infrastructure serving different counties, we first determined the normalized disaster impacts. The normalized disaster impacts were calculated by dividing the disaster impacts (i.e., functional loss and recovery time) experienced by infrastructure across different counties with the average sustained wind speeds during the disaster period across each impacted county.

## 6.2.4.2 Resilience Assessment

By leveraging the SW-Infra-RA framework developed in Chapter 4, we assessed infrastructure resilience by following four main steps. First, we measured the unequal distributions of disaster impacts on infrastructure serving multiple counties by calculating the Gini coefficients. For the infrastructure serving each group of counties, we calculated the Gini coefficients of functional loss and recovery time using Eqs. ((4-1)1) and (4-2), respectively. Second, we measured the line of vulnerability for functional loss and recovery time, using Eqs. (4-3) and (4-4), respectively. We then identified those counties whose infrastructure suffered from disaster impacts that are above the line of vulnerability as "vulnerable counties" and determined the percentage of vulnerable counties in each group. Third, we determined the collective disaster impacts (i.e., collective functional loss and collective recovery time) on the infrastructure serving multiple counties, using

Eqs. (4-5) and (4-6). Fourth, we quantified the collective loss of resilience of infrastructure serving a group of counties by using Eq (8).

# 6.3 Case Study Results

# 6.3.1 Results of Case Study I

The results of Case Study I are presented in Table 6-2. As per Table 6-2, in terms of comparisons between the coastal counties and inland counties in the context of Hurricane Michael, the Gini coefficients of communication infrastructure serving the group of coastal counties ( $G_{FL}$  =0.93,  $G_{RT}$  =0.84) and inland counties ( $G_{FL}$  =0.92,  $G_{RT}$  =0.92) are generally similar, indicating a similar level of disaster inequality between the coastal group and the inland group. By comparing the disaster impacts on communication infrastructure of each county to the line of vulnerability, the inland group has a higher percentage of vulnerable counties (71%) as compared to the coastal group (62%). In addition, the collective loss of resilience of communication infrastructure serving inland group (0.29) is higher as compared to that of the coastal group (0.16), which implies that the communication infrastructure serving the inland counties had poorer resilience performance against the impacts from Hurricane Michael compared to the one serving the coastal counties.

In terms of comparisons between the urban counties and rural counties, the results show that differences exist in the Gini coefficients for the urban group ( $G_{FL}$  =0.75,  $G_{RT}$  =0.83) and the rural group ( $G_{FL}$  =0.92,  $G_{RT}$  =0.92), and more severe disaster inequality exists in the rural group. There is also a higher percentage of vulnerable counties in the rural group (45%) as compared to that in the urban group (25%). In addition, there is a higher collective loss of resilience for communication service infrastructure serving group of rural counties (0.24) compared to urban counties (0.18).

In terms of comparisons between the two groups of counties with different levels of social vulnerability, the results show that the group with more vulnerable counties ( $G_{FL}$  =0.93,  $G_{RT}$  =0.89) have higher Gini coefficients compared to the group with less vulnerable counties ( $G_{FL}$  =0.72,  $G_{RT}$  =0.52), indicating a higher level of disaster inequality across counties that are more socially

vulnerable. In contrast, there is a higher percentage of disaster vulnerable counties in the less vulnerable group (66%) compared to the more vulnerable group (41%). In addition, the collective loss of resilience is higher in the less vulnerable group (0.37) comparing to the more vulnerable group (0.2).

	Communication service infrastructure												
Parameter	Coastal		Inland		Urban		Rural		More vulnerable		Less vulnerable		
	FL	RT	FL	RT	FL	RT	FL	RT	FL	RT	FL	RT	
Gini coefficient	0.93	0.84	0.92	0.92	0.75	0.83	0.92	092	0.93	0.89	0.72	0.52	
Line of vulnerability	0.56	0.46	0.56	0.46	0.56	0.46	0.56	0.46	0.56	0.46	0.56	0.46	
% Vulnerable counties	62%	12%	71%	28%	25%	25%	45%	36%	41%	25%	66%	66%	
Collective disaster impacts	0.78	0.42	0.87	0.67	0.8	0.46	0.73	0.65	0.74	0.55	0.92	0.81	
Collective loss of resilience	0.16		0.29		0.18		0.24		0.2		0.	37	

Table 6-2 Results for case study I

Notes: FL = functional loss; RT = recovery time

# 6.3.2 Results of Case Study II

The results of Case Study II are presented in Table 6-3. As per Table 6-3, comparing the results between the coastal group and the inland group, the Gini coefficients of electric power infrastructure serving the coastal group ( $G_{FL}$ = 0.74,  $G_{RT}$  =0.77) and the inland group ( $G_{FL}$ = 0.76,  $G_{RT}$  =0.8) are similar. This implies that there is a similar level of disaster inequality across both the coastal and inland counties. Comparing the disaster impacts on electric power infrastructure serving different counties to the line of vulnerability, the inland group has a higher percentage (50%) of vulnerable counties as compared to the coastal group (40%). Furthermore, the collective loss of resilience for electric power infrastructure are similar for both inland (0.18) and coastal counties (0.19), which shows a similar level of resilience performance of electric power infrastructure for both inland and coastal counties in the context of Hurricane Sally.

Comparing the results between the urban and rural counties in the context of Hurricane Sally, the Gini coefficients of electric power infrastructure serving urban group ( $G_{FL}$ = 0.74,  $G_{RT}$  =0.76) and rural group ( $G_{FL}$ = 0.76,  $G_{RT}$  =0.78) are generally similar, indicating a similar level of disaster inequality between the urban and rural counties. As per Table 3, there is a higher percentage of disaster vulnerable counties in the urban group (40%) compared to the rural group (25%). In addition, the collective loss of resilience of electric power infrastructure is higher within the urban counties (0.21) compared to the rural counties (0.15).

In terms of comparisons between the more vulnerable group and the less vulnerable group, the results show that the Gini coefficients are higher ( $G_{FL}=0.86$ ,  $G_{RT}=0.92$ ) within the more vulnerable counties as compared to within the less vulnerable counties ( $G_{FL}=0.6$ ,  $G_{RT}=0.59$ ), which indicates a higher level of disaster inequality in the more vulnerable counties. In addition, the percentage of disaster vulnerable counties is much higher within the more vulnerable group (50%), in comparison to within the less vulnerable group (33%). Overall, the collective loss of resilience of electric power infrastructure serving the more vulnerable (0.2) and the less vulnerable (0.21) groups are similar in the context of Hurricane Sally.

Electric power infrastructure												
Doromotor	Coastal		Inland		Urban		Rural		More vulnerable		Less vulnerable	
i urumeter	FL	RT	FL	RT	FL	RT	FL	RT	FL	RT	FL	RT
Gini Coefficient	0.74	0.77	0.76	0.8	0.74	0.76	0.76	0.78	0.86	0.92	0.6	0.59
Line of vulnerabilit y	0.43	0.52	0.43	0.52	0.43	0.52	0.43	0.52	0.43	0.52	0.43	0.52
% Vulnerable counties	40%	20%	25%	50%	40%	20%	25%	25%	30%	50%	33%	33%
Collective disaster impacts	0.59	0.66	0.51	0.71	0.59	0.7	0.51	0.59	0.5	0.79	0.67	0.63
Collective loss of resilience	0.19		0.18		0.21		0.15		0.2		0.2	21

Table 6-3 Results for case study II

#### Notes: FL = functional loss; RT = recovery time

# 6.3.3 Results of Case Study III

Table 6-4 shows the results of Case Study III. As per Table 6-4, in terms of comparisons between the coastal group and the inland group, the Gini coefficients of the transportation infrastructure serving the coastal counties ( $G_{FL}$ = 0.83,  $G_{RT}$  =0.96) and the inland counties ( $G_{FL}$ = 0.89,  $G_{RT}$  =0.8) are similar. There is a higher percentage of disaster vulnerable counties in the group of inland counties (57%) in comparison to the group of coastal counties (41%). In addition, the transportation infrastructure serving the inland counties (0.18) has a much higher collective loss of resilience compared to the one serving the coastal counties (0.08) in the context of Hurricane Irma.

In terms of comparisons between the urban and rural counties, the Gini coefficients of the urban group ( $G_{FL}$ = 0.95,  $G_{RT}$ =0.94) is higher compared to those of the rural group ( $G_{FL}$ = 0.76,  $G_{RT}$ =0.52), which indicates a higher level of disaster inequality within the urban counties. In contrast, there is a higher percentage of disaster vulnerable counties in the rural group (66%) than in the urban group (25%). In addition, the transportation infrastructure serving the rural counties (0.34) has a much higher collective loss of resilience compared to the one serving the urban counties (0.06) in the context of Hurricane Irma.

In terms of comparisons between the more vulnerable and less vulnerable groups of counties, the Gini coefficients for the more vulnerable group ( $G_{FL}$ = 0.95,  $G_{RT}$ =0.91) is higher compared to the less vulnerable group ( $G_{FL}$ = 0.77,  $G_{RT}$ =0.85). In addition, there is a higher percentage of disaster vulnerable counties within the more vulnerable group (54%) in comparison to the less vulnerable group (50%). Similarly, the transportation infrastructure serving the more vulnerable counties has a higher collective loss of resilience (0.16) compared to the one serving the less vulnerable counties (0.06) in the context of Hurricane Irma.

Transportation infrastructure												
Paramet	Coastal		Inland		Urban		Rural		More vulnerable		Less vulnerable	
	FL	RT	FL	RT	FL	RT	FL	RT	FL	RT	FL	RT
Gini Coeffici ent	0.83	0.96	0.89	0.8	0.95	0.94	0.76	0.52	0.95	0.91	0.77	0.85
Line of vulnerab ility	0.33	0.37	0.33	0.37	0.33	0.37	0.33	0.37	0.33	0.37	0.33	0.37
% Vulnera ble counties	41%	16%	57%	28%	25%	25%	66%	33%	54%	15%	16%	50%
Collecti ve disaster impacts	0.43	0.38	0.55	0.66	0.31	0.36	1.14	0.59	0.77	0.42	0.38	0.31
Collecti ve loss of resilienc e	0.	0.08 0.18		0.06		0.34		0.16		0.0	06	

Table 6-4 Results for case study III

Notes: FL = functional loss; RT = recovery time

# 6.3.4 Results of Case Study IV

The results for Case study IV is presented in Table 6-5. As per Table 6-5, in terms of comparisons between the coastal and inland counties, the Gini coefficients of the wastewater infrastructure serving the coastal group ( $G_{FL}$ = 0.92,  $G_{RT}$  =0.9) and the inland group ( $G_{FL}$ = 0.9,  $G_{RT}$  =0.85) are similar, indicating a similar level of disaster inequality between the coastal and inland counties. There is a much higher percentage of disaster vulnerable counties (60%) in the inland group as compared to in the coastal group (20%). Furthermore, the wastewater infrastructure serving the inland counties (0.14) has a higher collective loss of resilience as compared to the one serving the coastal counties (0.07).

In terms of comparisons between the urban and rural groups, the results show that the Gini coefficients of the urban group ( $G_{FL}$ = 0.94,  $G_{RT}$  =0.95) is higher compared to those of the rural
group ( $G_{FL}$ = 0.8,  $G_{RT}$  = 0.85). There is also a higher percentage of disaster vulnerable counties in the rural group (80%) in comparison to the urban group (26%). In addition, the wastewater infrastructure serving the rural group (0.29) is higher comparing to the one serving the urban group (0.07).

In terms of comparisons between the more vulnerable and less vulnerable groups, the results show that the more vulnerable counties ( $G_{FL} = 0.96$ ,  $G_{RT} = 0.94$ ) have higher Gini coefficients compared to the less vulnerable counties ( $G_{FL} = 0.68$ ,  $G_{RT} = 0.77$ ), which indicates a higher level of disaster inequality within the counties that are more socially vulnerable. In contrast, there is a higher percentage of disaster vulnerable counties in the less vulnerable group (50%) in comparison to the more vulnerable group (37%). In addition, the wastewater infrastructure serving the less vulnerable counties (0.17) has a higher collective loss of resilience compared to the one serving the more vulnerable counties (0.1) in the context of Hurricane Irma.

	Wastewater infrastructure											
Paramet er	Coastal		Inland		Urban		Rural		More vulnerable		Less Vulnerable	
	FL	RT	FL	RT	FL	RT	FL	RT	FL	RT	FL	RT
Gini Coeffici ent	0.92	0.9	0.9	0.85	0.94	0.95	0.8	0.85	0.96	0.94	0.68	0.77
Line of vulnerab ility	0.49	0.29	0.49	0.29	0.49	0.29	0.49	0.29	0.49	0.29	0.49	0.29
% Vulnera ble counties	20%	20%	60%	30%	20%	26%	80%	40%	37%	25%	25 %	50%
Collecti ve disaster impacts	0.42	0.37	0.74	0.38	0.44	0.31	1.17	0.49	0.74	0.27	0.54	0.63
Collecti ve loss of resilienc e	0.07		0.14		0.07		0.29		0.1		0.17	

Table 6-5 Results for case study IV

Notes: FL = functional loss; RT = recovery time

Overall, the results of Gini Coefficients are similar in both the coastal and inland counties across the three hurricane disasters. Higher percentages of disaster vulnerable counties are found in the inland group in Hurricanes Michael and Irma. Similarly, the infrastructure serving the inland counties tend to have higher collective loss of resilience as compared to the one serving the coastal counties in these two hurricanes. However, the context of Hurricane Sally, the coastal group has a higher percentage of disaster vulnerable counties and a higher collective loss of resilience in its infrastructure.

For comparisons between the urban and rural counties, the Gini coefficients vary across the three hurricane disasters. Higher percentages of disaster vulnerable counties are found in the rural group in Hurricanes Michael and Irma. Similarly, the infrastructure serving the rural counties tend to have higher collective loss of resilience as compared to the one serving the urban counties in these two hurricanes. However, the context of Hurricane Sally, the urban group has a higher percentage of disaster vulnerable counties and a higher collective loss of resilience in its infrastructure.

For comparisons between the two groups of counties with different levels of social vulnerability, the Gini coefficients are higher within the more vulnerable groups in all three disasters. However, the percentages of disaster vulnerable counties and the collective loss of resilience of the infrastructure serving these groups of counties vary in different contexts of disasters.

# 6.4 Results Discussion

### 6.4.1 Analysis of disaster inequality across different communities

As per the results presented in the above section, in the context of all three selected hurricanes, disaster inequality is at a similar level between the infrastructure serving the inland counties and the one serving the coastal counties. The results also show that disaster inequality is more severe within the socially vulnerable counties, which may imply that the disaster inequality is associated with social vulnerability.

The difference in disaster inequality between the groups of counties with varying levels of social vulnerability may be due to differences in (1) the quality and adequacy of infrastructure services and (2) disaster relief and aid for recovery. First, socially vulnerable communities, in general, lack adequate and stable infrastructure services. Studies show that socially vulnerable communities with higher percentages of minorities and disabled, poor, or unemployed populations have inadequate and substandard infrastructure, unmet infrastructure needs, and may lack even basic infrastructure services such as stable water supply, uninterrupted electricity, and safe and durable sanitation, etc. (SAMSHA 2017, Constible 2018). During disasters, such unstable infrastructure is more likely to experience varying levels of physical damage and/or service disruption. As a result, there is a higher possibly that these communities may experience varying levels of disaster impacts to their infrastructure, thus suffering from more severe disaster inequality. For example, in the aftermath of Hurricane Michael, Bay County experienced significant damage to public infrastructure (e.g., transportation), and a preliminary damage assessment showed that Bay County spent more fund for debris removal compared to other disaster impacted counties (Moline 2019). Second, socially vulnerable communities may lack access to disaster relief and assistance, which results in many uncertainties in their recovery. Previous studies show that socially vulnerable communities, following a disaster, face barriers in interacting with bureaucratic systems to receive disaster assistance (SAMSHA 2017). Socially vulnerable communities typically receive less reinvestment or recovery aid compared to those communities with higher socioeconomic statuses, thus they generally spend longer time in recovery (Nexus 2017). For example, in Hurricane Irma, the regions with higher percentages of minorities (Latinos and Hispanic populations) and disabled populations experienced longer electric power outages in the wake of Hurricane Irma (Mitsova et al. 2018).

# 6.4.2 Analysis of disaster vulnerability across different communities

As per above presentation of the results, we found that there are higher percentages of disaster vulnerable counties in the rural group as compared to those in the urban group in the contexts of

Hurricanes Michael and Irma. This may be due to lack of infrastructure investment funds, delay in disaster aid and relief, aging and substandard infrastructure, and socioeconomic statuses of individuals living in the rural areas. First, urban areas, comparing to rural areas, are more likely to receive federal assistance and funds for developing new infrastructure and maintaining existing ones (Todoroff 2022), and they are more likely to implement resilience strategies in new infrastructure planning and development (Venema 2017) due to adequacy in funds. In contrast, rural areas, with limited funding and support, may have limited capabilities in implementing disaster mitigation practices or resilience strategies (Kapucu et al. 2013). Second, rural areas are generally left behind in receiving disaster recovery assistance, aid, and resources (Todoroff 2022). For example, in the aftermath of Hurricane Irma, comparing to urban counties, rural counties (e.g., Taylor County) of Florida were delayed in receiving recovery resources and aid (Stofan 2017). Third, rural communities typically have aged, obsolete and substandard infrastructure that requires significant repair and maintenance. Unlike urban areas, rural areas lack diverse economic resources (e.g., tax base from large corporations and businesses) and may experience difficulty in attracting developers (Browne 2022). The infrastructure needs of rural areas are not adequately addressed as the infrastructure planning and initiatives have mostly focused on the urban areas. As a result, rural communities lack access to reliable essential infrastructure services (NCSL 2020). Fourth, the residents in the rural areas generally have lower socioeconomic statuses, such as higher level of poverty and unemployment rates, and fewer educational opportunities (Lowe 2017, The Conversation 2017). For example, according to the United States Department of Agriculture (USDA) Economic Research Service (USDA-ERS 2022), there is a higher poverty rate of rural communities (18.8%) in Florida compared to that of the urban communities (12.6%). (USDA-ERS 2022). In addition, variations in the severity of disaster impacts to infrastructure are found to be related to the socioeconomic characteristics of the communities. For example, poor communities

and minority communities experienced severe impacts from Hurricane Michael and are deprived of necessary resources during the recovery period (Moens 2022).

### 6.4.3 Analysis of collective resilience of infrastructure serving different communities

As per above presentation of the results, in the context of Hurricanes Michael and Irma, the infrastructure serving the inland counties has relatively worse overall resilience performance compared to the one serving the coastal counties. However, in the context of Hurricane Sally, the electric power infrastructure serving both coastal and inland counties has similar resilience performance. Such differences may be attributed to the characteristics of the hurricanes. First, Hurricanes Irma and Michael were both Category 5 hurricanes. Such hurricanes with high intensities have more potential to travel up to 100 to 200 miles inland after their landfall and they could bring extreme rainfall and wind forces resulting in extreme inland flooding (Raizner 2022). For example, after making landfall in Mexico Beach, Hurricane Michael remained at Category 3 strength while travelling through seven inland counties of Florida, such as Calhoun, Liberty and Gadsden Counties (NOAA 2019). It kept brining extreme rainfall that resulted in flash flooding in the inland counties. The combination of strong winds and heavy rainfall caused significant damage to roads and highways, interrupted the services of wastewater and communication infrastructure.

In contrast, as a Category 2 hurricane, Hurricane Sally caused extensive damage and destruction to infrastructure mostly across the coastal regions (NOAA 2021). The storm surges along with rainfall brought by the Category 2 hurricane caused widespread flash and river flooding on coastal counties (e.g., Pensacola and Escambia Counties) resulting in significant damage to electric power lines (NOAA 2021). Hurricane Sally started to lose its strength after landfall and travelled toward inland regions of Florida as a tropical storm (NOAA 2021). Thus, the electric power infrastructure serving inland counties experienced less severe impacts from the hurricane as compared to the one serving the coastal counties.

## 6.5 Conclusions

This chapter presents four case studies that analyze infrastructure resilience while accounting for disaster inequalities and vulnerabilities. By utilizing the social welfare-based infrastructure resilience assessment (SW-Infra-RA) framework proposed by Dhakal and Zhang (2022), the study analyzed (1) the levels of disaster inequality, (2) the levels of disaster vulnerability, and (3) the collective resilience of infrastructure serving counties with different characteristics (e.g., spatial, demographic, and socioeconomic statuses). These results indicate that the infrastructure serving the more vulnerable group of counties experience higher levels of disaster inequality. In the context of hurricanes with high intensity (Hurricanes Michael and Irma), there is a higher percentage of disaster vulnerable counties in the rural group. Also, the infrastructure of the inland group of counties, collectively, has poorer resilience performance compared to that of the coastal group of counties.

From practical perspectives, the results generated through the equity-incorporated resilience analysis could facilitate decision makers to better understand the level of inequality in disaster impacts across different communities and identify the communities that are highly vulnerable in a disaster. The analysis results could help decision makers and infrastructure planners better understand the links between collective resilience of infrastructure and the communities with various characteristics (e.g., spatial, demographic, and socioeconomic statuses). This study has the potential to promote equitable planning by allowing decision makers to prioritize infrastructure resilience initiatives and investment and disaster relief resources to those communities that are in dire need.

# **CHAPTER 7 CONCLUSIONS**

### 7.1 Summary

This dissertation aims to facilitate equitable infrastructure resilience planning by providing a new infrastructure resilience assessment framework that accounts for disaster inequality and vulnerability. The dissertation presents four chapters to achieve research objectives that includes (1) understanding the interrelationships between infrastructure resilience and social equity in the context of a disaster, (2) developing a new infrastructure resilience assessment framework that incorporates disaster inequality and vulnerability, (3) developing a prototype decision support system that facilitates automatic infrastructure resilience assessment, and (4) implementing the model by conducting resilience analyses of different infrastructure systems in the context of various hurricane disasters.

The first task of this dissertation focused on understanding infrastructure resilience, social equity, and their interrelationships in the context of Hurricane Michael. To perform this task, Twitter activities generated by the disaster impacted counties were utilized to examine the social equity conditions and infrastructure resilience conditions. Statistical correlation analyses were conducted across Twitter activities, social equity conditions, and infrastructure resilience conditions in the context of Hurricane Michael. In general, the results indicate that counties with different social equity conditions experienced different level of impacts from disasters. In other words, social equity factors could potentially impact the infrastructure resilience conditions.

The second task of this dissertation focused on developing a social-welfare-based infrastructure resilience assessment (SW-Infra-RA) framework while accounting for disaster inequalities and vulnerabilities. The proposed assessment framework is theoretically based on the social welfare theory and social welfare functions. This model utilized Gini coefficient to measure the unequal distributions of disaster impacts on infrastructure serving various communities. It proposed the

"line of vulnerability" to model disaster vulnerabilities. Two sets of case studies, a hypothetical and a real case study, were conducted to illustrate the application of the SW-Infra-RA model. This proposed model could potentially support the development of resilient infrastructure and facilitate equitable resilience planning.

The third task of this dissertation focused on developing a prototype of a decision support system that could potentially facilitate automatic infrastructure resilience assessment. This prototype system allows decision makers to measure infrastructure resilience while accounting for disaster inequalities and vulnerabilities. The input of the system is disaster impacts (e.g., functional loss, recovery time) data of infrastructure, and the output is the collective resilience of infrastructure serving multiple communities.

The fourth task of this dissertation focused on implementing the proposed infrastructure resilience assessment model (SW-Infra-RA) in the context of real disasters. This task analyzed the levels of disaster inequality, levels of disaster vulnerability, and resilience performance of infrastructure serving counties with different characteristics (e.g., spatial, demographic, and socioeconomic). The resilience of four different infrastructure (e.g., communication service, electric power, transportation, and wastewater infrastructure) in the context of three hurricanes (e.g., Hurricane Michael, Hurricane Irma, and Hurricane Sally) was analyzed.

## 7.2 Research Contributions

This dissertation offers contributions to the body of knowledge in advancing the understanding and methods of assessing infrastructure resilience; these contributions include:

1. This research advances the understanding of the interrelationships between infrastructure resilience and social equity in the context of a hurricane disaster. It shows how the communities with different social characteristics may experience disproportionate impacts from Hurricane Michael due to varying levels of infrastructure damage or time for recovery. This knowledge is

critical as it could support the (re)development and (re)investment of infrastructure in a way that not only addresses disaster resilience challenges but also facilitates social equity in the impacted communities.

2. This research utilized a data-based approach to derive useful meaning. Social media data analysis, which allows for easy collection of timely data, could potentially allow practitioners and decision makers to analyze how disasters could impact people and infrastructure in a more efficient and timely manner.

3. This research developed a new infrastructure resilience assessment framework that measures the collective resilience of infrastructure serving multiple communities while accounting for disaster inequality and vulnerability. It adapted methods from the social science and economics domains to mathematically measure the unequal distributions of disaster impacts across various communities and proposed new ways of evaluating the severe impacts on vulnerable communities.

4. The proposed model (SW-Infra-RA) provides a theoretical basis for equity-incorporated decision making by allowing decision makers to quantitatively assess infrastructure resilience while accounting for inequality and vulnerability. The results generated using this model can be utilized by decision makers to better understand the inequalities during disasters and to identify the communities that are more vulnerable in these disasters.

### 7.3 Limitations and Recommendations for Future Research

Based on the research development in this study, some research directions can be further investigated to improve the performance of the proposed assessment framework and its applications. The recommendations for future research are presented as follows.

First, this research study explored the interrelationships between infrastructure resilience and social equity and proposed a new social-welfare-based infrastructure resilience assessment

framework that accounts for disaster inequality and disaster vulnerability. Although the SW-infra model focuses on assessing collective resilience of infrastructure through aggregating the disaster impacts on the infrastructure of each individual community, it currently doesn't account for the interdependencies of infrastructure serving these communities. Other methods such as system of systems approach or system network analysis can be used to measure such interdependencies and can be further integrated to the proposed assessment framework.

Second, the findings about the interrelationships between social equity and infrastructure resilience cannot be generalize to the general population. The findings are based on data collected from Twitter, which may not fully represent the opinions of the general population. For example, previous research shows that, comparing to the older population, younger adults aged 15-34 years old are more represented in social media (e.g., Twitter). In general, vulnerable populations (e.g., minorities, disabled, low-income, elderly) are more likely to be unaware of communication tools and use of social media, and they may be less representative on social media (e.g., Twitter). Therefore, further investigations on the communication patterns of vulnerable populations are needed to analyze the behaviors of vulnerable populations in disasters.

Third, the proposed model (SW-Infra-RA) is designed to assess the collective resilience of a single type of infrastructure. Assessing the resilience of multiple types of infrastructure may be conducted, depending on the input of the data. For example, if the collected data on disaster impacts are for multiple types of infrastructure, it may be possible to derive the collective resilience of multiple infrastructures.

Fourth, the SW-Infra-RA model was implemented in analyzing the resilience of different types of infrastructure serving communities within Florida. In future studies, there is a need to further apply the model in analyzing the resilience of different types of infrastructure serving a larger number of communities in other parts of the country. Such application would offer more insight and

understanding of how inequalities in and vulnerabilities to disaster impacts could impact infrastructure resilience.

Fifth, the resilience analysis of infrastructure is conducted on a county level, which may not reveal the actual infrastructure resilience and social equity conditions within different communities in the same county. This can be further addressed by collecting the disaster impacts data based on zip codes or different types of communities by grouping several zip codes together.

Sixth, although multiple sets of case studies have been conducted to implement the proposed model in various hurricane disasters, the model requires further validation. Validation of the proposed model is a challenging task as there is no agreed upon "gold standard" for the results generated by the proposed model to be compared with. Previous studies (e.g., Ouyang and Wang 2015, Singhal et al. 2020, Argyroudis et al. 2021) show that infrastructure resilience assessment models or methods may lack validation approaches. Some research (e.g., Feldmeyer et al. 2020, Anderson et al. 2020, Cai et al. 2016) suggests the use of an empirical approach. In this research, expertinvolved experiments can be conducted to validate the proposed model. For example, a set of case studies that involve domain experts can be conducted to validate the proposed model. The experts may include people from government, non-government and private sectors that hold expertise in the domain of disaster resilience and social equity. First, the data of a disaster, disaster impacts, and infrastructure in the impacted communities will be provided to the experts. Second, the experts will be instructed to focus on inequality and vulnerability in the disasters, and they will be asked to evaluate how disaster impacts are distributed across different communities and how infrastructure of vulnerable communities are impacted during the disaster. Third, through open discussion, the experts will be asked to rank the communities based on the collective resilience of their infrastructure. Fourth, the collective resilience of infrastructure in different communities will be

measured using the SW-Infra-RA model, and a ranking of these communities will be provided. The correlation between the expert-based ranking and the model-based ranking will be evaluated.

Seventh, the proposed prototype system has not been used by the general public or decision makers. To address this limitation, community outreach programs can be implemented in the future. For example, seminars and workshops can be held at FIU to engage interested stakeholders and community collaborators. In these seminars and workshops, this research work including the findings and results will be presented. The participants will then have the opportunity to test use the prototype system. Additionally, seminars and information sessions regarding this research can be held with local resilience-focused organizations, such as 100 Resilient Cities, and the County's Office of Resilience. These meetings and workshops could potentially provide effective venues for reaching out to different sectors as they bring together representatives from public, private, and non-profit sectors in local communities.

### LIST OF REFERENCES

- Adler, M. D. 2017. "A better calculus for regulators: From cost-benefit analysis to the social welfare function." Duke Law School Public Law & Legal Theory Series. http://dx.doi.org/10.2139/ssrn.2923829
- Afonso, H., LaFleur, M., and Alarcón, D., 2015 "Inequality Measurement" Developmental Issue No. https://www.un.org/en/development/desa/policy/wess/wess\_dev\_issues/dsp\_policy\_02.pd f
- Ajao, O., Hong, J., and Liu, W. 2015. "A survey of location inference techniques on Twitter." Journal of Information Science, 41(6), 855– 864. https://doi.org/10.1177/0165551515602847
- AJMC (The American Journal of Managed Care) 2006. "Vulnerable Populations: Who Are They?" Last accessed on Jan 30, 2021. <a href="https://cdn.sanity.io/files/0vv8moc6/ajmc/7156d97747bada738caf6bb3e702169373491c">https://cdn.sanity.io/files/0vv8moc6/ajmc/7156d97747bada738caf6bb3e702169373491c</a> c8.pdf>
- Allen, G., 2018. "After Hurricane Michael, A Call for Stricter Building Codes in Florida's<br/>Panhandle"Last<br/>accessedFeb04,2020.<a href="https://www.weather.gov/tae/HurricaneMichael2018">https://www.weather.gov/tae/HurricaneMichael2018</a>>.
- Anderson, K., Hotchkiss, E., Myers, L., Stout, S., Grue, N., Gilroy, N., Aldred, J.R. and Rits, M., 2020. "After the hurricane: Validating a resilience assessment methodology." *International Journal of Disaster Risk Reduction*, 51, 101781.
- APA (American Planning Association) 2021. "Social Equity" Last accessed on Jan 25, 2021. < https://www.planning.org/knowledgebase/equity/>.
- Argyroudis, S. A., Achillopoulou, D. V., Livina, V., and Mitoulis, S. A. 2021. "Data-driven resilience assessment for transport infrastructure exposed to multiple hazards." In *Bridge Maintenance, Safety, Management, Life-Cycle Sustainability and Innovations*, 3267-3274, CRC Press.
- Argyroudis, S. A., Nasiopoulos, G., Mantadakis, N., and Mitoulis, S. A. 2020. "Cost-based resilience assessment of bridges subjected to earthquakes." *International Journal of Disaster Resilient Built Environment*.
- Ariyabandu, M. M., 2009. "Sex, gender and gender relations in disasters." Women, gender and disaster: Global issues and initiatives, 5-17.
- Arrow, K. J. 1963. "Social Choice and Individual Values." Vol 12, Yale University Press, New Haven, CT.
- ARUP 2021. "REDi Rating System" Last accessed June 20, 2021. <a href="https://www.arup.com/perspectives/publications/research/section/redi-rating-system">https://www.arup.com/perspectives/publications/research/section/redi-rating-system</a>

- Ashraf, M. A., & Azad, M. A. K. 2015. "Gender issues in disaster: Understanding the relationships of vulnerability, preparedness and capacity." *Environment and ecology research*, 3(5), 136-142. DOI: 10.13189/eer.2015.030504
- Assad, A., Moselhi, O., and Zayed, T. 2019. "A new metric for assessing resilience of water distribution networks." *Water*, 11(8), 1701.
- Atkinson, A., and Brandolini, A. 2010. "On Analyzing the World Distribution of Income." *World Bank Economic Review*, 24(1), 1-37.
- Aydin, N. Y., Duzgun, H. S., Heinimann, H. R., Wenzel, F., and Gnyawali, K. R. 2018. "Framework for improving the resilience and recovery of transportation networks under geohazard risks." International journal of disaster risk reduction, 31, 832-843.
- Badré, B., 2015. "Why infrastructure investment is key to ending poverty" Last accessed May 22, 2020. <a href="https://www.weforum.org/agenda/2015/10/why-infrastructure-investment-is-key-to-ending-poverty/">https://www.weforum.org/agenda/2015/10/why-infrastructure-investment-is-key-to-ending-poverty/</a>>.
- Balaei, B., Wilkinson, S., Potangaroa, R., Hassani, N., and Alavi-Shoshtari, M. 2018. "Developing a framework for measuring water supply resilience." *Natural Hazards Review*, 19(4), 04018013.
- Beigi G., Hu X., Maciejewski R., Liu H. 2016. "An Overview of Sentiment Analysis in Social Media and Its Applications in Disaster Relief." In Sentiment Analysis and Ontology Engineering. Studies in Computational Intelligence, pp 313-340, vol 639. Springer, Cham
- Berkeley, A. R., Wallace, M., and Coo, C. 2010. "A framework for establishing critical infrastructure resilience goals." *Final Report and Recommendations by the Council, National Infrastructure Advisory Council*, 18-21.
- Boscarino, J. A., Hoffman, S. N., Adams, R. E., Figley, C. R., and Solhkhah, R. 2014. "Mental health outcomes among vulnerable residents after Hurricane Sandy: Implications for disaster research and planning." American Journal of Disaster Medicine, 9(2), 97-106.
- Braese, J., Rentschler, J., and Hallegatte, S. 2019. "Resilient infrastructure for thriving firms: A review of the evidence." *Policy Research Working Paper*, 8895, *Global Facility for Disaster Reduction and Recovery, World Bank Group*.
- Browne, K. 2022. "Isolated and costly: The unique infrastructure challenges facing rural communities" Last accessed July 10, 2022 < https://www.westerncity.com/article/isolated-and-costly-unique-infrastructure-challenges-facing-rural-communities>
- Bruneau, M., Chang, S. E., Eguchi, R. T., Lee, G. C., O'Rourke, T. D., Reinhorn, A. M., Shinozuka, M., Tierney, K., Wallace, W.A. and Von Winterfeldt, D. 2003. "A framework to quantitatively assess and enhance the seismic resilience of communities." *Earthquake spectra*, 19(4), 733-752.
- Bruneau, M., and Reinhorn, A. 2006. "Overview of the resilience concept." In Proceedings of the 8th US national conference on earthquake engineering, 2040, pp. 18-22.

- Bullard, R. D. 2007. "Equity, unnatural man-made disasters, and race: Why environmental justice matters." In Equity and the Environment. Emerald Group Publishing Limited, ISBN: 978-0-7623-1417-1
- Burgess, B. 2018. "Hurricane Michael Sends Up Red Flags for The Future of Emergency Responder Communications in Florida" Last accessed Sep 12, 2022 <a href="https://thecapitolist.com/hurricane-michael-sends-up-red-flags-for-the-future-of-emergency-responder-communications-in-florida/">https://thecapitolist.com/hurricane-michael-sends-up-red-flags-for-the-future-of-emergency-responder-communications-in-florida/</a>
- Cai, H., Lam, N. S. N., Zou, L., Qiang, Y., and Li, K. 2016. "Assessing community resilience to coastal hazards in the Lower Mississippi River Basin." *Water*, 8(2), 46.
- Callan, T., and Nolan, B. 1991. "Concepts of poverty and the poverty line." J. Econ. Surv., 5(3), 243-261.
- Caragea, C., Squicciarini, A. C., Stehle, S., Neppalli, K., and Tapia, A. H. 2014. "Mapping moods: Geo-mapped sentiment analysis during hurricane sandy." In *Proceedings of the 11th International ISCRAM Conference- University Park*, Pennsylvania, USA, May.
- Cardoni, A., Cimellaro, G. P., Domaneschi, M., Sordo, S., and Mazza, A. 2020. "Modeling the interdependency between buildings and the electrical distribution system for seismic resilience assessment." *International Journal of Disaster Risk Reduction*, 42, 101315.
- Castillo, E. G., Ijadi-Maghsoodi, R., Shadravan, S., Moore, E., Mensah, M. O., Docherty, M., Nunez, M.G.A., Barcelo, N., Goodsmith, N., Halpin, L.E., and Morton, I. 2019. "Community interventions to promote mental health and social equity." *Current psychiatry reports*, 21(5), 35. https://doi.org/10.1007/s11920-019-1017-0.
- CCSF (City and County of San Francisco) 2021. "Hazus: A Tool for Modeling Damages and Economic Losses from Natural Disasters." *Office of Economic Analysis, CCSF*. https://sfcontroller.org/sites/default/files/FileCenter/Documents/6356-HAZUS%20Presentation.pdf
- CDC/ATSDR (Centers for Disease Control and Prevention/ Agency for Toxic Substances and Disease Registry) 2022. "CDC/ATSDR Social Vulnerability Index" Last accessed May 20, 2022. < <u>https://www.atsdr.cdc.gov/placeandhealth/svi/index.html></u>
- Cervone, G., Schnebele, E., Waters, N., Moccaldi, M., and Sicignano, R. 2017. "Using social media and satellite data for damage assessment in urban areas during emergencies." In Seeing Cities Through Big Data (pp. 443-457). Springer, Cham.
- Cimellaro, G. P., Reinhorn, A. M., and Bruneau, M. 2010. "Seismic resilience of a hospital system." *Structure and Infrastructure Engineering*, 6(1-2), 127-144.
- Chan, R., and Schofer, J. L. 2016. "Measuring transportation system resilience: Response of rail transit to weather disruptions." *Natural Hazards Review*, *17*(1), 05015004. https://doi.org/10.1061/(ASCE)NH.1527-6996.0000200
- Chandramouleeswaran, K. R., and Tran, H. T. 2018. "Data-driven resilience quantification of the US Air transportation network." In 2018 Annual IEEE International Systems Conference, 1-7, IEEE

- Chang, S. E., McDaniels, T., Fox, J., Dhariwal, R., and Longstaff, H. (2014). Toward disasterresilient cities: Characterizing resilience of infrastructure systems with expert judgments. *Risk analysis*, *34*(3), 416-434
- Chen, Y., Wang, Q., and Ji, W. 2020. "Rapid Assessment of Disaster Impacts on Highways Using Social Media." *Journal of Management in Engineering*, *36*(5), 04020068.
- Chopra, S. S., Dillon, T., Bilec, M. M., and Khanna, V. 2016. "A network-based framework for assessing infrastructure resilience: a case study of the London metro system." Journal of The Royal Society Interface, 13(118), 20160113.
- CivicPlus 2020. "The Importance of Social Media in Crisis Communications" Last accessed Dec 20, 2020. < https://www.civicplus.com/civicready/crisis-communications-guide >.
- Cobo, A., Parra, D., and Navón, J. 2015. "Identifying relevant messages in a twitter-based citizen channel for natural disaster situations." In *Proceedings of the 24th International Conference* on World Wide Web (pp. 1189-1194).
- Cohen, J., Cohen, P., West, S. G., and Aiken, L. S. 2013. "Applied multiple regression/correlation analysis for the behavioral sciences." *Routledge. 3rd edition, Lawrence Erlbaum Associates, Inc.*
- Coleman, N., Esmalian, A., and Mostafavi, A. 2020. "Equitable resilience in infrastructure systems: empirical assessment of disparities in hardship experiences of vulnerable populations during service disruptions." *Natural Hazards Review*, 21(4).
- Comes, T., Meesters, K., and Torjesen, S. 2019. "Making sense of crises: the implications of information asymmetries for resilience and social justice in disaster-ridden communities." Sustainable and Resilient Infrastructure, 4(3), 124-136.
- Conceição, P., and Ferreira, P. 2000. "The young person's guide to the Theil index: Suggesting intuitive interpretations and exploring analytical applications." *UTIP Working Paper No.* 14. http://dx.doi.org/10.2139/ssrn.228703
- Constible, J., 2018. "We Need to Keep Our Elders Safe from Climate Disasters". Last accessed on Sep 02, 2020. <a href="https://www.nrdc.org/experts/juanita-constible/we-need-keep-our-elders-safe-climate-disasters">https://www.nrdc.org/experts/juanita-constible/we-need-keep-our-elders-safe-climate-disasters</a>
- Croope, S. V., and McNeil, S. 2011. "Improving resilience of critical infrastructure systems postdisaster: recovery and mitigation." *Transportation research record*, 2234(1), 3-13. https://doi.org/10.3141/2234-01.
- Cutter, S. L., Boruff, B. J., and Shirley, W. L. 2003. "Social vulnerability to environmental hazards." Social science quarterly, 84(2), 242-261.
- Cutter, S. L., Burton, C. G., and Emrich, C. T. 2010. "Disaster resilience variables for benchmarking baseline conditions." Journal of Homeland Security and Emergency Management, 7(1), pp 5-7
- Deardorff, A.V., 2016. "Deardorffs' Glossary of International Economics" Last accessed September 20, 2021. < http://www-personal.umich.edu/~alandear/glossary/w.html>

- Decò, A., Bocchini, P., and Frangopol, D. M. 2013. "A probabilistic approach for the prediction of seismic resilience of bridges." *Earthquake Engineering & Structural Dynamics*, 42(10), 1469-1487.
- Dhakal, S., and Zhang, L., and Lv, X. 2021. "Understanding infrastructure resilience, social equity, and their interrelationships: Exploratory study using social media data in Hurricane Michael." *Natural Hazards Review, ASCE, 22*(4): 04021045.
- Dhakal, S., and Zhang, L. 2022. "A Social Welfare based Infrastructure Resilience Assessment Framework: Toward Equitable Resilience for Infrastructure Development", *Natural Hazards Review*, 24(1), 04022043.
- Diao, K., Sweetapple, C., Farmani, R., Fu, G., Ward, S., and Butler, D. 2016. "Global resilience analysis of water distribution systems." *Water research*, 106, 383-393.https://doi.org/10.1016/j.watres.2016.10.011
- DirectRelief. 2018. "Hurricane Michael strikes some of the nation's most vulnerable communities." Last accessed Feb 19, 2019. <a href="https://www.directrelief.org/2018/10/hurricane-michael-strikes-some-of-the-nations-most-vulnerable-communities/">https://www.directrelief.org/2018/10/hurricane-michael-strikes-some-of-the-nations-most-vulnerable-communities/</a>>.
- Dolan, P and Robinson, A. 2001. "The measurement of preferences over the distribution of benefits: the importance of reference points." *European Economic Review*, 45,1697–709.
- Domingue, S. J., and Emrich, C. T. 2019. "Social Vulnerability and Procedural Equity: Exploring the Distribution of Disaster Aid Across Counties in the United States." The American Review of Public Administration, 49(8), 897-913.
- Doorn, N. 2019. "How can resilient infrastructures contribute to social justice? Preface to the special issue of sustainable and resilient infrastructure on resilience infrastructures and social justice." *Sustainable and Resilient Infrastructure*, 4(3), 99-102
- Drakes, O., Tate, E., Rainey, J., and Brody, S. 2021. "Beyond Damages: Social Equity in Allocating Disaster Assistance." Last accessed Jun 26, 2022. https://hazards.colorado.edu/news/research-counts/beyond-damages-social-equity-inallocating-disaster-assistance
- Emrich, C. T., Tate, E., Larson, S. E., and Zhou, Y. 2019. "Measuring social equity in flood recovery funding." Environmental Hazards, pp.1-23.
- Espinoza, S., Panteli, M., Mancarella, P., and Rudnick, H. 2016. "Multi-phase assessment and adaptation of power systems resilience to natural hazards." *Electric Power Systems Research*, 136, 352-361.
- Fatemi, F., Ardalan, A., Aguirre, B., Mansouri, N., and Mohammadfam, I. 2017. "Social vulnerability indicators in disasters: Findings from a systematic review." International journal of disaster risk reduction, 22, 219-227.
- FCC (Federal Communications Commission) 2018. "Communication Status Report for Areas Impacted by Hurricane Michael" Last accessed Aug 10, 2019. <a href="https://docs.fcc.gov/public/attachments/DOC-354510A1.pdf">https://docs.fcc.gov/public/attachments/DOC-354510A1.pdf</a>.

- FDEM (Florida Division of Emergency Management). 2019. "Michael Recovery." Last accessed Aug 20, 2019. <a href="https://www.floridadisaster.org/info/">https://www.floridadisaster.org/info/</a>>
- FDEO (Florida Department of Economic Opportunity). 2019. "State of Florida Action Plan for Disaster Recovery". Last accessed Sep 05, 2020. <<u>https://floridajobs.org/docs/default-source/office-of-disaster-recovery/hurricane-michael/state-action-plan-for-disaster-recovery--hurricane-michael\_approved-by-hud.pdf?sfvrsn=956f46b0\_2.></u>
- FDEP (Florida Department of Environmental Protection) 2022. "Public Notice of Pollution" Last accessed March 20, 2022. < https://floridadep.gov/pollutionnotice>
- FDOT (Florida Department of Transportation). 2019. "Hurricane Michael Recovery". Last accessed Aug 04, 2019. <a href="https://www.fhwa.dot.gov/fldiv/michael.cfm">https://www.fhwa.dot.gov/fldiv/michael.cfm</a>>.
- Feldmeyer, D., Wilden, D., Jamshed, A., and Birkmann, J. 2020. "Regional climate resilience index: A novel multimethod comparative approach for indicator development, empirical validation and implementation." *Ecological indicators*, *119*, 106861.
- FEMA (Federal Emergency Management Agency) 2017. "Florida Hurricane Irma" Last accessed April 05, 2022. < https://www.fema.gov/disaster/4337>.
- FEMA (Federal Emergency Management Agency). 2018. "Florida Hurricane Michael (DR-4399)." Last accessed May 22, 2020. < <u>https://www.fema.gov/disaster/4399</u>>.
- FEMA (Federal Emergency Management Agency). 2019. "Hurricane Michael." Last accessed July 22, 2019. < https://www.fema.gov/hurricane-michael/>.
- FEMA (Federal Emergency Management Agency) 2021. "FEMA Preliminary Damage Assessment<br/>Guide"Guide"USDepartmentofHomelandSecurity.https://www.fema.gov/sites/default/files/documents/fema\_2021-pda-guide.pdf
- Fisher, R., and Norman, M. 2010. "Developing measurement indices to enhance protection and resilience of critical infrastructure and key resources." J. Bus, Conti. Emer. Plan., 4(3), 191-206.
- Flanagan, B. E., Gregory, E. W., Hallisey, E. J., Heitgerd, J. L., and Lewis, B. 2011. "A social vulnerability index for disaster management." J. Homel. Secur. Emerg. Manag., 8(1).https://doi.org/10.2202/1547-7355.1792
- Florida Health 2022. "Florida's Rural Counties" Last accessed May 12, 2022. < https://www.floridahealth.gov/programs-and-services/community-health/rural-health/\_documents/ruralcountiespdf.12.pdf>.
- FOIR (Florida Office of Insurance Regulation). 2019. "Hurricane Michael Claims Data." Last<br/>accessed on Sep 10, 2019.<br/><https://www.floir.com/Office/HurricaneSeason/HurricaneMichaelClaimsData.aspx>.
- Forsey, C., 2019. "What Is Twitter and How Does It Work?" Last accessed Feb 02, 2020, < https://blog.hubspot.com/marketing/what-is-twitter>.
- Fothergill, A., and Peek, L. A. 2004. "Poverty and disasters in the United States: A review of recent sociological findings." Natural hazards, 32(1), 89-110.

- FPSC (Florida Public Service Commission). 2021 "Hurricane season power outage report" Accessed July 17, 2021. http://www.psc.state.fl.us/Home/HurricaneReport.
- Frigerio, I., Zanini, F., Mattavelli, M., and De Amicis, M. 2019. "Understanding the interacting factors that influence social vulnerability: a case study of the 2016 central Italy earthquake." Disasters, 43(4), 867-890.
- Frost, J. 2020 "Understanding Significance Levels in Statistics" Last accessed on Sep 10. < https://statisticsbyjim.com/hypothesis-testing/significance-levels/>.
- Füssel, H. M. 2006. "Social welfare functions in global climate-economy models: Methodological Inconsistencies and their Policy Implications." Available at SSRN 900023, http://dx.doi.org/10.2139/ssrn.900023
- Galea, S., Nandi, A., and Vlahov, D. 2005. "The epidemiology of post-traumatic stress disorder after disasters." *Epidemiologic reviews*, 27(1), 78-91.
- Gardner, S., 2019 "Sewage spill correction plan complete, d by city" Last accessed Sep 10, 2022, <a href="https://www.staugustine.com/story/news/local/2019/03/22/city-finishes-state-requirements-after-hurricane-irma-sewage-spills-but-more-work-planned/5593427007/>">https://www.staugustine.com/story/news/local/2019/03/22/city-finishes-state-requirements-after-hurricane-irma-sewage-spills-but-more-work-planned/5593427007/>">https://www.staugustine.com/story/news/local/2019/03/22/city-finishes-state-requirements-after-hurricane-irma-sewage-spills-but-more-work-planned/5593427007/>">https://www.staugustine.com/story/news/local/2019/03/22/city-finishes-state-requirements-after-hurricane-irma-sewage-spills-but-more-work-planned/5593427007/>">https://www.staugustine.com/story/news/local/2019/03/22/city-finishes-state-requirements-after-hurricane-irma-sewage-spills-but-more-work-planned/5593427007/>">https://www.staugustine.com/story/news/local/2019/03/22/city-finishes-state-requirements-after-hurricane-irma-sewage-spills-but-more-work-planned/5593427007/>">https://www.staugustine.com/story/news/local/2019/03/22/city-finishes-state-requirements-after-hurricane-irma-sewage-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-but-more-work-planned/state-spills-spills-spills-spills-spills-spills-spills-spills-sp
- Gaus, W., Mayer, B., and Muche, R. 2015. "Interpretation of Statistical Significance—Exploratory Versus Confirmative Testing in Clinical Trials, Epidemiological Studies, Meta-Analyses and Toxicological Screening (Using Ginkgo biloba as an Example)". *Clin Exp Pharmacol*, 5(4).
- GBCI (Green Building Certification Inc.) 2021. "Resilient design for a changing world" Last accessed June 14, 2021. <a href="https://gbci.org/reli>">https://gbci.org/reli></a>
- Gil-Rivas, V., and Kilmer, R. P. 2016. "Building community capacity and fostering disaster resilience." *Journal of clinical psychology*, 72(12), 1318-1332.
- Goldgruber, E., Sackl-Sharif, S., Gutounig, R., and Ausserhofer, J. 2017. "Social media as a crisis communication arena: Digging into new communication spaces." In ECSM 2017. Proceedings of the 4th European conference on social media research (p. 115À123).
- Gooden, S., Jones, D., Martin, K. J., and Boyd, M. 2009. "Social equity in local emergency management planning." *State and Local Government Review*, *41*(1), 1-12.
- Hallegatte, S., Rentschler, J., Rozenberg, J., 2019. "Lifelines: The Resilient Infrastructure Opportunity." *Sustainable Infrastructure, Washington, DC*, World Bank. <a href="https://openknowledge.worldbank.org/handle/10986/31805">https://openknowledge.worldbank.org/handle/10986/31805</a>
- Hamstead, Z. A., Fisher, D., Ilieva, R. T., Wood, S. A., McPhearson, T., and Kremer, P. 2018. "Geolocated social media as a rapid indicator of park visitation and equitable park access." *Computers, Environment and Urban Systems*, 72, 38-50. https://doi.org/10.1016/j.compenvurbsys.2018.01.007
- Hao, H., Baireddy, S., Bartusiak, E. R., Konz, L., LaTourette, K., Gribbons, M., Chan, M., Comer, M. L., and Delp, E. J. 2020. "An attention-based system for damage assessment using satellite imagery". arXiv preprint arXiv:2004.06643

- Harrison, S., and Johnson, P. 2019. "Challenges in the adoption of crisis crowdsourcing and social media in Canadian emergency management." *Government Information Quarterly*, 36(3), 501-509.
- Harsanyi, J. C. 1955. "Cardinal welfare, individualistic ethics, and interpersonal comparisons of utility." *Journal of Political Economy*, 63(4), 309-321.
- Heinimann, H. R., and Hatfield, K. 2017. "Infrastructure resilience assessment, management and governance-state and perspectives." In Resilience and risk (pp. 147-187). Springer, Dordrecht.
- Hirsch, J. A., Green, G. F., Peterson, M., Rodriguez, D. A., and Gordon-Larsen, P. 2016. "Neighborhood sociodemographics and change in built infrastructure." *Journal of Urbanism: International Research on Placemaking and Urban Sustainability*, 10(2), 181-197.
- Holling, C. S. 1973. "Resilience and stability of ecological systems." *Annual review of ecology and systematics*, 4(1), 1-23.
- Holling, C. S. 1996. "Engineering resilience versus ecological resilience." *Engineering within* ecological constraints, 31, 32.
- Hoover, E. M. 1941. "Interstate Redistribution of Population: 1850-1940." Journal of Economic History 1,199–205
- Hossain, N. U. I., Jaradat, R., Hosseini, S., Marufuzzaman, M., and Buchanan, R. K., 2019. "A framework for modeling and assessing system resilience using a Bayesian network: A case study of an interdependent electrical infrastructure system." *Int. J. Criti. Infra. Prot.*, 25, 62-83.
- Hosseini, S., and Barker, K. 2016. "Modeling infrastructure resilience using Bayesian networks: A case study of inland waterway ports." *Computers & Industrial Engineering*, 93, 252-266.
- Houston, J. B., Hawthorne, J., Perreault, M. F., Park, E. H., Goldstein Hode, M., Halliwell, M. R., Turner McGowen, S.E., Davis, R., Vaid, S., McElderry, J.A., and Griffith, S. A. 2015.
  "Social media and disasters: a functional framework for social media use in disaster planning, response, and research." *Disasters*, 39(1), 1-22.
- Huang, C., and Taylor, R., 2019. "Any Federal Infrastructure package should boost investment in Low-Income Communities", Last accessed June 10, 2021 <<u>https://www.cbpp.org/research/federal-budget/any-federal-infrastructure-package-should-boost-investment-in-low-income</u>>
- Hutto, C.J. and Gilbert, E.E. 2014. "VADER: A Parsimonious Rule-based Model for Sentiment Analysis of Social Media Text." *Eighth International Conference on Weblogs and Social Media* (ICWSM-14). Ann Arbor, MI, June 2014
- Internet Archive, 2020. "Archiveteam-twitter-stream-2018-10." Accessed Oct 20, 2019 < https://archive.org/details/archiveteam-twitter-stream-2018-10>.

- IWR (Institute of Water Resources). 2016. "Identification and Engagement of Socially Vulnerable Populations in the USACE Decision Making Process" Last accessed Jan 24, 2021 <.https://www.iwr.usace.army.mil/Portals/70/docs/iwrreports/Identifying\_and\_Engaging \_Socially\_Vulnerable\_Populations\_% 20IWRv2\_08\_01\_2016.pdf>
- Jacobson, A., Milman, A. D., and Kammen, D. M. 2005. "Letting the (energy) Gini out of the bottle: Lorenz curves of cumulative electricity consumption and Gini coefficients as metrics of energy distribution and equity." *Energy Policy*, 33(14), 1825-1832. https://doi.org/10.1016/j.enpol.2004.02.017
- Jagtenberg, C. J. 2017. "Efficiency and fairness in ambulance planning", *Doctoral Dissertation*, VU Amsterdam.
- Jian, J. I. N., Jianxiang, W. A. N. G., Xiaoyi, M. A., Yuding, W. A. N. G., and Renyong, L. I. 2015. "Equality of medical health resource allocation in China based on the Gini coefficient method." *Iranian Journal of Public Health*, 44(4), 445.
- Johnson, A.E., 2019. "Florida's Rising Tide: Income Inequality Effects by County." *Honors Undergraduate Theses*, 521, UCF Theses and Dissertations. https://stars.library.ucf.edu/honorstheses/521
- Kangior, M. 2013. "Engineering Resilience: The Resilience STARTM Home Pilot Project." Last accessed July 25, 2020. <a href="https://www.dhs.gov/blog/2013/11/18/engineering-resilienceresilience-star%E2%84%A2-home-pilot-project">https://www.dhs.gov/blog/2013/11/18/engineering-resilienceresilience-star%E2%84%A2-home-pilot-project</a>>
- Kapucu, N., Hawkins, C. V., and Rivera, F. I. 2013. "Disaster preparedness and resilience for rural communities." *Risk, Hazards and Crisis in Public Policy*, *4*(4), 215-233.
- Karamouz, M., Rasoulnia, E., Olyaei, M. A., and Zahmatkesh, Z. 2018. "Prioritizing investments in improving flood resilience and reliability of wastewater treatment infrastructure." Journal of Infrastructure Systems, 24(4), 04018021.
- Karamouz, M., Taheri, M., Khalili, P., and Chen, X. 2019. "Building infrastructure resilience in coastal flood risk management." *Journal of Water Resources Planning and Management*, 145(4), 04019004. https://doi.org/10.1061/(ASCE)WR.1943-5452.0001043
- Kent, J. D., and Capello, H. T. 2013. "Spatial patterns and demographic indicators of effective social media content during the horse thief canyon fire of 2012." Cartography and Geographic Information Science, 40(2), 78–89. https://doi.org/10.1080/15230406.2013.776727.
- Kessler, R. C., Galea, S., Gruber, M. J., Sampson, N. A., Ursano, R. J., and Wessely, S. 2008. "Trends in mental illness and suicidality after Hurricane Katrina." Molecular psychiatry, 13(4), 374-384.
- Khan, M. L. 2017. "Social media engagement: What motivates user participation and consumption on YouTube?" *Computers in Human Behavior*, 66, 236-247.
- Kim, H., Marcouiller, D. W., and Woosnam, K. M. 2018. "Rescaling social dynamics in climate change: The implications of cumulative exposure, climate justice, and community resilience." Geoforum, 96, 129-140.

- Kim, J. H., and Choi, I. 2019. "Choosing the level of significance: A decision-theoretic approach." Abacus.
- Kinjo, K., and Ebina, T. 2017. "Optimal program for autonomous driving under Bentham-and Nash-type social welfare functions." *Procedia computer science*, 112, 61-70.
- Klise, K. A., Bynum, M., Moriarty, D., and Murray, R. 2017. "A software framework for assessing the resilience of drinking water systems to disasters with an example earthquake case study." *Environmental modelling & software*, 95, 420-431. https://doi.org/10.1016/j.envsoft.2017.06.022.
- Kopf, B. 2022. "The Power of Figma as a Design Tool" Last accessed Sep 30, 2022. <a href="https://www.toptal.com/designers/ui/figma-design-tool">https://www.toptal.com/designers/ui/figma-design-tool</a>
- Krause, E., and Reeve, Richard V., 2017. "Hurricanes hit the poor the hardest" Last accessed April 06, 2020. <a href="https://www.brookings.edu/blog/social-mobility-memos/2017/09/18/hurricanes-hit-the-poor-the-hardest/">https://www.brookings.edu/blog/social-mobility-memos/2017/09/18/hurricanes-hit-the-poor-the-hardest/</a>.
- Kryvasheyeu, Y., Chen, H., Obradovich, N., Moro, E., Van Hentenryck, P., Fowler, J., and Cebrian, M. 2016. "Rapid assessment of disaster damage using social media activity." Science advances, 2(3), e1500779.
- Kwak, N., Lane, D. S., Weeks, B. E., Kim, D. H., Lee, S. S., and Bachleda, S. 2018. "Perceptions of social media for politics: Testing the slacktivism hypothesis." *Human Communication Research*, 44(2), 197-221.
- Kwasinski, A. 2011. "Field damage assessments as a design tool for information and communications technology systems that are resilient to natural disasters." In *Proceedings* of the 4th International Symposium on Applied Sciences in Biomedical and Communication Technologies, 1-6.
- Laerd 2020a. "Pearson's Product-Moment Correlation using SPSS Statistics" Last accessed Aug 20, 2020. <a href="https://statistics.laerd.com/spss-tutorials/pearsons-product-moment-correlation-using-spss-statistics.php">https://statistics.laerd.com/spss-tutorials/pearsons-product-moment-correlation-using-spss-statistics.php</a>
- Laerd 2020b. "Spearman's Rank-Order Correlation using SPSS Statistics." Last accessed Jan 10, 2020. <a href="https://statistics.laerd.com/spss-tutorials/spearmans-rank-order-correlation-using-spss-statistics.php">https://statistics.laerd.com/spss-tutorials/spearmans-rank-order-correlation-using-spss-statistics.php</a>
- Lam, C. Y., and Tai, K. 2018. "Modeling infrastructure interdependencies by integrating network and fuzzy set theory." *International Journal of Critical Infrastructure Protection*, 22, 51-61.
- Lazo, L. 2017 "Irma washed away part of a Florida highway; roads remain dangerous and impassable" Last accessed Sep 10, 2022, <a href="https://www.washingtonpost.com/news/dr-gridlock/wp/2017/09/11/irma-washed-away-part-of-a-florida-highway-roads-remain-dangerous-and-impassable/">https://www.washingtonpost.com/news/dr-gridlock/wp/2017/09/11/irma-washed-away-part-of-a-florida-highway-roads-remain-dangerous-and-impassable/</a>
- Li, Z., Huang, Q., and Emrich, C. T. 2019. "Introduction to social sensing and big data computing for disaster management" 1198-1204.

- Li, Z., Wang, C., Emrich, C. T., and Guo, D. 2018. "A novel approach to leveraging social media for rapid flood mapping: a case study of the 2015 South Carolina floods." *Cartography and Geographic Information Science*, 45(2), 97-110. https://doi.org/10.1080/15230406.2016.1271356.
- Lindsay, B. R. 2011. "Social media and disasters: Current uses, future options, and policy considerations." *Congressional Research Service*.
- Liu, H. J., Love, P. E., Sing, M. C., Niu, B., and Zhao, J. (2019). Conceptual framework of lifecycle performance measurement: Ensuring the resilience of transport infrastructure assets. *Transportation Research* Part D: *Transport and Environment*, 77, 615-626
- Lowe, T. 2017. "The Future of Florida's Rural Communities" Last accessed June 20, 2022. < https://www.flchamber.com/the-future-of-floridas-rural-communities-2/>
- Lynn, K., MacKendrick, K., and Donoghue, E. M. 2011. "Social vulnerability and climate change: synthesis of literature." Gen. Tech. Rep. PNW-GTR-838. Portland, OR: US Department of Agriculture, Forest Service, Pacific Northwest Research Station. 70 p., 838.
- Maass, W., Parsons, J., Purao, S., Storey, V. C., and Woo, C. 2018. "Data-driven meets theorydriven research in the era of big data: opportunities and challenges for information systems research." *Journal of the Association for Information Systems*, 19(12), 1.
- MacKenzie, C. A., and Barker, K. 2013. "Empirical data and regression analysis for estimation of infrastructure resilience with application to electric power outages". J. Infra. Syst., 19(1), 25-35.
- MacKenzie, C. A., and Zobel, C. W. 2016. "Allocating resources to enhance resilience, with application to superstorm sandy and an electric utility." Risk Analysis, 36(4), 847-862.
- Mao, Q., and Li, N. 2018. "Assessment of the impact of interdependencies on the resilience of networked critical infrastructure systems." *Natural hazards*, 93(1), 315-337.
- Martín, Yago, Zhenlong Li, and Susan L. Cutter. 2017. "Leveraging Twitter to Gauge Evacuation Compliance: Spatiotemporal Analysis of Hurricane Matthew." *PLOS ONE*, 12 (7), e0181701. doi: 10.1371/journal.pone.0181701.
- Masozera, M., Bailey, M., and Kerchner, C. 2007. "Distribution of impacts of natural disasters across income groups: A case study of New Orleans." *Ecological Economics*, 63(2-3), 299-306
- Massarra, Carol C., 2012. "Hurricane damage assessment process for residential buildings." *LSU Master's Theses, 5220.* https://digitalcommons.lsu.edu/gradschool\_theses/520
- Mawgoud, A. A., Eltabey, M. M., and Abu-Talleb, A. 2021. "A Distributed Artificial Intelligence Framework to Evolve Infrastructure Resilience in Telecommunications Sector." In International Conference on Advanced Machine Learning Technologies and Applications, 774-784, Springer.
- Meerow, S., and Neweell, J.P., 2019. "Urban Resilience for Whom, What, When, Where, and Why?". Urban Geography 40, 309–329.

- Meerow, S., Pajouhesh, P., and Miller, T. R. 2019. "Social equity in urban resilience planning." *Local Environment*, 24(9), 793-808.
- Mills, A., Chen, R., Lee, J., and Raghav Rao, H. 2009. "Web 2.0 emergency applications: How useful can Twitter be for emergency response?". *Journal of Information Privacy and Security*, 5(3), 3-26.
- Mills, M. A., Edmondson, D., and Park, C. L. 2007. "Trauma and stress response among Hurricane Katrina evacuees." American Journal of Public Health, 97(S1), S116-123.
- Mitsova, D., Esnard, A. M., Sapat, A., and Lai, B. S. 2018. "Socioeconomic vulnerability and electric power restoration timelines in Florida: The case of Hurricane Irma." *Natural Hazards*, *94*(2), 689-709.
- Moens, J. 2022. "Hurricane Michael Hit the Florida Panhandle in 2018 With 155 MPH Winds. Some Black and Low-Income Neighborhoods Still Haven't Recovered" Last accessed June 25, 2022. < https://insideclimatenews.org/news/22032022/hurricane-michael-black-lowincome-neighborhoods/>
- Moline M. 2019. "It's taking forever for Hurricane Michael disaster aid to reach the wrecked Florida Panhandle" Last accessed June 10, 2022 <<u>https://floridaphoenix.com/2019/08/23/its-taking-forever-for-hurricane-michael-disaster-aid-to-reach-the-wrecked-florida-panhandle/</u>>
- Morris, J. T., Mueller, J. L., and Jones, M. L. 2014. "Use of social media during public emergencies by people with disabilities." *Western Journal of Emergency Medicine*, 15(5), 567.
- Mostafa, M. A., and El-Gohary, N. M. 2014. "Stakeholder-sensitive social welfare-oriented benefit analysis for sustainable infrastructure project development." J. Constr. Eng. Manag., 140(9), 04014038
- Mottahedi, A., Sereshki, F., Ataei, M., Nouri Qarahasanlou, A., and Barabadi, A. 2021. "The Resilience of Critical Infrastructure Systems: A Systematic Literature Review." *Energies*, 14(6), 1571.
- Mukaka, M. M. 2012. "A guide to appropriate use of correlation coefficient in medical research." *Malawi medical journal*, 24(3), 69-71.
- NASEM (National Academies of Sciences, Engineering, and Medicine) 2022. "Equitable and Resilient Infrastructure Investments" Washington, DC, The National Academies Press. https://doi.org/10.17226/26633.
- Nateghi, R. 2018. "Multi-dimensional infrastructure resilience modeling: an application to hurricane-prone electric power distribution systems." *IEEE Access*, *6*, 13478-13489.
- Nazer, T. H., Morstatter, F., Dani, H., and Liu, H. 2016. "Finding requests in social media for disaster relief." In 2016 IEEE/ACM International Conference on Advances in Social Networks Analysis and Mining (ASONAM) (pp. 1410-1413). IEEE.
- NCSL (National Conference of State Legislatures) 2020. "Challenges Facing Rural Communities" Last accessed June 05, 2022. < https://www.ncsl.org/research/agriculture-and-rural-

development/challenges-facing-rural-

communities.aspx#:~:text=Rural%20communities%20face%20challenges%20related,and %20environment%20and%20community%20preservation.>

- Neumayer, E., and Plümper, T. 2007. "The gendered nature of natural disasters: The impact of catastrophic events on the gender gap in life expectancy, 1981–2002." Annals of the Association of American Geographers, 97(3), 551-566.
- Nexus 2017. "Hurricane Harvey hit low-income communities hardest" Last accessed Feb 05, 2020. <a href="https://archive.thinkprogress.org/hurricane-harvey-hit-low-income-communities-hardest-d13506b7e60/">https://archive.thinkprogress.org/hurricane-harvey-hit-low-income-communities-hardest-d13506b7e60/</a>
- Nicholson, G. 2014. "Inequality and its impact on the resilience of societies." Association of Caribbean States.
- NHC (National Hurricane Centre) 2018. "Hurricane Michael" Last accessed Jan 10, 2020. < https://www.nhc.noaa.gov/data/tcr/AL142018\_Michael.pdf>.
- NOAA (National Oceanic and Atmospheric Administration) 2018a. "Hurricane Michael upgraded to a Category 5 at time of U.S. landfall" Last accessed Feb 10, 2020. < https://www.noaa.gov/media-release/hurricane-michael-upgraded-to-category-5-at-timeof-us-landfall>
- NOAA (National Oceanic and Atmospheric Administration) 2018b. "Hurricane Irma", *National Hurricane Center, Tropical Cyclone Report*. Last accessed May 14, 2022, < <u>https://www.nhc.noaa.gov/data/tcr/AL112017\_Irma.pdf</u>>
- NOAA (National Oceanic and Atmospheric Administration) 2019. "Hurricane Michael", *National Hurricane Center, Tropical Cyclone Report*. Last accessed May 14, 2022, < <u>https://www.nhc.noaa.gov/data/tcr/AL142018\_Michael.pdf</u>>
- NOAA (National Oceanic and Atmospheric Administration) 2021a. "Hurricane Sally", *National Hurricane Center, Tropical Cyclone Report*. Last accessed May 14, 2022, < https://www.nhc.noaa.gov/data/tcr/AL192020\_Sally.pdf>
- NOAA (National Oceanic and Atmospheric Administration) 2021b. "Define Coastal Counties" Last accessed June 10, 2022, < https://coast.noaa.gov/digitalcoast/training/definingcoastal-counties.html>.
- NOAA (National Oceanic and Atmospheric Administration) 2022. "Billion-Dollar Weather and Climate Disasters." National Centers for Environmental Information, Last accessed June 10, 2022. <u>https://www.ncei.noaa.gov/access/monitoring/billions/</u>.
- Nogal, M., O'Connor, A., Martinez-Pastor, B., and Caulfield, B. 2017. "Novel probabilistic resilience assessment framework of transportation networks against extreme weather events." ASCE-ASME J.of Ris. and Uncer. in Eng. Syst., Part A: Civil Engineering, 3(3), 04017004.
- NRC (National Research Council) 2012. "Disaster Resilience: A National Imperative" The National Academics Press, 2012

- NWS (National Weather Service) 2018. "Catastrophic Hurricane Michael Strikes Florida Panhandle" October 10, 2018" last accessed June 10, 2021. https://www.weather.gov/tae/HurricaneMichael2018>
- Olson, A., 2017. "The Man's Role in Society and Family: the Crucial Change of the 21st Century" Last accessed April 10, 2020. <a href="https://goodmenproject.com/featured-content/mans-role-society-family-crucial-change-21st-century-spnsr/">https://goodmenproject.com/featured-content/mans-role-society-family-crucial-change-21st-century-spnsr/</a>
- Ouyang, M., and Duenas-Osorio, L. 2014. "Multi-dimensional hurricane resilience assessment of electric power systems." *Structural Safety*, 48, 15-24.
- Ouyang, M., and Wang, Z. 2015. "Resilience assessment of interdependent infrastructure systems: With a focus on joint restoration modeling and analysis." *Reliability Engineering & System Safety*, 141, 74-82.
- Pagano, A., Sweetapple, C., Farmani, R., Giordano, R., and Butler, D. 2019. "Water distribution networks resilience analysis: A comparison between graph theory-based approaches and global resilience analysis." *Water Resources Management*, 33(8), 2925-2940. https://doi.org/10.1007/s11269-019-02276-x
- Panteli, M., Mancarella, P., Trakas, D. N., Kyriakides, E., and Hatziargyriou, N. D. 2017. "Metrics and quantification of operational and infrastructure resilience in power systems." *IEEE, Transactions on Power Systems*, 32(6), 4732-4742.
- Pathak, A., Zhang, L., and Ganapati, N.E. 2020. "Understanding Multi-Sector Stakeholder Value Dynamics in Hurricane Michael to Facilitate Collaborative Decision Making in Disaster Contexts." *Natural Hazards Review*, ASCE, in press.
- Petit, F. D., Eaton, L. K., Fisher, R. E., McAraw, S. F., and Collins III, M. J. 2012. "Developing an index to assess the resilience of critical infrastructure." *Intl J. Ris. Ass. and Mgmt.*, 16(1-3), 28-47.
- Planitz, A., 1999. "A Guide to Successful Damage And Needs Assessment." *South Pacific Disaster Reduction Programme* (SPDRP).
- PyPI (Python Package Index). 2018. "Twitter Intelligence Tool" Last accessed May 02, 2019. <<u>https://pypi.org/project/twint/</u>>
- Raizner 2021. "Why Hurricanes Pose Property Damage Concerns For Inland Communities" Last accessed July 10, 2022. < https://www.raiznerlaw.com/blog/why-hurricanes-pose-property-damage-concerns-for-inland-communities/ >.
- Rao, K. D., Makimoto, S., Peters, M., Leung, G. M., Bloom, G., and Katsuma, Y. 2019. "Vulnerable Populations and Universal Health Coverage." Leave No One Behind: Time for Specifics on the Sustainable Development Goals.
- Rasoulkhani, K., Mostafavi, A., Cole, J., and Sharvelle, S. 2019. "Resilience-based infrastructure planning and asset management: Study of dual and singular water distribution infrastructure performance using a simulation approach. Sustainable Cities and Society, 48, 101577. https://doi.org/10.1016/j.scs.2019.101577

Rawls, J. 1971. "A Theory of Justice." Harvard University Press, Cambridge, MA

- Rehak, D., Senovsky, P., Hromada, M., and Lovecek, T. 2019. "Complex approach to assessing resilience of critical infrastructure elements." *Int. J. of C. Infra. Prot.*, 25, 125-138.
- Resch, B., Usländer, F., and Havas, C. 2018. "Combining machine-learning topic models and spatiotemporal analysis of social media data for disaster footprint and damage assessment." Cartography and Geographic Information Science, 45(4), 362-376.
- RF (The Rockefeller Foundation) 2021. "100 Resilient Cities" Last accessed June 2, 2021. (https://www.rockefellerfoundation.org/our-work/initiatives/100-resilient-cities/).
- Rhodes, J., Chan, C., Paxson, C., Rouse, C. E., Waters, M., and Fussell, E. 2010. "The impact of Hurricane Katrina on the mental and physical health of low-income parents in New Orleans." American journal of orthopsychiatry, 80(2), 237.
- Riccucci, N. M., and Van Ryzin, G. G. 2017. "Representative bureaucracy: A lever to enhance social equity, coproduction, and democracy." Public Administration Review, 77(1), 21-30.
- Rockefeller Foundation. 2020. "What is Urban Resilience?" Last accessed April 06, 2020. <<u>https://www.rockefellerfoundation.org/100-resilient-cities/</u>>.
- Rodríguez-Izquierdo, R. M. 2018. "Researching the links between social-emotional learning and intercultural education: strategies for enacting a culturally relevant teaching." *Intercultural Education*, 29(5-6), 609-623. https://doi.org/10.1080/14675986.2018.1528527.
- Roshan, M., Warren, M., and Carr, R. 2016. "Understanding the use of social media by organizations for crisis communication." *Computers in Human Behavior*, 63, 350-361. https://doi.org/10.1016/j.chb.2016.05.016
- Rus, K., Kilar, V., and Koren, D. 2018. "Resilience assessment of complex urban systems to natural disasters: a new literature review." *International journal of disaster risk reduction*, 31, 311-330. https://doi.org/10.1016/j.ijdrr.2018.05.015
- SAMSHA (Substance Abuse and Mental Health Services Association). 2017. "Greater Impact : How Disasters Affect People of Low Socioeconomic Status", *Supplemental Research Bulletin, Disaster Technical Assistance Center, <* https://www.samhsa.gov/sites/default/files/dtac/srb-low-ses\_2.pdf>.
- Santos, R., 2019. "Fairness, Not Just Accuracy, Is Vital to the 2010 Census" Last accessed Jan 20, 2021. < https://www.urban.org/urban-wire/fairness-not-just-accuracy-vital-2020-census>
- Saunders, J. 2020 "Hurricane Sally estimated to have cost Gulf Power \$200 million to restore electricity" Last accessed Sep 10, 2022. <a href="https://www.pnj.com/story/news/2020/10/21/hurricane-sally-estimated-have-cost-gulf-power-200-million-restore-electricity/3717578001/">https://www.pnj.com/story/news/2020/10/21/hurricane-sally-estimated-have-cost-gulf-power-200-million-restore-electricity/3717578001/</a>
- Schempp, T., Zhang, H., Schmidt, A., Hong, M., and Akerkar, R. 2019. "A framework to integrate social media and authoritative data for disaster relief detection and distribution optimization." International Journal of Disaster Risk Reduction, 39, 101143.

- Schneider, M., and Kim, B. C. 2020. "The utilitarian-maximin social welfare function and anomalies in social choice." Southern Economic Journal, 87(2), 629-646
- Schneiderbauer, S., Ehrlich, D., and Birkmann, J. 2006. "Social levels and hazard (in) dependence in determining vulnerability." Measuring vulnerability to natural hazards: Towards disaster resilient societies, *Birkmann, J. eds., United University Press*, 78-102
- Sen, A. 1997. On Economic Inequality. Oxford University Press, Oxford, UK.
- Senkbeil, J. C., Myers, L., Jasko, S., Reed, J. R., and Mueller, R. 2020. "Communication and hazard perception lessons from category five hurricane Michael". *Atmosphere*, 11(8), 804.
- Shafieezadeh, A., and Burden, L. I. 2014. "Scenario-based resilience assessment framework for critical infrastructure systems: Case study for seismic resilience of seaports." *Reliability Engineering & System Safety*, 132, 207-219. https://doi.org/10.1016/j.ress.2014.07.021
- Shang, Q., Wang, T., and Li, J. 2020. "A Quantitative Framework to Evaluate the Seismic Resilience of Hospital Systems." *Journal of Earthquake Engineering*, 1-25.
- Shareef, M. A., Mukerji, B., Dwivedi, Y. K., Rana, N. P., and Islam, R. 2019. "Social media marketing: Comparative effect of advertisement sources." *Journal of Retailing and Consumer Services*, 46, 58-69.
- Shin, S., Lee, S., Judi, D. R., Parvania, M., Goharian, E., McPherson, T., and Burian, S. J. 2018. "A systematic review of quantitative resilience measures for water infrastructure systems." *Water*, 10(2), 164
- Shively, D., 2017. "Flood risk management in the USA: implications of National Flood Insurance Program changes for social justice." Reg Environ Change 17, 1663–1672. https://doi.org/10.1007/s10113-017-1127-3.
- Shultz, J. M., and Galea, S. 2017. "Mitigating the mental and physical health consequences of Hurricane Harvey." Jama, 318(15), 1437-1438.
- Singhal, T. K., Kwon, O. S., Bentz, E. C., and Christopoulos, C. 2020. "Development of a civil infrastructure resilience assessment framework and its application to a nuclear power plant." *Structure and Infrastructure Engineering*, 18(1), 1-14.
- Starbird, K., and Palen, L. 2010. "Pass It On?: Retweeting in Mass Emergency." In Proceedings of the 7th International ISCRAM Conference–Seattle, USA, (Vol. 1), May.
- Statistics Solutions 2020. "Correlation (Pearson, Kendall, Spearman)" Last accessed on Aug 30. < https://www.statisticssolutions.com/correlation-pearson-kendall-spearman/>
- Stofan, J., 2017. "Rural counties still awaiting Irma relief" Last accessed June10, 2022. < https://www.wctv.tv/content/news/Rural-counties-still-awaiting-Irma-relief-444273433.html>
- Stough, L., 2017. "What Comes After Hurricane Harvey for People with Disabilities? Restoring, Recovering, and Rebuilding" Last accessed Jan 12, 2020. <

https://hazards.colorado.edu/news/research-counts/what-comes-after-hurricane-harvey-for-people-with-disabilities-restoring-recovering-and-rebuilding >

- Surani, Z., Hirani, R., Elias, A., Quisenberry, L., Varon, J., Surani, S., and Surani, S. 2017. "Social media usage among health care providers." *BMC research notes*, 10(1), 654.
- Tate, E., Rahman, M. A., Emrich, C. T., and Sampson, C. C. 2021. "Flood exposure and social vulnerability in the United States." *Natural Hazards*, *106*(1), 435-457.
- Thomas, D. S., Phillips, B. D., Lovekamp, W. E., and Fothergill, A. 2013. "Social vulnerability to disasters". *CRC Press*. Taylor and Francis Group, LLC.
- Todoroff, A., 2020. "Rural Communities Rise to the Challenge of Dual Disasters" Last accessed June 15, 2022. < https://www.eesi.org/briefings/view/061820rural>
- Tonn, G., Czajkowski, J., Kunreuther, H., Angotti, K., and Gelman, K. 2020. "Measuring transportation infrastructure resilience: Case study with Amtrak." *Journal of Infrastructure Systems*, *26*(1), 05020001.
- Trapeznikova, I. 2019. "Measuring income inequality." IZA, *World of Labor*. doi: 10.15185/izawol.462
- Tselios, V., and Tompkins, E. L. 2019. "What causes nations to recover from disasters? An inquiry into the role of wealth, income inequality, and social welfare provisioning." *Int. J. of Dis. Ris. Red.*, *33*, 162-180.
- Twitter Inc 2020. "Tweet location FAQs." Last accessed June 05, 2020. <<u>https://help.twitter.com/en/safety-and-security/tweet-location-settings</u>>
- UN (United Nations). 2016. "Global Sustainable Development Report 2016" Department of Economic and Social Affairs, New York, July.
- UNCTAD (United Nations Conference on Trade and Development) 2021. "The many faces of inequality" Last accessed May 15, 2022. < https://sdgpulse.unctad.org/in-focus-inequality/#Ref\_Cobham2013>
- US Census. 2019. "Access Local Data." Last accessed on Oct 02, 2019. < https://www.census.gov/>
- USDA-ERS (United States Department of Agriculture- Economic Research Service) 2022. "State Facts Sheet: Florida" Last accessed June 20, 2022. < https://data.ers.usda.gov/reports.aspx?StateFIPS=12&StateName=Florida&ID=17854>.
- USDOC (United States Department of Commerce) 2019. "Exploring the digital nation- America emerging online experience" Last accessed Jan 20, 2020. <<u>http://www.ntia.doc.gov/files/ntia/publications/exploring\_the\_digital\_nation\_-</u> americas\_emerging\_online\_experience.pdf.>
- USRC (U.S. Resiliency Council) 2021. "Resilient Communities Need Resilient Buildings" Last accessed July 25, 2021. <a href="http://www.usrc.org/">http://www.usrc.org/></a>

- Venema, H. 2017 "Eight Ways Cities Are Building Climate Resilience" Last accessed Sep 12, 2022. <a href="https://www.iisd.org/articles/building-a-climate-resilient-city">https://www.iisd.org/articles/building-a-climate-resilient-city</a>
- Wamsley, L., 2019. "Hurricane Michael Was A Category 5, NOAA Finds The First Since Andrew In 1992" Last accessed Jan 12, 2020. < https://www.npr.org/2019/04/19/715134716/hurricane-michael-was-a-category-5-noaafinds-the-first-since-andrew-in-1992>
- Wang, X. J., Zhang, J. Y., Shahid, S., ElMahdi, A., He, R. M., Wang, X. G., and Ali, M. 2012. "Gini coefficient to assess equity in domestic water supply in the Yellow River." *Mitigation and Adaptation Strategies for Global Change*, 17(1), 65–75. doi:10.1007/s11027-011-9309-7.
- Ward, P. S., and Shively, G. E. 2016. "Disaster risk, social vulnerability, and economic development." *Disasters*, 41(2), 324-351.
- Wei, W., Jiayue, Z., Xiaoling, Y., Haiqing, X., and Chuangxin, G. 2020. "Resilience Assessment Using Simulation System of Distribution Network under Extreme Weather." In 2020 IEEE Sustainable Power and Energy Conference (iSPEC) (pp. 895-902), IEEE.
- Wescoat, J., Vale, L., Ocampo, M. A., Mazereeuw, M., Segal, R., Spirn, A. W., Frenchman, D., Scott, A., Berger A. M., Adams M. L., Susskind, J., Norford, L., Ryan, B., Hsu, D., Williams, S., Steil, J., Ghosn, R., Reinhart, C., and Sass, L., 2018. "Equitable Resilience, 2018-2021." *Norman B. Leventhal Center for Advanced Urbanism*, Massachusetts Institute of Technology, *School of Architecture + Planning*, Last accessed June 6, 2022. <<u>https://lcau.mit.edu/equitableresilience</u>>
- Weymark, J. A. 2016. "Social welfare functions." The Oxford handbook of well-being and public policy, 126-59.
- Wodon, Q., and Yitzhaki, S. 2008. "Inequality in Multidimensional Indicators of Well-Being: Methodology and Application to the Human Development Index." In *Modeling Income Distributions and Lorenz Curves*, pp. 303-317, *Springer*, New York, NY.
- Wu, D., and Cui, Y. 2018. "Disaster early warning and damage assessment analysis using social media data and geo-location information." Decision Support Systems, 111, 48-59.
- Wukich, C., Hu, Q., and Siciliano, M. D. 2019. "Cross-Sector Emergency Information Networks on Social Media: Online Bridging and Bonding Communication Patterns." The American Review of Public Administration, 49(7), 825-839.
- Yabe, T., and Ukkusuri, S. V. 2020. "Effects of income inequality on evacuation, reentry and segregation after disasters." *Transportation research part D: transport and environment*, 82, 102260.
- Yang, Y., Tang, W., Liu, Y., Xin, Y., and Wu, Q. 2018. "Quantitative resilience assessment for power transmission systems under typhoon weather." *IEEE Access*, *6*, 40747-40756.
- Yodo, N., and Wang, P. 2016. "Engineering resilience quantification and system design implications: A literature survey." Journal of Mechanical Design, 138(11).

- Yoon, D. K. 2012. "Assessment of social vulnerability to natural disasters: a comparative study." *Natural hazards*, *63*(2), 823-843.
- Young, C. E., Kuligowski, E. D., and Pradhan, A. 2020. "A Review of Social Media Use During Disaster Response and Recovery Phases." *National Institute of Standards and Technology*, U.S. Department of Commerce. https://doi.org/10.6028/NIST.TN.2086
- Young, H., 2020. "How does the language you speak shape your experience of the internet?" Last accessed Feb 12, 2020. <a href="http://labs.theguardian.com/digital-language-divide/">http://labs.theguardian.com/digital-language-divide/</a>
- Zamanian, S., Rahimi, M., and Shafieezadeh, A. 2020. "Resilience of Sewer Networks to Extreme Weather Hazards: Past Experiences and an Assessment Framework." In *Pipelines 2020*, 50-59. Reston, VA: *American Society of Civil Engineers*
- Zhang, C., Kong, J. J., and Simonovic, S. P. 2018. "Restoration resource allocation model for enhancing resilience of interdependent infrastructure systems." Safety science, 102, 169-177. https://doi.org/10.1016/j.ssci.2017.10.014
- Zhang, L., and Sanake, S. 2020. "Social Welfare-Based Human Comfort Aggregation Model to Facilitate Healthy and Comfortable Indoor Environments." J. Arch. Eng., 26(3), 04020027
- Zhao, X., Chen, Z., and Gong, H. 2015. "Effects comparison of different resilience enhancing strategies for municipal water distribution network: A multidimensional approach." *Mathematical Problems in Engineering*, 2015. https://doi.org/10.1155/2015/438063.
- Zhu, Y., Ozbay, K., Xie, K., and Yang, H. 2016. "Using big data to study resilience of taxi and subway trips for hurricanes Sandy and Irene." *Transportation research record*, 2599(1), 70-80. https://doi.org/10.3141/2599-09.
- Zhu, Y., Xie, K., Ozbay, K., Zuo, F., and Yang, H. 2017. "Data-driven spatial modeling for quantifying networkwide resilience in the aftermath of hurricanes Irene and Sandy." *Transportation research record*, 2604(1), 9-18.
- Zou, L., Lam, N. S., Cai, H., and Qiang, Y., 2018a. "Mining Twitter Data for Improved Understanding of Disaster Resilience." Annals of the American Association of Geographers 108: 1422–1441. doi:10.1080/24694452. 2017.1421897.
- Zou, L., Lam, N. S., Shams, S., Cai, H., Meyer, Yang, S., Lee, K., Park, S. J., and Reams, M. A., 2018b. "Social and geographical disparities in Twitter use during Hurricane Harvey", *International Journal of Digital Earth*, DOI: 10.1080/17538947.2018.1545878

# VITA

# SUNIL DHAKAL

	Born, Biratnagar, Nepal
2009-2013	B.S., Civil Engineering Kathmandu University Dhulikhel, Nepal
2014-2015	Shiva Laxmi Consultants, Biratnagar, Nepal
2015-2016	Building Design Authority Pvt Ltd. Kathmandu, Nepal
2016-2017	National Reconstruction Authority Government of Nepal, Sindhupalchowk
2017 -2022	Teaching and Research Assistant Florida International University Miami, Florida
2017 -2022	Doctoral Candidate Florida International University Miami, Florida

# PUBLICATIONS AND PRESENTATIONS

Dhakal, S., Zhang, L., and Lv, X. (2019). Ontology-based semantic modelling to support knowledge-based document classification on disaster-resilient construction practices. *International Journal of Construction Management*, 22(11), 2059-2078.

Dhakal, S., Zhang, L., and Lv, X. (2021). Understanding Infrastructure Resilience, Social Equity, and Their Interrelationships: Exploratory Study Using Social Media Data in Hurricane Michael. *Natural Hazards Review*, 22(4), 04021045.

Muhaimin, A. M. M., Zhang, L., Dhakal, S., Lv, X., Pradhananga, N., Kalasapudi, V. S., and Azizinamini, A. (2021). Identification and analysis of factors affecting the future of bridge design, construction, and operation. *Journal of Management in Engineering*, *37*(5), 04021049.

Dhakal, S., and Zhang, L. (2022). A Social-Welfare-Based Infrastructure Resilience Assessment Framework: Toward Equitable Resilience for Infrastructure Development. *Natural Hazards Review* (in press)

Dhakal, S., and Zhang, L. (2019). Ontology-based semantic modeling of disaster resilient construction operations: towards a knowledge-based decision support system. In *Advances in Informatics and Computing in Civil and Construction Engineering*, (pp. 789-796), Cham, Switzerland, *Springer Nature*.

Zhang, L., Lv, X., and Dhakal, S. (2019). A reinforcement learning-based stakeholder value aggregation model for collaborative decision making on disaster resilience. In *Computing in Civil Engineering 2019: Smart Cities, Sustainability, and Resilience* (pp. 490-497). Reston, VA, *ASCE*.

Dhakal, S., and Zhang, L. (2020). Understanding the Interrelationships between Infrastructure Resilience and Social Equity Using Social Media Data. In *Construction Research Congress 2020*, Reston, VA, *ASCE*.

Dhakal, S., and Zhang, L. (2022) Integrating Social Equity and Vulnerability with Infrastructure Resilience Assessment. In *Construction Research Congress 2022* (pp. 299-309), Reston, VA, *ASCE*.